.

·

**CWI** Tract

69

# Monotonicity properties of infinitely divisible distributions

B.G. Hansen



Centrum voor Wiskunde en Informatica Centre for Mathematics and Computer Science

1980 Mathematics Subject Classification: 60E07, 60E10, 60F05. ISBN 90 6196 380 X NUGI-code: 811

Copyright  $^{\odot}$  1990, Stichting Mathematisch Centrum, Amsterdam Printed in the Netherlands

### ACKNOWLEDGEMENTS

This monograph is a result of my research carried out at the Eindhoven University of Technology, The Netherlands, during the years 1985 to 1988. The subject matter was suggested to me by my thesis advisor Prof. F.W. Steutel, to whom I am very grateful and indebted for his enthusiasm, stimulance and encouragement, with out which this monograph would not have been possible to write. I would also like to thank Prof. J.Th. Runnenburg, Prof. J. de Graaf and Prof. L. de Haan for their useful suggestions to improvements of an earlier version of this monograph. My special thanks goes to my co-promotor Prof. W. Vervaat for his careful reading and many helpful comments on this manuscript.

Further I would like to thank the Mathematical Centre for the opportunity to publish this monograph in their series: Mathematical Centre Tracts. Finally I would like to thank Prof. L.A. Baxter for his two inspiring courses in Probability Theory at the State University of New York at Stony Brook, which introduced me to this subject.

, .

# Monotonicity Properties of Infinitely Divisible Distributions

# **Table of Contents**

| 1. | 1. PRELIMINARIES  |  |    |  |  |  |  |
|----|---|--|----|--|--|--|--|
|    | 1.1   | Introduction and summary                                   | 1  |  |  |  |  |
|    | 1.2   | Notations and conventions                                  | 2  |  |  |  |  |
|    | 1.3   | Infinitely divisible distributions                         | 3  |  |  |  |  |
|    | 1.4   | Self-decomposable and stable distributions                 | 8  |  |  |  |  |
|    | 1.5   | Motivation and methodology                                 | 11 |  |  |  |  |
| 2. | 2. INFINITELY DIVISIBLE SEQUENCES AND RENEWAL SEQUENCES |  |    |  |  |  |  |
|    | 2.1   | Introduction   | 12 |  |  |  |  |
|    | 2.2   | A taste of renewal theory                                  | 12 |  |  |  |  |
|    | 2.3   | Monotonicity results in renewal theory                     | 13 |  |  |  |  |
|    | 2.4   | Monotonicity results of infinitely divisible distributions | 15 |  |  |  |  |
|    | 2.5   | Renewal sequences and infinitely divisible sequences       | 17 |  |  |  |  |
| 3. | MOMENT  | SEQUENCES AND MOMENT FUNCTIONS                             |    |  |  |  |  |
|    | 3.1   | Introduction   | 18 |  |  |  |  |
|    | 3.2   | Preliminaries  | 19 |  |  |  |  |
|    | 3.3   | Moment sequences and infinitely divisible sequences        | 23 |  |  |  |  |
|    | 3.4   | Moment sequences and renewal sequences                     | 30 |  |  |  |  |
|    | 3.5   | Moment functions and infinitely divisible functions        | 31 |  |  |  |  |
|    | 3.6   | Moment functions and renewal functions                     | 34 |  |  |  |  |
|    | 3.7   | Applications and special cases                             | 34 |  |  |  |  |

| 4.   | CAVE AND LOGCONVEX SEQUENCES AND DENSITIES |  |    |  |  |
|--|--|--|----|--|--|
|  | 4.1  | Introduction                                       | 37 |  |  |
|  | 4.2  | Discrete distributions                             | 38 |  |  |
|  | 4.3  | Absolutely continuous distributions                | 41 |  |  |
|  | 4.4  | Applications and counterexamples                   | 43 |  |  |
| 5. A GENERALIZED SELF-DECOMPOSABILITY                    |  |  |    |  |  |
|  | 5.1  | Introduction                                       | 46 |  |  |
|  | 5.2  | α-Unimodality                                      | 47 |  |  |
|  | 5.3  | Distributions on $\mathbb{N}_0$                    | 48 |  |  |
|  | 5.4  | Distributions on IR                                | 52 |  |  |
|  | 5.5  | Distributions on $\mathbb{R}_{\perp}$              | 64 |  |  |
|  | 5.6  | A classification of ID(I)                          | 66 |  |  |
| 6. α-SELF-DECOMPOSABILITY AND LIMIT LAWS                 |  |  |    |  |  |
|  | 6.1  | Introduction                                       | 68 |  |  |
|  | 6.2  | Sums of uan triangular arrays                      | 69 |  |  |
|  | 6.3  | Sums of shrunken random variables                  | 72 |  |  |
|  | 6.4  | Subsets of the set of self-decomposable limit laws | 83 |  |  |
|  | 6.5  | Remarks and comments                               | 84 |  |  |
| A  | PPENDIX                                    |  |    |  |  |
| Limit Distributions of Sums of Shrunken Random Variables |  |  |    |  |  |
|  | A.1  | Introduction                                       | 85 |  |  |
|  | A.2  | Preliminaries                                      | 86 |  |  |
|  | A.3  | A characterization of U(S,p)                       | 88 |  |  |
|  | A.4  | Stability in U(S,p)                                | 92 |  |  |
|  | A.5  | Some subsets of U(S,p)                             | 94 |  |  |
|  | A.6  | Examples   | 95 |  |  |
| R  | REFERENCES                                 |  |    |  |  |
| L  | LIST OF SYMBOLS                            |  |    |  |  |
| I  | INDEX                                      |  |    |  |  |

ii

**Chapter 1** 

# PRELIMINARIES

# **1.1 Introduction and summary**

The field of infinitely divisible random variables has grown during the last few decades to take a permanent position in the theory of probability. Almost all standard text books on probability now include at least one chapter devoted to the field of infinite divisibility. This is mainly due to its importance in solving the general central limit problem and to its applications to stochastic processes with stationary independent increments.

The notion of infinite divisibility originated in connection with the central limit problem. The classical central limit problem is concerned with characterizing the random variables X which can be obtained as

$$\sum_{k=1}^{n} a_n X_k + b_n \xrightarrow{w} X \text{ as } n \to \infty,$$

where  $(X_k)$  is a sequence of identically distributed random variables with finite variance a and expectation b, and  $a_n = (na)^{-\frac{1}{2}}$  and  $b_n = -nb(a_n)^{-\frac{1}{2}}$ . It turns out (cf. Loève (1977)) that X must be a normally distributed random variable with mean zero and variance one. Natural generalizations of the classical central limit problem (and thus of the normal distribution) is to drop the condition that the  $X_k$ 's have finite mean and variance (these limit random variables are called stable) and to consider limits where  $(X_k)$  does not necessarily have finite mean and variance and are not necessarily identically distributed (these limit random variables are called self-decomposable). Infinitely divisible random variables generalize the stable and self-decomposable ones as they are defined as the solution of the following (general) central limit problem:

$$\sum_{k=1}^{n} X_{k,n} \xrightarrow{w} X \text{ as } n \to \infty,$$

with some conditions on  $(a_n)$ ,  $(b_n)$  and  $(X_{k,n})$ .

Though practical applications have started to appear (cf. for example Ahmad and Abouammoh (1977), Thorin (1977), Carasso (1987), Keilson and Servi (1987) and Hansen and Willekens (1989)), infinite divisibility is (still) mainly of theoretical interest. In this monograph the emphasis is on the theory. Examples and practical applications are, as a rule, not sought.

Our starting point is the Lévy canonical representation of infinitely divisible distributions (cf. Theorem 1.3.2 of this monograph), where the characteristic function of an infinitely divisible distribution function F is related with a function M, called the Lévy spectral function. We are interested in characterizing the distributions F which have a Lévy spectral function M satisfying some monotonicity requirement.

In Chapter 2 we give a review of known monotonicity results in infinite divisibility and we present a curious connection with analogous results in renewal theory. In Chapters 3 and 4 we consider non-negative infinitely divisible random variables whose Lévy spectral functions are either absolutely continuous or supported by the non-negative integers. Chapter 3, which is based on the article On Moment Sequences and Infinitely Divisible Sequences, Hansen and Steutel (1988), studies these Lévy spectral functions in the context of moment sequences and moment functions. The results of On Logconcave and Logconvex Infinitely Divisible Sequences and log-concave functions is characterized in Chapter 5. Chapter 6 deviates from the theme of this monograph, as it studies infinitely divisible random variables with  $\alpha$ -unimodal Lévy spectral functions as limits of sums of strunken random variables.

#### **1.2 Notations and conventions**

In this section we list notations and conventions, which will be used throughout this monograph, often without further reference.

Let  $\mathbb{R} := (-\infty, \infty)$ ,  $\mathbb{R}_+ := [0, \infty)$ ,  $\mathbb{R}_- := (-\infty, 0]$ ,  $\mathbb{N}_0 := \{0, 1, ...\}$  and  $\mathbb{N}_+ := \{1, 2, ...\}$ . Random variables will always be one-dimensional and real-valued. They will be denoted by the capitals X, Y, Z, .... The distribution functions of X and Y will be denoted by F and G; their densities, if they exist, by f and g; their characteristic functions by  $\phi$  and  $\gamma$ , with  $\phi$  defined by

$$\phi(t) = \int_{-\infty}^{\infty} e^{itx} dF(x), \quad t \in \mathbb{R}.$$

If the random variables X and Y are non-negative then we denote their Laplace-

#### 1.2 Notations and conventions

Stieltjes transforms by f and 
$$\hat{g}$$
, where f is given by

$$\hat{f}(\tau) = \int_{\mathbb{R}_+} e^{-\tau x} dF(x), \ \tau \in \mathbb{R}_+.$$

If the random variables are non-negative and concentrated on the non-negative integers  $\mathbb{N}_0$ , then we call their distributions discrete and denote them by  $(p_n)_{n=0}^{\infty}$ ,  $(q_n)_{n=0}^{\infty}$ ,  $(r_n)_{n=0}^{\infty}$ , .... Their probability generating functions will be denoted by P, Q, R, ..., with P defined by

$$P(z) = \sum_{n=0}^{\infty} p_n z^n , |z| \le 1$$
.

In general, we shall denote the generating function of a sequence  $(a_n)_{n=0}^{\infty}$  by the corresponding capital letter A, where

$$A(z) = \sum_{n=0}^{\infty} a_n z^n , |z| \le r ,$$

for some  $r \in \mathbb{R}_+$ . All sequences considered in this monograph will be real-valued and indexed by  $\mathbb{N}_0$ , and henceforth denoted by  $(a_n), (b_n)$  etc..

A classification of a set C is a class  $\{C_t \mid t \in I\}$  with  $I \subseteq \mathbb{R}$  an index set, such that the sets  $C_t$  are non-decreasing, i.e., for t(1) < t(2) < ... < t(n) with  $t(i) \in I$ , we have that  $C_{t(1)} \subseteq C_{t(2)} \subseteq ... \subseteq C_{t(n)} \subseteq C$ .

A set C of characteristic functions is said to be closed under limits if any characteristic function which is the limit of a sequence of characteristic functions in C is itself a member of C.

The letters M and N will always denote Lévy spectral functions (cf. Theorem 1.3.2 of Section 1.3) and H will always be a canonical measure (cf. Theorem 1.3.3). The sequences  $(p_n)$  and  $(r_n)$  will always be assumed to be related through equation (1.2) of Theorem 1.3.4.

#### **1.3 Infinitely divisible distributions**

We begin this section with a definition of an infinitely divisible random variable (cf. Lukacs (1970), p. 107). This definition is equivalent to the one given in the introduction (cf. Theorem 1.3.11).

DEFINITION 1.3.1. A random variable X is said to be infinitely divisible if for every positive integer n, there exists independent and identically distributed random variables  $X_{1n}, X_{2n}, ..., X_{nn}$  such that

$$X = X_{1n} + X_{2n} + \dots + X_{nn}$$
,

d

with = denoting equality in distribution.

We say that  $\phi$ ,  $\hat{f}$ , P or F is infinitely divisible if it stems from an infinitely divisible random variable. We state three representation theorems for infinitely divisible distributions on  $\mathbb{R}$ ,  $\mathbb{R}_+$  and  $\mathbb{N}_0$ . The proof of the first theorem may be found as Theorem 5.5.2 in Lukacs (1970), of the second partly in Feller (1971), p. 450, Theorem 2 and partly in Steutel (1970), p. 86, Theorem 4.2.4 and the proof of the last theorem in Feller (1968), p. 290 and Steutel (1970), p. 83, Corollary 4.2.1.

THEOREM 1.3.2. A function  $\phi$  is an infinitely divisible characteristic function if and only if it can be written in the form

$$\ln \phi(t) = ita_{\phi} - \frac{1}{2}\sigma_{\phi}^2 t^2 + \int_{\mathbb{R}\setminus\{0\}} k(t, x) dM(x) ,$$

where  $a_{\phi} \in \mathbb{R}$ ,  $\sigma_{\phi}^2 \in \mathbb{R}_+$ ,  $k(t, x) = e^{itx} - 1 - itx (1 + x^2)^{-1}$ , and such that the function *M* (called the Lévy spectral function) satisfies

- (i) M(x) is non-decreasing on  $(-\infty, 0)$  and  $(0, \infty)$ ;
- (ii)  $M(-\infty) = M(\infty) = 0;$ (iii) The integrals  $\int_{-1}^{0} x^2 dM(x)$  and  $\int_{0}^{1} x^2 dM(x)$  are finite.

The representation is unique.

THEOREM 1.3.3. A function  $\hat{f}$  is an infinitely divisible Laplace-Stieltjes transform if and only if it can be written in the form

$$\ln \hat{f}(\tau) = \int_{0}^{\infty} (e^{-\tau x} - 1) x^{-1} dH(x),$$

where the function H (called the canonical measure) is non-decreasing. Equivalently,  $\hat{f}$  is infinitely divisible if and only if its distribution function F satisfies

$$\int_{0}^{x} u \, dF(u) = \int_{0}^{x} F(x-u) \, dH(u) \, , \, x \in \mathbb{R}_{+} \, .$$
(1.1)

Necessarily  $\int_{1}^{\infty} x^{-1} dH(x) < \infty$ . The representations are unique.

THEOREM 1.3.4. A function P with P(0) > 0 is an infinitely divisible probability generating function if and only if it can be written in the form

$$\ln P(z) = -\theta \left(1 - G(z)\right) ,$$

#### 1.3 Infinitely divisible distributions

where  $\theta > 0$  and G is a probability generating function with distribution  $(g_n)$  such that G(0)=0. Equivalently, P is infinitely divisible with P(0)>0 if and only if its distribution  $(p_n)$  satisfies  $p_0 > 0$  and

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_{n-k} r_k , n \in \mathbb{N}_0 , \qquad (1.2)$$

with  $(r_n)$  (the canonical measure of  $(p_n)$ ) non-negative; necessarily  $r_n = \theta(n+1)g_{n+1}$ and  $\sum_{k=0}^{\infty} r_k/(k+1) = \theta < \infty$ . The representations are unique.

Relations between the function M in Theorem 1.3.2, the function H in Theorem 1.3.3 and the sequence  $(r_n)$  in Theorem 1.3.4 are derived in van Harn (1978), Section 1.7. Equations (1.1) and (1.2) are derived from the canonical representation of the integral transform of the corresponding infinitely divisible distribution by differentiation (and then inverting the integral transforms). Since the derivative of a characteristic function does not necessarily exist, we do not have a representation similar to (1.1) and (1.2) in Theorem 1.3.2. We will use the following notations throughout this monograph.

NOTATION 1.3.5. Let  $ID(\mathbb{R})$ ,  $ID(\mathbb{R}_+)$  and  $ID(\mathbb{N}_0)$  denote the set of infinitely divisible characteristic functions, Laplace-Stieltjes transforms and probability generating functions, respectively.

NOTATION 1.3.6. An infinitely divisible characteristic function  $\phi$  is uniquely determined by the triple  $[a_{\phi}, \sigma_{\phi}^2, M]$  in Theorem 1.3.2. An infinitely divisible characteristic function with Lévy spectral function M and constants  $a_{\phi}$  and  $\sigma_{\phi}^2$  (cf. Theorem 1.3.2) will therefore be denoted by  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$ .

The next two theorems give some useful properties and another characterization of ID(I),  $I \in \{\mathbb{R}, \mathbb{R}_+, \mathbb{N}_0\}$ . For a proof of the first theorem we refer to Theorems 5.3.2 and 5.3.3 in Lukacs (1970). The second theorem is a basic consequence of Definition 1.3.1.

THEOREM 1.3.7. For  $I = \mathbb{R}$ ,  $\mathbb{R}_+$  or  $\mathbb{N}_0$  the sets ID(I) (cf. Notation 1.3.5) are multiplication semigroups, closed under limits.

THEOREM 1.3.8. The following equivalences hold (cf. Notation 1.3.5)

(i)  $\phi \in ID(\mathbb{R})$  if and only if  $\phi^{1/n}$  is a characteristic function for all  $n \in \mathbb{N}_+$ ;

- (ii)  $\hat{f} \in ID(\mathbb{R}_+)$  if and only if  $\hat{f}^{1/n}$  is a Laplace-Stieltjes transform for all  $n \in \mathbb{N}_+$ ;
- (iii)  $P \in ID(\mathbb{N}_0)$  if and only if  $P^{1/n}$  is a probability generating function for all  $n \in \mathbb{N}_+$ .

Before continuing we need two definitions.

DEFINITION 1.3.9. A sequence  $(X_k)$  of random variables is said to be bounded if there exists a constant  $c \ge 0$  such that

$$\mathbb{P}(|X_k| \le c) = 1, \quad \text{for all } k \in \mathbb{N}_0.$$

DEFINITION 1.3.10. By a triangular array of random variables is meant a double sequence of random variables  $(X_{k,n})$ , k=1, 2, ..., n,  $n \in \mathbb{N}_+$  (hence forth denoted  $(X_{k,n})$ ), such that the random variables  $X_{1,n}, ..., X_{n,n}$  of the *n*-th row are mutually independent.

The triangular array  $(X_{k,n})$ , is said to be uniformly asymptotically negligible (uan) if  $X_{k,n} \to 0$  in probability, uniformly in k as  $n \to \infty$ , i.e., if for every  $\varepsilon > 0$ 

 $\lim_{n \to \infty} \sup_{1 \le k \le n} \mathbb{P}(|X_{k,n}| \ge \varepsilon) = 0.$ 

The triangular array of characteristic functions  $(\phi_{k,n})$  and the triangular array of distribution functions  $(F_{k,n})$  will be called uan if they stem from a uan triangular array of random variables. As mentioned in the introduction, the notion of infinitely divisibility originated in the context of the central limit theorem. We now state a theorem which shows that the set of infinitely divisible random variables is the solution of the so called general central limit theorem. For a proof we refer to Theorem A, p. 321 in Loève (1977).

THEOREM 1.3.11. A random variable X is infinitely divisible if and only if there exists a uan triangular array  $(X_{k,n})$ , k = 1, 2, ..., n,  $n \in \mathbb{N}_+$ , of random variables such that

$$\sum_{k=1}^n X_{k,n} \xrightarrow{w} X \text{ as } n \to \infty.$$

Here  $\rightarrow$  denotes weak convergence. The next theorem shows how the Lévy spectral function M and the constant  $\sigma_{\phi}$  in Theorem 1.3.2 are determined by the sums of the distribution functions of  $(X_{k,n})$ . This theorem is proved in Loève (1977) as Criterion B, p. 326.

THEOREM 1.3.12. Let X be a random variable with characteristic function  $\phi$  and let  $(X_{k,n}), k = 1, 2, ..., n, n \in \mathbb{N}_+$  be a uan triangular array of random variables with distribution functions  $(F_{k,n}), k = 1, 2, ..., n, n \in \mathbb{N}_+$ . There exists a sequence  $(b_n)$  such that

$$\sum_{k=1}^{n} X_{k,n} + b_n \xrightarrow{w} X \text{ as } n \to \infty,$$

if and only if

(i) there exists a function M satisfying (i) and (iii) of Theorem 1.3.2 such that

 $M_n := \sum_{k=1}^n F_{k,n} \xrightarrow{w} M \text{ as } n \to \infty$ 

outside every neighbourhood of the origin.

(ii)  $\lim_{\varepsilon \to 0} \lim_{n \to \infty} \sum_{k=1}^{n} \left[ \int_{|x| \le \varepsilon} x^2 dF_{k,n}(x) - \left( \int_{|x| \le \varepsilon} x dF_{k,n}(x) \right)^2 \right] = \sigma_{\phi}^2.$ 

Necessarily  $\phi$  is infinitely divisible with  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$  for some  $a_{\phi} \in \mathbb{R}$ .

We finish this section with two lemmas, which will be used in Chapter 6 and the Appendix. For a proof we refer to Loève (1977), p. 314, Theorem A.

LEMMA 1.3.13. The triangular array of characteristic functions  $(\phi_{k,n})$ , k = 1, 2, ..., n,  $n \in \mathbb{N}_+$ , is uan if and only if

 $\lim_{n\to\infty}\sup_{1\le k\le n}|\phi_{k,n}(t)-1|=0,$ 

uniformly on every finite interval.

LEMMA 1.3.14. If  $(X_{k,n}^{(1)})$  and  $(X_{k,n}^{(2)})$  are uan triangular arrays, then  $(X_{k,n}^{(1)} + X_{k,n}^{(2)})$  is a uan triangular array.

# 1.4 Self-decomposable and stable distributions

The sets of self-decomposable and stable distributions are two important subsets of the set of infinitely divisible distributions. Stable distributions are widely studied (cf. Lukacs (1970)) as they provide a natural generalization of the normal distribution; self-decomposable distributions are, in turn, a generalization of the stable distributions. In Chapters 5 and 6 we generalize both concepts.

A random variable X is called stable if there exists sequences  $(a_n)$  and  $(b_n)$  of real numbers with  $a_n \ge 0$  and a sequence  $(X_k)$  of independent and identically distributed random variables such that  $(X_{k,n})$  with  $X_{k,n} = a_n X_k$  is uan and

$$\sum_{k=1}^{n} a_n X_k + b_n \xrightarrow{w} X \text{ as } n \xrightarrow{w} , \qquad (1.3)$$

If  $(X_k)$  are not necessarily identically distributed then X is called self-decomposable. The uan condition on  $(a_nX_k)$  implies that  $a_n \to 0$  and  $a_{n+1}/a_n \to 1$  as  $n \to \infty$ . Let the linear operator  $T_t$  be defined by  $T_t x = tx$ . Then (1.3) can be rewritten as

$$\sum_{k=1}^{n} T_{a_{k}} X_{k} + b_{n} \xrightarrow{w} X \quad \text{as } n \to \infty .$$
(1.4)

In Chapter 6 and in the Appendix, we consider limits of the form (1.4), with  $T_{a_n}$  replaced by a more general operator  $U_{a_n}$ . Self-decomposability of X is equivalent to

$$\mathbf{X} = c\mathbf{X}' + \mathbf{X}_c \,, \tag{1.5}$$

for all  $c \in (0,1)$ , where X' and X<sub>c</sub> are independent and X' is distributed as X. For the corresponding characteristic functions this means that for every  $c \in (0,1)$  there exists a characteristic function  $\phi_c$  such that

$$\phi(t) = \phi(ct) \phi_c(t) \quad , \ t \in \mathbb{R} . \tag{1.6}$$

The above results can be derived from Loève (1977), Section 24. If  $\phi_c(t) = \phi((1-c^{\delta})^{1/\delta} t) e^{iat}$  for some  $a \in \mathbb{R}$  and  $\delta \in (0, 2]$ , then  $\phi$  is stable (cf. the proof of Theorem 5.7.2, Lukacs (1970)). If X is non-negative, then we call  $\hat{f}$  self-decomposable (on  $\mathbb{R}_+$ ) if

 $\hat{f}(\tau) = \hat{f}(c\tau) \hat{f}_c(\tau) , \ \tau \in \mathbb{R}_+ ,$ 

d

and stable (on  $\mathbb{R}_+$ ) if  $\hat{f}_c(\tau) = \hat{f}((1-c^{\delta})^{1/\delta} \tau)$ . Steutel and van Harn (1979) proposed a discrete analogue of self-decomposability and stability. A random variable X is said to be discrete self-decomposable if for every  $c \in (0,1)$  there exists a random variable  $X_c$ , independent of X' such that

$$\mathbf{X} = c \otimes \mathbf{X}' + \mathbf{X}_c \,, \tag{1.7}$$

with X' and X identically distributed. The random variable  $c \otimes X$  is defined in distribution by

1.4 Self-decomposable and stable distributions

$$P_{c \otimes X}(z) = P_X(1 - c(1 - z)), |z| \le 1,$$

with  $P_X$  denoting the probability generating function of X. For a probabilistic interpretation of  $\otimes$  see Steutel and van Harn (1979). In terms of probability generating functions (1.7) reads

$$P(z) = P(1 - c(1 - z))P_c(z), |z| \le 1, c \in (0, 1),$$

for some probability generating functions  $P_c$ . If  $P_c(z) = P(1-(1-c^{\delta})^{1/\delta}(1-z))$  for some  $\delta > 0$ , then P is said to be (discrete) stable.

We now give another notation and list a series of representation theorems for self-decomposable and stable distributions. The proof of Theorem 1.4.1 can be found in Lukacs (1970), Section 5.7. This proof contains a minor error, which is corrected in Hall (1981). The proof of Theorem 1.4.5 is given in Lukacs (1970), Section 5.11. For the proofs for distributions on  $\mathbb{R}_+$  we refer to Feller (1971), Section XIII.6 and those for distributions on  $\mathbb{N}_0$  to Steutel and van Harn (1979).

THEOREM 1.4.1. A function  $\phi$  is the characteristic function of a stable distribution if and only if  $\phi$  is either normal or  $\phi$  can be written in the form

 $\ln \phi(t) = ita_{\phi} - c |t|^{\delta} (1 + i\beta \operatorname{sgn}(t) w (|t|, \delta)),$ 

where  $c \ge 0$ ,  $|\beta| \le 1$ ,  $\delta \in (0,2)$  and  $a_{\phi} \in \mathbb{R}$ . The function  $w(|t|,\delta)$  is given by

 $w(|t|,\delta) = \begin{cases} \tan(\pi \delta/2) & \text{if } \delta \neq 1\\ -(2/\pi)\ln|t| & \text{if } \delta = 1 \end{cases}.$ 

Equivalently,  $\phi$  is the characteristic function of a stable distribution if and only if  $\phi$  is infinitely divisible and either (cf. Theorem 1.3.2)  $\sigma_{\phi}^2 > 0$  and  $M(x) \equiv 0$  or  $\sigma_{\phi}^2 = 0$  and  $M(x) = C_1 |x|^{-\delta}$  for x < 0 and  $M(x) = -C_2 x^{-\delta}$  for x > 0. The parameters satisfy  $\delta \in (0,2), C_1 \ge 0, C_2 \ge 0$  and  $C_1 + C_2 \ge 0$ . The parameter  $\delta$  is called the exponent of stability of  $\phi$ .

THEOREM 1.4.2. A function  $\hat{f}$  is the Laplace-Stieltjes transform of a stable distribution on  $\mathbb{R}_+$  if and only if it can be written in the form

 $\ln \hat{f}(\tau) = -\lambda \tau^{\delta}$ ,  $\tau \in \mathbb{R}_+$ ,

with  $\lambda \ge 0$  and  $\delta \in (0,1]$ . The parameter  $\delta$  is called the exponent of stability of f.

THEOREM 1.4.3. A function P is the probability generating function of a stable distribution on  $\mathbb{N}_0$  if and only if it can be written in the form

$$\ln P(z) = -\lambda(1-z)^{\delta}, |z| \leq 1,$$

with  $\lambda \ge 0$  and  $\delta \in (0,1]$ . The parameter  $\delta$  is called the exponent of stability of P.

The degenerate distribution is trivially self-decomposable and stable. It is therefore usual in the literature not to call the degenerate distribution a stable distribution. For our purposes it is however desirable to include the degenerate distributions in the set of stable distributions. We therefore introduce the following notation.

NOTATION 1.4.4. The characteristic function of a stable, possibly degenerate, distribution with exponent  $\delta$ , will be denoted by  $\phi_{\text{STABLE}(\delta)}$ . Similarly, we denote by  $P_{\text{STABLE}(\delta)}$  and  $\hat{f}_{\text{STABLE}(\delta)}$  the probability generating function and the Laplace-Stieltjes transform of a stable, possibly degenerate, distribution with exponent  $\delta$ .

THEOREM 1.4.5. A function  $\phi$  is the characteristic function of a self-decomposable distribution if and only if  $\phi$  is infinitely divisible with Lévy spectral function M (cf. Theorem 1.3.2) having left and right derivatives and such that |x|M'(x) is non-decreasing on  $(-\infty, 0)$  and non-increasing on  $(0, \infty)$ .

THEOREM 1.4.6. A function P is the probability generating function of a selfdecomposable distribution if and only if it can be written in the form

$$\ln P(z) = \int_{0}^{1} \ln Q (1 - v (1 - z)) v^{-1} dv ,$$

with Q a unique infinitely divisible probability generating function. Equivalently, P is a self-decomposable probability generating function if and only if it is infinitely divisible and its canonical measure  $(r_n)$  (cf. Theorem 1.3.4) is non-increasing.

The analogue of Theorem 1.4.6 for distributions on  $\mathbb{R}_+$  is proved in van Harn et. al. (1982) and mentioned for distributions on  $\mathbb{R}$  in Steutel and van Harn (1979). In Chapter 5 we prove two theorems which include these analogues. We finish this section with a definition, which we use in Chapters 5 and 6.

DEFINITION 1.4.7. The characteristic function  $\phi$  is said to be in the domain of normal attraction of a stable characteristic function with exponent  $\delta$  if and only if for suitable  $(b_n)$ ,  $\lim_{n \to \infty} e^{ib_n t} \phi^n((1/n)^{1/\delta} t) = \phi_{\text{STABLE}(\delta)}(t)$ .

10

# 1.5 Motivation and methodology

In this monograph we study the relationship between an infinitely divisible distribution and its Lévy spectral function (or canonical measure). We prove that an infinitely divisible distribution whose Lévy spectral function (or canonical measure) possesses some (well-known) monotonicity property, for example complete monotonicity, log-concavity, log-convexity or  $\alpha$ -unimodality, belongs to a (well-known) set of distributions, for example the set of mixtures of geometric distributions, strongly unimodal distributions or  $\alpha$ -self-decomposable distributions.

It turns out that in studying monotonicity properties in this context, it is easiest to first consider the discrete case (cf. Theorem 1.3.4). Here the canonical measure can be explicitly expressed in terms of the probabilities, which makes developing and proving hypotheses and constructing counterexamples easier than in the case of distributions on  $\mathbb{R}$  or  $\mathbb{R}_+$ . In many cases we can even prove the equivalent result for distributions on  $\mathbb{R}_+$  from those on  $\mathbb{N}_0$  by applying some simple argument (cf. Sections 3.5, 4.3 and 5.5). Also, by studying the discrete case we develop insight which can be helpful in proving the result for  $\mathbb{R}$  (cf. Section 5.4).

As is seen in Chapter 2, many of the monotonicity results obtained in renewal theory and infinite divisibility are quite similar. This observation led us to consider log-convexity and moment sequences in the context of infinite divisibility (cf. Theorems 2.4.2 and 2.4.3 of the next chapter).

Chapter 2

# INFINITELY DIVISIBLE SEQUENCES AND RENEWAL SEQUENCES

### 2.1 Introduction

Much of renewal theory is concerned with determining properties of the renewal function. This includes study of the relationship between the renewal function and its underlying distribution. Similarly, the interplay between an infinitely divisible distribution and its Lévy spectral function plays an important role in the field of infinite divisibility. Many of the results obtained in these two, very different, fields are quite similar. This correspondence, between results in infinite divisibility and in renewal theory, proves to be very useful (cf. Chapters 3 and 4). In this chapter we give a brief review of the results concerning monotonicity properties in these two fields. Section two of this chapter gives a short introduction to renewal theory. The following two sections state the results, with little mention of possible applications. For a more complete description we refer to the references. The last section discusses the interplay between renewal sequences and infinitely divisible sequences.

## 2.2 A taste of renewal theory

In this section we give a brief introduction to renewal theory. For simplicity we restrict ourselves to renewal sequences on  $\mathbb{N}_0$ . For a more rigorous introduction to renewal theory we refer to Feller (1968). Let E be an event (for example a renewal) and define the probability distribution  $(f_n)$  by

 $f_n := \mathbb{P}(\mathbb{E} \text{ occurs for the first time at time } n+1), n \in \mathbb{N}_0$ .

Define the sequence of probabilities  $(u_n)$  (called the renewal sequence with underlying

### 2.2 A taste of renewal theory

distribution  $(f_n)$  by

 $u_n := \mathbb{P}(\text{E occurs at time } n), u_0 = 1, n \in \mathbb{N}_+$ .

It then follows that  $(f_n)$  and  $(u_n)$  are related by

$$u_{n+1} = \sum_{k=0}^{n} u_{n-k} f_k , u_0 = 1, n \in \mathbb{N}_0 .$$
(2.1)

It is usual in the literature (cf. for example Feller (1968)) to define  $f_n$  as the probability that E occurs for the first time at time n, with  $f_0=0$ . We prefer our approach because it results in a recurrence relation which, in appearance, resembles equation (1.2). The probability  $f_n$  can be interpreted as the probability that a machine first breaks down at time n+1 given that it broke down at time zero. If a broken down machine gets fixed instantaneously, and after reparation is 'as good as new', then  $u_n$  gives the probability that the machine breaks down at time n. In many practical situations the distribution  $(f_n)$  is known, or at least some property (for example a monotonicity property) is known, and the sequence  $(u_n)$  or its behaviour in some sense, is sought. One of the most important results from renewal theory is:

if 
$$\mu := \sum_{n=0}^{\infty} (n+1) f_n < \infty$$
 and  $(f_n)$  is aperiodic, then  $\lim_{n \to \infty} u_n = \mu^{-1}$ .

In the following section we give a review of monotonicity results in renewal theory.

# 2.3 Monotonicity results in renewal theory

To avoid repetition, we only review the results in discrete renewal theory, i.e., on sequences related through (2.1). The analogous results for general renewal functions, related through the so-called renewal equation (cf. Ross (1983)), are also true. In fact in many cases the result for renewal functions can be obtained by applying a limiting argument to the result on renewal sequences (cf. Hansen and Frenk (1989)). The list is by no means complete and will be given with very few comments to applications.

Kaluza (1928) was the first (to the author's knowledge) to study sequences related through (2.1). Among other results he proved

THEOREM 2.3.1. Let the sequences  $(u_n)$  and  $(f_n)$  be related by (2.1). The following implications hold.

- (i) If  $f_n \ge 0$ ,  $n \in \mathbb{N}_0$  then  $u_n \ge 0$ ,  $n \in \mathbb{N}_0$ ;
- (ii) If  $(u_n)$  is log-convex then  $f_n \ge 0$ ,  $n \in \mathbb{N}_0$ .

The definitions of log-convexity and log-concavity are given in Section 4.2. In de Bruijn and Erdös (1953), for part (i), and in Hansen and Frenk (1989), for parts (ii) and (iii), we find the following theorem.

THEOREM 2.3.2. Let the sequences  $(u_n)$  and  $(f_n)$  be related by (2.1). Suppose  $\sum f_n \le 1$ and  $\sum (n+1)f_n = \mu < \infty$ . Let

$$F(n) := \sum_{k=0}^{n} f_k, F_1(n) := \mu^{-1} \sum_{k=0}^{n} (1-F(k)).$$

The following implications hold.

- (i) If  $(f_n)$  is log-convex then  $(u_n)$  is log-convex;
- (ii) If (1 F(n)) is log-convex then  $(u_n)$  is non-increasing;
- (iii) If  $(1-F_1(n))$  is log-convex then  $u_n \ge \mu^{-1}$ ,  $n \in \mathbb{N}_0$ .

For condition (i) in Theorem 2.3.2 it is only necessary to assume that  $f_n \ge 0$  for  $n \in \mathbb{N}_0$ . The analogue of Theorem 2.3.2, parts (ii) and (iii), for distributions on  $\mathbb{R}_+$  was first proved in Brown (1980), by coupling methods. Hansen and Frenk (1989) provides a simpler proof by applying a limiting argument to Theorem 2.3.2. The conditions on  $(f_n)$  given in (ii) and (iii) are related to distributions with decreasing failure rates and increasing mean residual life-times. Note that

 $(f_n)$  log-convex  $\Rightarrow (1 - F(n))$  log-convex  $\Rightarrow (1 - \mu^{-1}F_1(n))$  log-convex.

 $(u_n)$  log-convex  $\Rightarrow$   $(u_n)$  non-increasing  $\Rightarrow$   $u_n \ge \mu^{-1}$ ,  $n \in \mathbb{N}_0$ .

Horn (1970) studied (2.1) in the context of moment sequences. For the definition of moment sequences we refer to Notation 3.2.1. Horn (1970) dropped the condition that the sequences are non-negative and convergent.

THEOREM 2.3.3. Let the sequences  $(u_n)$  and  $(f_n)$  be related by (2.1). Then

- (i)  $(u_{n+1})$  is a Hamburger moment sequence on  $\mathbb{R}$  if and only if  $(f_n)$  is a Hamburger moment sequence on  $\mathbb{R}$ ;
- (ii)  $(u_n)$  is a Stieltjes moment sequence on  $\mathbb{R}_+$  if and only if  $(f_n)$  is a Stieltjes moment sequence on  $\mathbb{R}_+$ ;
- (iii)  $(u_n)$  is a Hausdorff moment sequence on [0, 1] if and only if  $(f_n)$  is a Hausdorff moment sequence on [0, 1] with  $\sum f_n \leq 1$ .

14

Kaluza (1928) proved part (ii) and the 'if' part of (iii), with a proof different from Horn's (1970). Theorem 2.3.3 is stronger then Theorem 2.3.2 in the sense that

 $(a_n)$  Hausdorff moment sequence  $\Rightarrow$   $(a_n)$  Stieltjes moment sequence  $\Rightarrow$   $(a_n)$  log-convex.

This is easily verified by using Schwarz' inequality. Part (i) of the following, and last, theorem of this section is proved in Kingman (1972), p. 7, Theorem 1.4. The other two parts are easily proved.

THEOREM 2.3.4. Let the sequences  $(u_n)$  and  $(f_n)$  be related by (2.1). Then.

- (i)  $u_{n+m} \ge u_n u_m$  for  $n \in \mathbb{N}_0$ ,  $m \in \mathbb{N}_0$ ;
- (ii) If  $u_0 u_1 > 0$  then  $u_n > 0$ ,  $n \in \mathbb{N}_0$ ;
- (iii) If  $f_0 \in [0,1]$  and  $f_{n+1}/f_n \le 1-f_0$  then  $(u_n)$  is non-increasing.

# 2.4 Monotonicity results of infinitely divisible distributions

As in Section 2.3 we only review the results for discrete distributions. Most results are here also true for distributions on  $\mathbb{R}_+$  and can usually be obtained by some limiting argument (cf. Chapter 4) or some other argument (cf. Chapters 3 and 5) from those on  $\mathbb{N}_0$ . The analogues of Theorems 2.3.2 and 2.3.3 are proved in Chapters 4 and 3, respectively. For completeness we also state them here.

THEOREM 2.4.1. Let the sequences  $(p_n)$  and  $(r_n)$  be related by (1.2), i.e., by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_{n-k} r_k , p_0 \ge 0, n \in \mathbb{N}_0 .$$
(2.2)

The following implications hold.

- (i) If  $r_n \ge 0$ ,  $n \in \mathbb{N}_0$  then  $p_n \ge 0$ ,  $n \in \mathbb{N}_0$ ;
- (ii) If  $(p_n)$  is log-convex then  $r_n \ge 0$ ,  $n \in \mathbb{N}_0$ .

Part (ii) of Theorem 2.4.1 was first proved in Steutel (1970), Theorem 4.2.2. For a different proof see Section 4.2. Part (ii) states that all log-convex distributions are infinitely divisible. The absolutely continuous analogue of Theorem 2.4.1 (ii) is proved as Theorem 4.2.6, p. 89 in Steutel (1970). The following theorem and its absolutely continuous analogue is proved in Chapter 4. THEOREM 2.4.2. Let the sequences  $(p_n)$  and  $(r_n)$  be related by (2.2). The following implications hold.

- (i) If  $(r_n)$  is log-convex and  $r_0^2 \le r_1$  then  $(p_n)$  is log-convex;
- (ii) If  $(r_n)$  is log-concave and  $r_0^2 \ge r_1$  then  $(p_n)$  is log-concave.

In Chapter 3 we prove the analogue of Theorem 2.3.3 for infinitely divisible sequences, i.e., sequences  $(p_n)$  and  $(r_n)$  related by (2.2). This analogue also provides an alternative proof of Theorem 2.3.3 (cf. Section 3.4).

THEOREM 2.4.3. Let the sequences  $(p_n)$  and  $(r_n)$  be related by (2.2). Then

- (i)  $(p_{n+1})$  is a Hamburger moment sequence on  $\mathbb{R}$  if and only if  $(r_n/(n+1))$  is a Hamburger moment sequence on  $\mathbb{R}$  with  $\mu \leq \lambda$  (cf. (3.6));
- (ii)  $(p_n)$  is a Stieltjes moment sequence on  $\mathbb{R}_+$  if and only if  $(r_n/(n+1))$  is a Stieltjes moment sequence on  $\mathbb{R}_+$  with  $\mu \le \lambda$  (cf. (3.6));
- (iii)  $(p_n)$  is a Hausdorff moment sequence on [0, 1] if and only if  $(r_n/(n+1))$  is a Hausdorff moment sequence on [0, 1] with  $\mu \leq \lambda$  (cf. (3.6)).

The absolutely continuous analogue of part (iii) is proved in Steutel (1970), Theorem 2.12.1. Parts (iii) and (iv) of the following theorem are proved in Steutel and van Harn (1979). The first part is due to F.W. Steutel (private communication) and can be proved as Theorem 1.4, p.7 in Kingman (1972) is proved. The second part is proved in Steutel (1970), Theorem 4.2.3.

THEOREM 2.4.4. Let the sequences  $(p_n)$  and  $(r_n)$  be related by (2.2). Then

- (i)  $\binom{m+n}{m}p_{n+m} \ge p_n p_m$  for  $n \in \mathbb{N}_0$ ,  $m \in \mathbb{N}_0$ ;
- (ii) If  $p_0 p_1 > 0$  then  $p_n > 0$ ,  $n \in \mathbb{N}_0$ ;
- (iii) If  $r_0 \in [0,1]$  and  $(r_n)$  is non-increasing then  $(p_n)$  is non-increasing;
- (iv) If  $(r_n)$  is non-increasing then  $(p_n)$  is unimodal.

Condition (iv) implies that all discrete self-decomposable distributions are unimodal (cf. Theorem 1.4.6).

16

#### 2.5 Renewal sequences and infinitely divisible sequences

## 2.5 Renewal sequences and infinitely divisible sequences

As seen in the two previous sections there is a connection between the behaviour of sequences related through (2.1) and sequences related through (2.2). This section is intended to give some understanding to why this is true.

Let  $(p_n)$  be a sequence of real numbers and define the two sequences  $(r_n)$  and  $(f_n)$  through

$$p_{n+1} = \sum_{k=0}^{n} p_{n-k} f_k , \ n \in \mathbb{N}_0 ,$$
(2.3)

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_{n-k} r_k , \ n \in \mathbb{N}_0 .$$
(2.4)

Equation (2.4) is more general than equation (2.3) in the sense that if the sequence  $(f_n)$  is non-negative, then  $(r_n)$  is also non-negative (cf. Steutel (1970), p.83). In fact, if  $(p_n)$  is a probability distribution and  $(f_n)$  is non-negative, then  $(p_n)$  is a compound geometric distribution and hence infinitely divisible.

Taking generating functions on both sides of (2.3) and (2.4) and eliminating P(z) yields

$$R(z) = -\frac{d}{dz} \ln(1 - zF(z)) .$$
(2.5)

Let  $f_{-1} \coloneqq -1$ . Equation (2.5) is equivalent to

$$(n+1)[(-1)^n f_n] = \sum_{k=0}^n \left[ (-1)^{n-k-1} f_{n-k-1} \right] \left[ (-1)^k r_k \right], \ n \in \mathbb{N}_0.$$
(2.6)

One could therefore expect that the relationship between  $(p_n)$  and  $(r_n)$  is similar to that between  $(f_n)$  and  $(r_n)$ . Hence, if  $(r_n)$  has some property which  $(p_n)$  inherits and  $((-1)^n r_n)$  also has this property, then  $((-1)^{n-1} f_{n-1})$  will also inherit this property. In a similar way one can start with  $(p_n)$ . Equation (2.6) does not only provide an idea as to why the theorems in Sections 2.3 and 2.4 are similar, but also provides alternative proofs of some of the results in Section 2.3 (see Chapter 3).

# **Chapter 3**

# MOMENT SEQUENCES AND MOMENT FUNCTIONS

# **3.1 Introduction**

In this chapter we consider sequences  $(p_n)$  and  $(r_n)$  related through equation (1.2), i.e., through

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_k r_{n-k} , \quad n \in \mathbb{N}_0 , \qquad (3.1)$$

where either  $(p_n)$  or  $(r_n/(n+1))$  is a moment sequence (cf. Notation 3.2.1), and we consider functions f and h related through (cf. (1.1))

$$xf(x) = \int_{0}^{x} f(x-u) h(u) du , \quad x \in \mathbb{R}_{+} , \qquad (3.2)$$

where either f or h(u)/u is a moment function (cf. Definition 3.5.1). The discrete case is considered in the same vein as renewal sequences were by Horn (1970) (cf. Theorems 2.3.3 and 2.4.3). The proofs in the discrete case are based on a theorem in monotone matrix function theory by Bendat and Sherman (1955). At first glance it may seem as though this approach uses too powerful tools, but its great advantage is that it results in some very elegant proofs. The somewhat more straightforward, but tedious, approach used to prove Theorems 3.3.2 and 3.3.7 can also be adapted to prove Theorems 3.3.4 and 3.3.8, but fails to prove Theorem 3.3.1 (cf. Remark 3.3.3).

In all but the last section of this chapter we will consider equation (3.1), and its absolutely continuous analogue (3.2), outside its probabilistic context; for (3.1) we drop the conditions that  $r_n \ge 0$  and  $\sum r_n/(n+1) < \infty$  (thus  $(p_n)$  need not be non-negative and  $\sum p_n$  need not be convergent) and take, for convenience,  $p_0 = 1$ ; for (3.2) we drop

#### 3.1 Introduction

the condition that  $\int_{1}^{\infty} h(x)/x \, dx < \infty$  (thus  $\int f \, dx$  need not be finite). In Section 3.2 we give some preliminaries. Statements (i), (ii) and (iii) of Theorem 2.4.3 will be proved as separate theorems in Section 3.3. The absolutely continuous analogue is proved in Section 3.5 using the results on infinitely divisible moment sequences. Using equation (2.6) and Theorem 2.4.3 we prove Theorem 2.3.3 in Section 3.4. Its absolutely continuous analogue is mentioned in Section 3.6. Our result on Hausdorff moment sequences has applications in probability theory, as sequences of this type are mixtures of geometric distributions. This case and its absolutely continuous analogue will be investigated in more detail in Section 3.7.

# **3.2 Preliminaries**

For ease in notation we introduce the following sets.

NOTATION 3.2.1. Let  $MS(\mathbb{R})$ ,  $MS(\mathbb{R}_+)$  and MS([0, 1]) denote the set of Hamburger, Stieltjes and Hausdorff moment sequences, respectively, i.e., for  $I \in \{\mathbb{R}, \mathbb{R}_+, [0, 1]\}$  let

$$(a_n) \in MS(I)$$
 if and only if  $a_n = \int_I x^n d\mu(x)$ ,  $n \in \mathbb{N}_0$ , (3.3)

where  $\mu$  is a nonnegative measure. Also let the sets  $MS^*(I)$  (cf. (3.3)) be given by

$$(a_n) \in MS^*(I)$$
 if and only if  $(a_n) \in MS(I)$  with  $\mu \le \lambda$ ; (3.4)

here  $\lambda$  denotes Lebesgue measure and  $\mu_1 \leq \mu_2$  means that  $\mu_1(B) \leq \mu_2(B)$  for all Borel sets *B*, i.e., that  $\mu_1$  is absolutely continuous with respect to  $\mu_2$  and the Radon-Nikodym derivative  $d\mu_1/d\mu_2 \leq 1$ . We also define (compare (3.3) and (3.4)) the set  $MS_T(I)$  by

$$(a_n) \in MS_T(I)$$
 if and only if  $(a_n) \in MS(I)$  and  $\mu$  is supported by  $[-T, T]$ . (3.5)

and let  $MS_T^*(I) = MS_T(I) \cap MS^*(I)$ . In a similar way we define  $MS(\mathbb{R}_{-})$  and  $MS^*(\mathbb{R}_{-})$ . The measure  $\mu$  in (3.3) is called the representing measure of the sequence  $(a_n)$ .

According to equation (2.1) the functions U and F are related by

$$zF(z) = 1 - \frac{1}{U(z)},$$
 (3.6)

and P and R, according to (3.1), by

$$\tilde{R}(z) := z \sum_{n=0}^{\infty} \frac{r_n}{n+1} z^n = \log P(z) .$$
(3.7)

### MOMENT SEQUENCES AND MOMENT FUNCTIONS

The symbols U, F, P and  $\tilde{R}$  in (3.6) and (3.7) can be regarded as formal power series. We only use them as functions when  $\mu$  in (3.3) is supported by [-T,T]. Then U, F, P and  $\tilde{R}$  are well-defined functions of the form

$$f(z) = \int_{[-T,T]} (1-xz)^{-1} d\mu(x) , z \in (-T^{-1}, T^{-1}) .$$

Hence they are Stieltjes transforms and so the measure  $\mu$  is unique.

As will become apparent, the essential difference between (3.6) and (3.7) in the context of moment sequences stems from the fact that

 $1 - \frac{1}{w}$  maps the upper half-plane onto the upper half-plane,

log w maps the upper half-plane onto the strip  $0 < \text{Im } w < \pi$ .

This difference explains the difference between Theorems 2.3.3 and 2.4.3 and somewhat complicates the proof of Theorem 2.4.3. We will use the following lemmas describing some properties of moment sequences. The proofs of the first two lemmas can be found in Shohat and Tamarkin (1943). The third lemma follows from Helly's first theorem and the corollary to Theorem 25.12, p. 292 in Billingsley (1979).

LEMMA 3.2.2.  $(a_n) \in MS(\mathbb{R}_+)$  if and only if  $(a_n) \in MS(\mathbb{R})$  and  $(a_{n+1}) \in MS(\mathbb{R})$ .

LEMMA 3.2.3.  $(a_n) \in MS([0, 1])$  if and only if  $(a_n) \in MS(\mathbb{R}_+)$  and  $a_n$  is bounded.

LEMMA 3.2.4. Let  $I \in \{\mathbb{R}, \mathbb{R}_+, [0, 1]\}$ . If  $(a_n(t)) \in MS(I)$  (or  $MS^*(I)$ ) for all  $t \ge t_0$ , and if  $\lim_{t \to \infty} a_n(t) = a_n$ ,  $n \in \mathbb{N}_0$ , then  $(a_n) \in MS(I)$  (or  $MS^*(I)$ ).

REMARK 3.2.5. In proving Theorem 2.4.3 we will use Lemma 3.2.4 together with truncation of integrals. If  $(p_n)$  is given by

$$p_n = \int_{\mathbb{R}} x^n \, d\mu(x) \,, \qquad n \in \mathbb{N}_0 \,,$$

and  $(r_n)$  is defined by (3.1), then we define  $(p_n(T))$  by

$$p_n(T) = \int_{|x| \le T} x^n d\mu(x), \quad n \in \mathbb{N}_0.$$

If we now define  $(r_n(T))$  from  $(p_n(T))$  by means of (3.1), then  $p_n(T) \to p_n$  and  $r_n(T) \to r_n$  as  $T \to \infty$ . In a similar way one can start from  $(r_n)$ .

Before we proceed we need a definition.

#### 20

DEFINITION 3.2.6. For T > 0, let  $A_T$  denote the set of real analytic functions on  $(-T^{-1}, T^{-1})$  that have an analytic continuation to the upper half-plane, and that are either constant or map the upper half-plane into itself;  $A_T^*$  denotes the subset of functions in  $A_T$  that map the upper half-plane into the strip  $0 < \text{Im } z < \pi$ .

The following lemma, proved in Bendat and Sherman (1955), is as basic here as it was in Horn (1970).

LEMMA 3.2.7. Let T > 0 and let C be a real-valued function defined on  $(-T^{-1}, T^{-1})$ . Then the following statements are equivalent.

- (i)  $C \in A_T$ ;
- (ii)  $C(x) = \sum_{0}^{\infty} c_n x^n \text{ for } x \in (-T^{-1}, T^{-1}) \text{ with } (c_{n+1}) \in MS_T(\mathbb{R});$
- (iii)  $C(z) = C(0) + \int_{[-T,T]} \frac{z}{1-tz} d\mu(t)$  (Im z > 0).

Lemma 3.2.7 shows that  $(c_{n+1})$  is a (truncated) Hamburger moment sequence if and only if its generating function (seen as a mapping) maps the upper half plane into itself. Hence U has this mapping property if and only if zF(z) does (cf. (3.6)). The next lemma enables us to use Lemma 3.2.7 in the situation of equation (3.7).

LEMMA 3.2.8. Let T > 0 and let  $\mu$  be a finite measure on [-T, T]. Let f be defined by

$$f(z) = \int_{[-T,T]} \frac{z}{1-zt} d\mu(t) \quad (\text{Im } z > 0) .$$
(3.8)

Then  $\mu \leq \lambda$  (cf. (3.4)) if and only if  $0 < \text{Im } f(z) < \pi$  for Im z > 0.

PROOF. Clearly, Im f(z) > 0 if Im z > 0. If  $\mu \le \lambda$  then for Im z > 0 we have

$$\operatorname{Im} f(z) \le \operatorname{Im} \int_{[-T,T]} \frac{z}{1-zt} dt = \arg(1+zT) - \arg(1-zT) < \pi .$$

Now suppose that  $0 < \text{Im } f(z) < \pi$ . The function

$$g(s) := -f(-(s+T)^{-1}) = \int_{[0,2T]} \frac{1}{s+t} d\mu(t-T)$$

is a Stieltjes transform. From the inversion formula for such transforms (cf. Widder (1946), p. 340) we have for any  $\xi_1$ ,  $\xi_2$  with  $0 < \xi_1 < \xi_2$ , and writing  $\tilde{\mu}(\xi) = (\mu(\xi + 0) + \mu(\xi - 0))/2$ ,

$$\tilde{\mu}(\xi_2) - \tilde{\mu}(\xi_1) = \lim_{\eta \downarrow 0} \frac{1}{\pi} \int_{\xi_1}^{\xi_2} \operatorname{Im} f(\frac{1}{s - i\eta}) ds \le \xi_2 - \xi_1$$

since  $\text{Im}(1/\overline{z}) > 0$  if and only if Im z > 0. It follows that  $\mu \le \lambda$ .

COROLLARY 1. Let f be given by (3.8) and let c > 0. Then  $0 < Im f(z) < c \pi$  for Im z > 0 if and only if  $\mu \le c \lambda$ .

Now from Lemmas 3.2.7 and 3.2.8 we obtain

LEMMA 3.2.9. Let T > 0 and let C be a real-valued function on  $(-T^{-1}, T^{-1})$ . Then the following three statements are equivalent.

- (i)  $C \in A_T^*$ ;
- (ii)  $C(x) = \sum_{n=0}^{\infty} c_n x^n \text{ for } x \in (-T^{-1}, T^{-1}) \text{ with } (c_{n+1}) \in MS_T^*(\mathbb{R});$
- (iii)  $C(x) = C(0) + \int_{[-T,T]} \frac{x}{1-xt} d\mu(t)$ ,

for a measure  $\mu \leq \lambda$ .

The next lemma is an analogue for our situation of Lemma B in Horn (1970). It reduces to this lemma if all \*'s are deleted. We will need both versions of the lemma, and we will refer to it as Lemma 3.2.10<sup>\*</sup> if we need it with the \*'s and as Lemma 3.2.10 otherwise. These lemmas are the key lemmas for the results of the next section, as they enable us to study moment sequences by considering the mapping properties of their generating functions. Here we use  $A_T^* \subset A_T$  and  $MS_T^*(I) \subset MS_T(I)$ , which follows from Notation 3.2.1 and Definition 3.2.6.

LEMMA 3.2.10.\* (3.2.10.) Let T > 0 and let  $(c_n)$  be a sequence of real numbers such that  $C(x) = \sum_{0}^{\infty} c_n x^n$  converges for all  $x \in (-T^{-1}, T^{-1})$ . Then (cf. Definition 3.2.6 and equation (3.5))

- (a)  $(c_n) \in MS_T^*(\mathbb{R})$  if and only if  $x C(x) \in A_T^*$ ;
- (b)  $(c_{n+1}) \in MS_T^*(\mathbb{R})$  if and only if  $C(x) \in A_T^*$ ;
- (c)  $(c_n) \in MS_T^*(\mathbb{R}_+)$  if and only if  $x C(x) \in A_T^*$  and  $C(x) \in A_T$ ;

(d) 
$$(c_{n+1}) \in MS_T^*(\mathbb{R}_+)$$
 if and only if  $C(x) \in A_T^*$  and  $x^{-1}(C(x) - C(0)) \in A_T$ .

PROOF. (a) and (b) follow directly from Lemma 3.2.9; the proofs of (c) and (d) are quite similar. We prove (c). First, let  $(c_n) \in MS_T^*(\mathbb{R}_+)$ . Then by (a) we have  $x C(x) \in A_T^*$  and by (c) of Lemma 3.2.10 that  $C(x) \in A_T$ . Conversely, if  $xC(x) \in A_T^*$ , then  $(c_n) \in MS_T^*(\mathbb{R})$  by (a), whereas  $C(x) \in A_T$  by (b) of Lemma 3.2.10 implies that  $(c_{n+1}) \in MS_T(\mathbb{R})$ . From Lemma 3.2.2 and the fact that  $MS_T^*(\mathbb{R}) \subset MS_T(\mathbb{R})$  we conclude that  $(c_n) \in MS_T(\mathbb{R}_+)$ . Finally, since moment sequences on  $MS_T(I)$  have unique representing measures we have  $(c_n) \in MS_T^*(\mathbb{R}) \cap MS_T(\mathbb{R}_+) = MS_T^*(\mathbb{R}_+)$ .

The following lemma is immediate from Lemma 3.2.2.

LEMMA 3.2.11.  $(c_n) \in MS_T^*(\mathbb{R}_+)$  if and only if  $(c_n) \in MS_T^*(\mathbb{R})$  and  $(c_{n+1}) \in MS(\mathbb{R})$ .

### 3.3 Moment sequences and infinitely divisible sequences

We are now ready to prove Theorem 2.4.3. We present the statements (i), (ii) and (iii) as separate theorems.

#### 3.3.1. Hamburger moment sequences

THEOREM 3.3.1. Let  $(p_n)$  and  $(r_n)$  be related by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_k r_{n-k} , \quad n \in \mathbb{N}_0 .$$

Then (cf. (3.3) and (3.4))

$$(p_{n+1}) \in MS(\mathbb{R})$$
 if and only if  $(\frac{r_n}{n+1}) \in MS^*(\mathbb{R})$ .

PROOF. In view of Lemma 3.2.4 and Remark 3.2.5 we only have to prove the equivalence for  $MS_T(\mathbb{R})$  and  $MS_T^*(\mathbb{R})$ . Since (cf. (3.7)) the generating functions of  $(p_n)$  and  $(r_n/(n+1))$  are related by  $\tilde{R}(z) = \log P(z)$ ), in view of Lemma 3.2.10\* (3.2.10) it is sufficient to prove that  $P \in A_T$  if and only if  $\log P \in A_T^*$  for some T' > 0. This last statement is true since (cf. Definition 3.2.6) P (is constant or) maps the upper half-plane into the strip  $0 < \operatorname{Im} z < \pi$ . Finally, P(z) is convergent for |z| < T with T > 0 if and only if  $\log P(z)$  converges for |z| < T' for some T' > 0.

The next theorem, of some interest in its own right, is a preparation for the proof of (ii) in Theorem 2.4.3. Its proof is similar to the proof of Theorem 2.12.1 in Steutel (1970) concerning mixtures of exponential distributions.

THEOREM 3.3.2. Let  $(p_n)$  and  $(r_n)$  be related as in Theorem 3.3.1. Then

$$(p_n) \in MS(\mathbb{R})$$
 if and only if  $\frac{r_n}{n+1} = b_n - (-1)^n c_n$  with  $(b_n), (c_n) \in MS^*(\mathbb{R}_+)$ .

PROOF. Any sequence in  $MS(\mathbb{R})$  can be approximated by sequences in  $MS_T(\mathbb{R})$  as follows. If  $(p_n) \in MS_T(\mathbb{R})$  has a discrete representing measure with positive atoms  $q_i$  at  $t_i$  (i = 1, 2, ..., N) where  $-T < t_N < ... < t_{l+1} < 0 < t_l < ... < t_1 < T$ , then P(x) takes the form

$$P(x) = \sum_{i=1}^{N} q_i / (1 - xt_i) = Q(x) / \prod_{i=1}^{N} (1 - xt_i),$$

with Q a polynomial of degree at most N-1. Observe that

 $P(x) < 0 \text{ for } x \downarrow t_k^{-1}, \text{ for } k = 1, 2, ..., l ,$   $P(x) > 0 \text{ for } x \uparrow t_k^{-1}, \text{ for } k = 1, 2, ..., l ,$   $P(x) > 0 \text{ for } x \downarrow t_k^{-1}, \text{ for } k = l+1, 2, ..., N ,$  $P(x) < 0 \text{ for } x \uparrow t_k^{-1}, \text{ for } k = l+1, 2, ..., N .$ 

Since Q is continuous on  $\mathbb{R}$  except maybe at  $t_k^{-1}$ , k = 1, 2, ..., N, we see that Q has at least N-2 zeros, denoted by  $s_k^{-1}$ , k = 1, 2, ..., l-1, l+2, ..., N, which satisfy

$$-T < t_N < s_N < \dots < s_{l+2} < t_{l+1} < 0 < t_l < s_{l-1} < \dots < s_1 < t_1 < T.$$

The graph of P(x) is sketched in figure 3.1 (see the next page). If a zero of Q,  $s^{-1}$  say  $(s \neq 0)$ , is the site of a local extremum of Q we must have

$$0 = P'(x) |_{x=s^{-1}} = \int_{[-T,T]} \frac{t}{(1-s^{-1}t)^2} d\mu(t),$$
  
$$0 = P(x) |_{x=s^{-1}} = \int_{[-T,T]} \frac{1-s^{-1}t}{(1-s^{-1}t)^2} d\mu(t),$$

which implies that

$$0 = \int_{[-T,T]} \frac{1}{(1-s^{-1}t)^2} d\mu(t),$$

i.e., that  $\mu = 0$ , *a.e.*. Thus no zero of Q is the site of a local extremum and so the reciprocal of the (N-1)st zero, s say, must lie in  $(t_{l+1}, t_l)$ . Let  $s_{l+1} = \min(0, s)$  and  $s_l = \max(0, s)$ . If Q only has N-2 zeros (this corresponds to s = 0 above and is the case when  $\sum q_i t_i = 0$ ), then let  $s_l = s_{l+1} = 0$ . In either case

3.3 Moment sequences and infinitely divisible sequences

$$P(x) = \prod_{k=1}^{N} (1 - xs_k) / \prod_{i=1}^{N} (1 - xt_i).$$
(3.9)

It follows that  $\tilde{R}$  can be written as (cf. (3.7))

$$\tilde{R}(x) = \log P(x) = \int_{0}^{T} \frac{x}{1-xt} d\mu_{N,1}(t) - \int_{0}^{T} \frac{x}{1-xt} d\mu_{N,2}(t) ,$$

with  $\mu'_{N,1}(t) = \sum_{k=1}^{l} \mathbf{1}_{(s_k,t_k)}(t)$ , and  $\mu'_{N,2}(t) = \sum_{k=l+1}^{N} \mathbf{1}_{(-s_k,-t_k)}(t)$ , so  $\mu_{N,1}$  and  $\mu_{N,2}$  are bounded by Lebesgue measure, i.e.,  $r_n / (n+1)$  has the desired property. A limiting argument completes the proof.



# figure 3.1

Conversely, we approximate the representing measures  $\mu_1$  and  $\mu_2$  of  $(b_n)$  and  $(c_n)$ , respectively. For k = 1, 2, ..., N-1, let

$$t_{k,N} = T(N-k)/N ,$$
  

$$s_{k,N} = t_{k,N} - \mu_1(0, t_{k,N}] + \mu_1(0, t_{k+1,N}] ,$$
  

$$\mu'_{N,1}(t) = \sum_{k=1}^{N-1} 1_{(s_{k,N}, t_{k,N})} (t) ,$$
  

$$\mu_{N,1}(0, t_{k,N}] = \mu_1(0, t_{k,N}] .$$

For k = N, N + 1, ..., 2N - 1, let

$$\begin{split} s_{k,N} &= T(N-k)/N , \\ t_{k,N} &= s_{k,N} - \mu_2(0, -s_{k+1,N}] + \mu_2(0, -s_{k,N}] , \\ \mu'_{N,2}(t) &= \sum_{k=N}^{2N-1} \mathbf{1}_{(-s_{k,N}, -i_{k,N})} (t) , \end{split}$$

.

 $\mu_{N,2}(0,-s_{k,N}] = \mu_2(0,-s_{k,N}]$ .

Since  $\mu_{N,1} \rightarrow \mu_1$  and  $\mu_{N,2} \rightarrow \mu_2$  we have by Helly's first and second theorems (cf. (3.7))

$$P(x) = \lim_{N \to \infty} \exp\left\{ \int_{0}^{T} \frac{x}{1 - xt} d\mu_{N, 1}(t) - \int_{0}^{T} \frac{x}{1 + xt} d\mu_{N, 2}(t) \right\}$$
$$= \lim_{N \to \infty} \prod_{k=1}^{2N-1} \frac{1 - xs_{k}}{1 - xt_{k}} = \lim_{N \to \infty} \sum_{k=1}^{2N-1} \frac{q_{k}}{1 - xt_{k}} = \int_{[-T, T]} \frac{1}{1 - xt} d\mu(t) .$$

COROLLARY 1.  $(p_n) \in MS(\mathbb{R})$  if and only if  $(r_{n+1}/(n+2)) \in MS(\mathbb{R})$  with representing measure  $\mu$  satisfying

$$\int_{-\infty} |t|^{-1} d\mu(t) < \infty \text{ and } \mu(B) \le \int_{B} |t| dt \text{ for all Borel sets } B$$

**PROOF.** Let  $\mu_b$  and  $\mu_c$  be the representing measures of  $(b_n)$  and  $(c_n)$  in Theorem 3.3.2, and let  $d\mu_1 = td\mu_b$ ,  $d\mu_2 = td\mu_c$ . Then the measure  $\mu$  defined by

$$\mu(-\infty, t] = \begin{cases} \mu_2(-t, \infty) & t \le 0\\ \mu_2(0, \infty) + \mu_1(0, t) & t > 0 \end{cases}$$

satisfies the requirements.

~

REMARK 3.3.3. Suppose that  $(p_{n+1}) \in MS_T(\mathbb{R})$  with a discrete representing measure and  $p_0 = 1$ . Proceeding as in the proof of Theorem 3.3.2 we see that P (the generating function of  $(p_n)$ ) takes the form

$$P(x) = 1 + \sum_{i=1}^{N} q_i x / (1 - xt_i) = Q(x) / \prod_{i=1}^{N} (1 - xt_i),$$

with Q a polynomial of degree at most N. As in the proof of Theorem 3.3.2 we can identify N-2 zeros of Q by observing the sign of P(x) as  $x \to t_k^{-1}$  from the left and from the right (cf. figure 3.1). The method of proof used to prove Theorem 3.3.2 now fails to prove Theorem 3.3.1 since the last two zeros cannot be confined to the interval  $(t_{l+1}, t_l)$ .

#### 3.3 Moment sequences and infinitely divisible sequences

#### 3.3.2. Stieltjes moment sequences

THEOREM 3.3.4. Let  $(p_n)$  and  $(r_n)$  be related by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_k r_{n-k} , \quad n \in \mathbb{N}_0 .$$

Then (cf. (3.3) and (3.4))

$$(p_n) \in MS(\mathbb{R}_+)$$
 if and only if  $(\frac{r_n}{n+1}) \in MS^*(\mathbb{R}_+)$ .

PROOF. For the first part we restrict attention to  $MS_T(\mathbb{R}_+)$  and  $MS_T^*(\mathbb{R}_+)$  (cf. Remark 3.2.5). If  $(p_n) \in MS_T(\mathbb{R}_+)$ , then by Lemma 3.2.2 we have  $(p_n) \in MS_T(\mathbb{R})$  and  $(p_{n+1}) \in MS_T(\mathbb{R}) \subset MS(\mathbb{R})$ . Hence, by the proof of Theorem 3.3.1 and the corollary to Theorem 3.3.2 one has  $(r_n/(n+1)) \in MS_T^*(\mathbb{R})$  and  $(r_{n+1}/(n+2)) \in MS(\mathbb{R})$ . By Lemma 3.2.11, then  $(r_n/(n+1)) \in MS_T^*(\mathbb{R}_+)$ .

Conversely, by Theorem 3.3.2 with  $c_n = 0$  (n = 0, 1, ...) we have  $(p_n) \in MS(\mathbb{R})$ . Since  $MS^*(\mathbb{R}_+) \subset MS^*(\mathbb{R})$ , by Theorem 3.3.1 we also have  $(p_{n+1}) \in MS(\mathbb{R})$ , and therefore, by Lemma 3.2.2,  $(p_n) \in MS(\mathbb{R}_+)$ .

COROLLARY 1. Let  $(u_n)$  and  $(v_n)$  be in  $MS(\mathbb{R}_+)$ , with generating functions U and V and let  $W := U^{\dot{a}}V^{b}$  be the generating function of  $(w_n)$ . Then  $(w_n) \in MS(\mathbb{R}_+)$  if  $a \ge 0$ ,  $b \ge 0$  and  $a + b \le 1$ .

PROOF. See (3.7) and (3.4).

REMARK 3.3.5. Theorem 3.3.4 can also be proved without use of Theorem 3.3.2 in a similar way as its analogue in Horn (1970) is proved. This proof, however, uses Theorem 2.3.3 (ii). As mentioned in Remark 3.3.3, the proof of Theorem 3.3.2 can be adapted to give yet another proof of Theorem 3.3.4.

REMARK 3.3.6. If  $(p_n)$  and  $(r_n/(n+1))$  are in  $MS(\mathbb{R})$  then by Theorem 3.3.1 and Lemma 3.2.2,  $(p_n) \in MS(\mathbb{R}_+)$  and by the corollary to Theorem 3.3.2 and Lemma 3.2.2,  $(r_n/(n+1)) \in MS(\mathbb{R}_+)$ . Hence, as in the renewal case (cf. Horn (1970)),  $(p_n)$  and  $(r_n/(n+1))$  cannot both be in  $MS(\mathbb{R})$  without being in  $MS(\mathbb{R}_+)$ .

It is interesting to note the difference between Theorems 3.3.1 and 3.3.4. Theorem 3.3.4 considers  $(p_n)$  as a moment sequence, whereas Theorem 3.3.1 considers  $(p_{n+1})$ . This shift in indices is necessary to ensure that the sequence  $(p_n)$  has the right sign. For example, if  $(r_n/(n+1)) \in MS^*(\mathbb{R})$ , then  $r_0 > 0$  and since  $p_0 > 0$  we must have  $p_1 > 0$ , which  $(p_{n+1}) \in MS(\mathbb{R})$  ensures. This observation led us to consider the set  $MS(\mathbb{R})$ , where the odd terms of the sequence are non-positive.

THEOREM 3.3.7. Let  $(p_n)$  and  $(r_n)$  be related by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_k r_{n-k}$$
,  $n \in \mathbb{N}_0$ .

Then (cf. (3.3) and (3.4))

$$(p_{n+1}) \in MS(\mathbb{R}_{-})$$
 if and only if  $(\frac{r_n}{n+1}) \in MS^*(\mathbb{R}_{-})$ .

**PROOF.** Any sequence in  $MS(\mathbb{R})$  can be approximated by sequences in  $MS_T(\mathbb{R})$  as follows. If  $p_0 = 1$  and  $(p_{n+1}) \in MS_T(\mathbb{R})$  has a discrete representing measure with positive atoms  $q_i$  at  $t_i$  where (i = 1, 2, ..., N) and  $-T < t_N < ... < t_{N-1} < ... < t_1 < 0$ , then P(x) takes the form

$$P(x) = 1 + \sum_{i=1}^{N} q_i x / (1 - xt_i) = Q(x) / \prod_{i=1}^{N} (1 - xt_i),$$

with Q a polynomial of degree at most N. Observe that

$$P(x) < 0$$
 for  $x \downarrow t_k^{-1}$ , for  $k = 1, 2, ..., N$ ,  
 $P(x) > 0$  for  $x \uparrow t_k^{-1}$  for  $k = 1, 2, ..., N$ 

$$P(x) > 0$$
 for  $x | t_k^{-1}$ , for  $k = 1, 2, ..., N$ ,

thus, since Q is continuous on  $\mathbb{R}$  except maybe at  $t_k^{-1}$ , k = 1, 2, ..., N, we see that Q has at least N-1 zeros, denoted by  $s_k^{-1}$ , k = 1, 2, ..., N-1, and satisfying

 $-T < t_N < s_{N-1} < t_{N-1} < s_{N-2} < \dots < s_2 < t_2 < s_1 < t_1 < 0.$ 

The graph of P(x) is sketched in figure 3.2 (see the next page). Since  $P(x) \to -\infty$  as  $x \downarrow t_N^{-1}$  and P(0)=1, then the last zero,  $s_N^{-1}$  say, must satisfy  $-T' < s_N < t_N$  for some T' > 0. Hence

$$P(x) = \prod_{k=1}^{N} (1 - xs_k) / \prod_{i=1}^{N} (1 - xt_i),$$

and so

$$\tilde{R}(x) = \log P(x) = \int_{-T'}^{0} \frac{x}{1-xt} d\mu_N(t)$$

with  $\mu'_N(t) = \sum_{k=1}^N \mathbf{1}_{(s_k,t_k)}(t)$ , so  $\mu_N$  is bounded by Lebesgue measure, i.e.,  $r_n / (n + 1)$  has the desired property. A limiting argument completes the proof.




The converse is proved as in Theorem 3.3.2.

## 3.3.3. Hausdorff moment sequences

THEOREM 3.3.8. Let  $(p_n)$  and  $(r_n)$  be related by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} p_k r_{n-k} , \quad n \in \mathbb{N}_0 .$$

Then (cf. (3.3) and (3.4)).

$$(p_n) \in MS([0, 1])$$
 if and only if  $(\frac{r_n}{n+1}) \in MS^*([0, 1])$ .

PROOF. If  $(p_n) \in MS([0,1])$ , then  $(p_n)$  is non-increasing and since  $p_0 = 1$ , we have  $p_n \le 1$ . By Lemma 3.2.3,  $(p_n) \in MS(\mathbb{R}_+)$ . By Theorem 3.3.4,  $(r_n/(n+1)) \in MS^*(\mathbb{R}_+)$  and by (3.1),  $p_n \le 1$  implies that  $r_n \le n+1$ . Hence  $(r_n/(n+1)) \in MS^*([0,1])$  by Lemma 3.2.3.

Conversely, if  $(r_n/(n+1)) \in MS^*([0,1])$ , then by Lemma 3.2.3,  $(r_n/(n+1)) \in MS^*(\mathbb{R}_+)$ and  $r_n \leq 1$ . By Theorem 3.3.4,  $(p_n) \in MS(\mathbb{R}_+)$  and by (3.1),  $r_n \leq 1$  implies that  $p_n \leq 1$ and hence, by Lemma 3.2.3,  $(p_n) \in MS([0,1])$ . COROLLARY 1. Let  $(u_n)$  and  $(v_n)$  be in MS ([0, 1]) with generating functions U and V and let  $W = U^a V^b$  be the generating function of  $(w_n)$ . Then  $(w_n) \in MS([0, 1])$  if  $a \ge 0, b \ge 0$  and  $a + b \le 1$ .

COROLLARY 2. Let  $(p_n)$  and  $(r_n)$  be related by (3.1). Then (cf. (3.3) and (3.4))

$$(p_n) \in MS([0,T])$$
 if and only if  $(\frac{r_n}{n+1}) \in MS^*([0,T])$ ,

for any T > 0.

PROOF. If  $(p_n) \in MS([0,T])$  then  $(T^{-n}p_n) \in MS([0,1])$ . From (3.1) it follows that

$$(n+1)[T^{-n-1}p_{n+1}] = \sum_{k=0}^{n} [T^{-k}p_k] [T^{-n+k-1}r_{n-k}], \ n \in \mathbb{N}_0$$

Hence, by Theorem 3.3.8,  $(T^{-n-1}\frac{r_n}{n+1}) \in MS^*([0,1])$ . Observe that

$$r_n/(n+1) = T^{n+1} \int_0^1 x^n \, d\mu(x) = \int_0^T y^n \, dT \mu(y/T) \, .$$

Since  $T\mu(y/T)$  is bounded by Lebesgue measure, then  $(r_n/(n+1)) \in MS^*([0,T])$ . The converse is proved similarly.

## 3.4 Moment sequences and renewal sequences

Let the two sequences  $(p_n)$  and  $(r_n)$  be related through (3.1). Define the sequence  $(f_n)$  by

$$p_{n+1} = \sum_{k=0}^{n} p_k f_{n-k} , \quad n \in \mathbb{N}_0 .$$
(3.10)

In Section 2.5 we proved that for  $f_{-1} \coloneqq -1$ 

$$(n+1)[(-1)^n f_n] = \sum_{k=0}^n \left[ (-1)^{k-1} f_{k-1} \right] \left[ (-1)^{n-k} r_{n-k} \right], \ n \in \mathbb{N}_0 .$$
(3.11)

We now give an alternative proof of Theorem 2.3.3.

**PROOF** (of Theorem 2.3.3). From the definitions of  $MS(\mathbb{R})$  and  $MS^*(\mathbb{R})$  it is clear that

- (a)  $(a_n) \in MS(\mathbb{R})$  (or  $MS^*(\mathbb{R})$ ) iff  $((-1)^n a_n) \in MS(\mathbb{R})$  (or  $MS^*(\mathbb{R})$ );
- (b)  $(a_n) \in MS(\mathbb{R}_+)$  (or  $MS^*(\mathbb{R}_+)$ ) iff  $((-1)^n a_n) \in MS(\mathbb{R}_-)$  (or  $MS^*(\mathbb{R}_-)$ ).

Theorem 2.3.3 (i) is proved by noting the equivalence of the following statements.

| $(p_{n+1}) \in MS(\mathbb{R})$ iff $(r_n/(n+1)) \in MS^*(\mathbb{R})$ | (cf. (3.1) and Theorem 3.3.1)  |
|---|--------------------------------|
| $\inf\left((-1)^n r_n/(n+1)\right) \in MS^*(\mathbb{R})$              | (cf. statement (a))            |
| $\inf\left((-1)^n f_n\right) \in MS(\mathbb{R})$                      | (cf. (3.11) and Theorem 3.3.1) |
| iff $(f_n) \in MS(\mathbb{R})$ .                                      | (cf. statement (a))            |

For the result on Stieltjes moment sequences we note the following.

| (cf. (3.1) and Theorem 3.3.4)  | $(p_n) \in MS(\mathbb{R}_+)$ iff $(r_n/(n+1)) \in MS^*(\mathbb{R}_+)$      |
|--------------------------------|--|
| (cf. statement (b))            | $\operatorname{iff}\left((-1)^n r_n/(n+1)\right) \in MS^*(\mathbb{R}_{-})$ |
| (cf. (3.11) and Theorem 3.3.7) | $\operatorname{iff}\left((-1)^n f_n\right) \in MS(\mathbb{R}_{-})$         |
| (cf. statement (b))            | iff $(f_n) \in MS(\mathbb{R}_+)$ .   |

The result on MS([0,1]) is easiest proved as in Horn (1970).

The same reasoning for  $MS(\mathbb{R}_{)$  yields

THEOREM 3.4.1. Let  $(u_n)$  and  $(f_n)$  be related by equation (2.1), i.e., by

$$u_{n+1} = \sum_{k=0}^{n} u_k f_{n-k} , \quad n \in \mathbb{N}_0 .$$

Then (cf. (3.3) and (3.4))

 $(u_{n+1}) \in MS(\mathbb{R})$  if and only if  $(f_n) \in MS(\mathbb{R})$ .

## 3.5 Moment functions and infinitely divisible functions

Let F and H be two non-decreasing, right continuous, not necessarily bounded functions on  $\mathbb{R}_+$  related through equation (1.1) of Theorem 1.3.3, i.e., through

$$\int_{0}^{x} u \, dF(u) = \int_{0}^{x} F(x-u) \, dH(u) \, , \quad x \in \mathbb{R}_{+} \, . \tag{3.12}$$

Let F and H be absolutely continuous with Radon-Nikodym derivatives f and h, respectively. Then

$$x f(x) = \int_{0}^{x} f(x-u) h(u) du, \quad x \in \mathbb{R}_{+}.$$
(3.13)

In this section we consider (3.13) where either f or h(u)/u are moment functions (cf.

Definition 3.5.1). In the proof of Theorem 3.5.3 of this section it is essential whether or not the distribution function of f or of h(u)/u is bounded. We therefore state the theorem in terms of the densities of the distribution functions in (3.12), instead of in terms of the functions in (3.13). We consider (3.13) in the context of moment functions instead of (3.12), partly because (3.13) is the analogue of (3.1) and partly because c - F is a bounded moment function for some c > 0 if and only if f is a moment function with  $\int f = c$ .

The set MS([0,1]) coincides with the set of completely monotone sequences (cf. Feller (1971), Section VII,3). The set of completely monotone functions is equal to the set of Laplace-Stieltjes transforms (cf. Feller (1971), Section XIII,4). In order to keep the analogy between completely monotone sequences and completely monotone functions we introduce the following definition.

DEFINITION 3.5.1. Let  $MF((0, \infty))$  and MF((0,1]) denote the sets of Stieltjes and Hausdorff moment functions, respectively, i.e., for  $I \in \{(0, \infty), (0, 1]\}$  let

$$f \in MF(I)$$
 if and only if  $f(\tau) = \int_{-\ln(I)} e^{-x\tau} d\mu(x)$ ,  $\tau \in \mathbb{R}_+$ , (3.14)

where  $\mu$  is a nonnegative measure and  $-\ln((0,1]) = [0,\infty)$  and  $-\ln((0,\infty)) = \mathbb{R}$ . Also let the sets  $MS^*(I)$  (cf. 3.14)) be given by

 $f \in MF^*(I)$  if and only if  $f \in MF(I)$  with  $\mu \leq \lambda$ .

The measure  $\mu$  in (3.14) is called the representing measure of f.

We will need the following lemma. The proof is almost identical with that of Lemma 3.2.4 and therefore omitted.

LEMMA 3.5.2. Let  $I \in \{ (0, \infty), (0, 1] \}$ . If  $f_t \in MF(I)$  (or  $MF^*(I)$ ) for all  $t \ge t_0$ , and if  $\lim_{t \to 0} f_t(\tau) = f(\tau), \tau \in \mathbb{R}_+$ , then  $f \in MF(I)$  (or  $MF^*(I)$ ).

Part (ii) of the following theorem was proved by Steutel (1970), p. 44, Theorem 2.12.1, for distribution functions. We slightly generalize this result in

THEOREM 3.5.3. Let F and H be related by (3.12) and suppose they have densities f and h, respectively. Then

(i)  $f \in MF((0, \infty))$  if and only if  $\tau^{-1}h(\tau) \in MF^*((0, \infty))$ ;

(ii) 
$$f \in MF((0,1])$$
 if and only if  $\tau^{-1}h(\tau) \in MF^*((0,1])$ .

PROOF. Theorem 2.12.1 of Steutel (1970) proves part (ii) for bounded F. Any function F with density  $f \in MF((0,1])$  can be written as a limit of bounded functions  $F_t$  with densities  $f_t \in MF((0,1])$ . For each  $F_t$  a function  $H_t$  can be found satisfying (3.12). By Theorem 3.5.3 (ii) for bounded F,  $\tau^{-1}h_t(\tau) \in MF^*((0,1])$ . Since  $F_t \to F$  and  $H_t \to H$ , then by Lemma 3.5.2,  $\tau^{-1}h(\tau) \in MF^*((0,1])$ . Similarly when starting with  $\tau^{-1}h(\tau)$ .

Part (i) can be proved from part (ii) as follows; if  $f \in MF((0, \infty))$ , then by truncation of integrals we have a sequence  $f_t \in MF((0,t])$  such that  $f_t \to f$ . Since  $f_t \in MF((0,t])$  if and only if  $t^{-\tau}f_t(\tau) \in MF((0,1])$ , then by part (ii)  $t^{-\tau}\tau^{-1}h_t(\tau) \in MF^*((0,1])$  and hence  $\tau^{-1}h_t(\tau) \in MF^*((0,t])$ . By Lemma 3.5.2, on letting  $t \to \infty$  it follows that  $\tau^{-1}h(\tau) \in MF^*((0,\infty))$ . Similarly when starting with  $\tau^{-1}h(\tau)$ .

**REMARK 3.5.4.** Part (i) of Theorem 3.5.3 can also be proved by applying a limiting argument to Theorem 3.3.8 as in the proof of Theorem 4.3.2 (see Chapter 4). Yet another way to prove Theorem 3.5.3 (i) (or another way to prove Theorem 3.3.8 using Theorem 3.5.3 (i)) is as follows:

Suppose  $f \in MF((0,1])$ . By Steutel (1969) f is infinitely divisible and therefore satisfies (3.13). For any  $\tau \le c$ ,  $c \ge 1$ ,  $\hat{f}$  can be put in the form (cf. (3.14))

$$\hat{f}(\tau) = \int_{0}^{\infty} x(x + \tau/c)^{-1} d\mu(cx)$$
  
= 
$$\int_{0}^{1} (1 - x(1 - \tau/c))^{-1} d\mu_{1,c}(x) \coloneqq P_{c}(1 - \tau/c), \qquad (3.15)$$

where  $d\mu_{1,c}(x) = -(1-x)d\mu(c(x^{-1}-1))$ . Let  $(p_n(c))$  have generating function  $P_c$ . Then  $(p_n(c)) \in MS([0,1])$ . Let  $(r_n(c))$  be defined by using  $(p_n(c))$  in (3.1) and let  $R_c$  denote its generating function. By Theorem 3.3.8,  $(r_n(c)/(n+1)) \in MS^*([0,1])$ , and hence

$$R_{c}(z) = \int_{0}^{1} (1 - xz)^{-2} d\mu_{1,c}(x) = \int_{0}^{\infty} c (y + c (1 - z))^{-2} d\mu_{2,c}(y) , \qquad (3.16)$$

with  $d\mu_{2,c}(y) = -c ((y/c) + 1)^2 d\mu_{1,c}(((y/c) + 1)^{-1})$ . Hence  $\mu_{2,c} \le \lambda$ . Observe that (cf. (3.13))

$$\hat{h}(\tau) = -\frac{d}{d\tau} \ln f(\tau) = -\frac{d}{d\tau} \ln P_c(1-\tau/c) = c^{-1} R_c(1-\tau/c) .$$
(3.17)

From (3.16) and (3.17) it follows that  $h(\tau)/\tau \in MF^*((0,1])$ . The converse is shown similarly.

#### 3.6 Moment functions and renewal functions

Let F be a distribution function on  $\mathbb{R}_+$  with density f and Laplace-Stieltjes transform  $\hat{f}$ . Suppose F is a compound geometric distribution, i.e.,

$$F(x) = (1-p) \sum_{n=0}^{\infty} p^n G^{*n}(x), \quad x \in \mathbb{R}_+,$$

with G a distribution function, having density g, and  $G^{*n}$  being the *n*-th convolution power of G. Applying the same type of argumentation as used in Remark 3.5.4 to Theorem 2.3.3 or applying a limit argument to Theorem 2.3.3 as in Hansen and Frenk (1988), it can be shown that (this is partly shown in Sumita and Masuda (1987), Theorem 3.3.2, p. 644 and is proved in Frenk (1988), both with proofs similar to that of Theorem 3.3.2))

$$f \in MF((0,1])$$
 if and only if  $g \in MF((0,1])$ .

Let  $U_p := (1-p)^{-1}F$  and  $U := \lim_{p \to 1} U_p$ . If u is the density of U, then by Lemma 3.5.2,

$$u \in MF((0,1])$$
 if and only if  $g \in MF((0,1])$ . (3.18)

The function U (called the renewal function (cf. Ross (1983))) is the unique solution of the renewal equation,

$$U(x) = 1 + \int_{0}^{x} G(x - y) dU(y), \quad x \in \mathbb{R}_{+}.$$
(3.19)

Using the method of proof of Theorem 3.5.3 on statement (3.18) we obtain

THEOREM 3.6.1. Let U and G be related by (3.19) and suppose they have densities u and g respectively. Then

(i)  $u \in MF((0, \infty))$  if and only if  $g \in MF((0, \infty))$ ; (ii)  $u \in MF((0, 1])$  if and only if  $g \in MF((0, 1])$  and  $\int_{0}^{\infty} g(x) dx \le 1$ .

# 3.7 Applications and special cases

\*

The main occurrence of (3.1) is in infinite divisibility. In this context  $(p_n)$  is a Hausdorff moment sequence if and only if  $(p_n)$  is a mixture of geometric distributions, i.e.,

$$p_n = \int_0^1 x^n (1-x) dF(x) , \quad n \in \mathbb{N}_0 , \qquad (3.20)$$

#### 3.7 Applications and special cases

#### where F is a distribution function on [0, 1]. Theorem 3.3.8 can be restated as

THEOREM 3.7.1.  $(p_n)$  in (3.1) is a mixture of geometric distributions if and only if  $(g_n) = (r_n/\{\theta(n+1)\})$  (cf. Theorem 1.3.4) is a mixture of geometric distributions with F in (3.20) satisfying  $(1-t)dF \le d\lambda$  (cf. (3.4)). Equivalently, the probability generating function P is of the form

$$P(z) = \int_{0}^{1} (1-x)/(1-xz) \, dF(x),$$

with F a distribution function on [0,1], if and only if P can be represented as

$$\ln P(z) = \int_{z}^{1} \int_{0}^{1} (1-uv)^{-2} d\mu(v) du,$$

and  $\mu$  bounded by Lebesgue measure. The representation is unique.

The continuous analogue of Theorem 3.7.1 is implicit in Theorem 2.12.1 of Steutel (1970).

As a curiosity we prove that it is possible to have  $(p_n) = (g_{n+1})$  in Theorem 3.7.1, i.e., zP(z) = G(z) in Theorem 1.3.4. Solving for  $(p_n)$  in  $P(z) = P(0) \exp [\theta z P(z)]$  one finds, e.g. by Lagrange expansion,

$$p_n = \frac{(n+1)^n}{(n+1)!} \ (\theta e^{-\theta})^n \ e^{-\theta} \ , \quad n \in \mathbb{N}_0 \ . \tag{3.21}$$

Since

$$\frac{n^n}{n!} = \frac{1}{\pi} \int_0^{\pi} (\frac{\sin x}{x} \exp[x \cot x])^n dx , \quad n \in \mathbb{N}_0 ,$$

as has been proved by Bouwkamp (1986),  $(p_n)$  is a Hausdorff moment sequence. The distribution in (3.21) and its continuous analogue are busy period distributions. This application is discussed in Steutel and Hansen (1988).

We finish this chapter by stating a conjecture, the continuous analogue of which is discussed in Steutel (1970) (p.28, 94). In probabilistic terms the conjecture is that mixtures of negative binomial distributions of order 2, i.e., of probabilities of the form  $(n+1)p^n(1-p)^2$  (n = 0, 1,...), are infinitely divisible. We state the conjecture more formally as follows.

CONJECTURE. If  $(a_n)$  is a Hausdorff moment sequence, then  $(p_n) := ((n+1)a_n)$  satisfies (3.1) with  $r_k \ge 0$ .

Note that the conjecture is true if and only if the analogous conjecture in Steutel (1970) concerning mixtures of gamma two distributions is true (this can be seen by applying the method used to couple distributions on  $\mathbb{N}_0$  and  $\mathbb{R}_+$  in Remark 3.5.4). It is however numerically easier to look for counter examples in the discrete case than in the gamma two case. We computed<sup>(\*)</sup> the first 200  $r_k$ 's for 20 different mixtures of negative binomial distributions of order two and did not find any negative  $r_k$ 's.

<sup>(\*)</sup> My thanks go to A.A Stoorvogel for his programming help.

Chapter 4

# LOGCONCAVE AND LOGCONVEX SEQUENCES AND DENSITIES

## **4.1 Introduction**

Log-concavity and log-convexity of functions and sequences in probability has been of interest to several authors, e.g. Karlin (1968). Ibragimov (1956) calls a distribution strongly unimodal if its convolution with any unimodal distribution is unimodal. He proves that the set of strongly unimodal probability densities is equal to the set of log-concave densities. An equivalent result for log-concave discrete probability distributions has been proved by Keilson and Gerber (1971). Much work has been done on the unimodality of infinitely divisible distributions (cf. Wolfe (1971), Yamazato (1978) and Sato and Yamazato (1978)), but little on strong unimodality. The study of log-concave functions and sequences is thus a relatively unknown field in probability, with important applications in the fields of statistics and optimization. Log-convexity is of interest in the study of reliability and of infinitely divisible random variables. Steutel (1970) proves that all log-convex discrete probability distributions are infinitely divisible (Theorem 2.4.1 (ii)). The absolutely continuous analogue is also proved in Steutel (1970).

In this chapter we consider distributions of non-negative infinitely divisible random variables whose canonical measures are either absolutely continuous or supported by the integers. We prove that for such distributions to be log-concave (log-convex), it is sufficient that their canonical measures be log-concave (log-convex). Our results in the discrete case contain an analogue of Yamazato's (1982) concavity result (it also provides an alternative proof of this result), and an analogue to the convexity result for renewal sequences in de Bruijn and Erdös (1953) (cf. Theorem 2.3.2 (i)).

## 4.2 Discrete distributions

In this section we consider infinitely divisible discrete probability distributions  $(p_n)$  on  $\mathbb{N}_0$ . A sequence  $(a_n)$  is log-concave if  $(a_n)$  is non-negative and  $(\log(a_n))$  is concave (here  $\ln 0 := -\infty$ ), or equivalently if  $a_n \ge 0$  and

$$a_n^2 \ge a_{n+1} a_{n-1}$$
,  $n \in \mathbb{N}_+$ . (4.1)

If the sequence satisfies (4.1) with strict inequality, then the sequence is said to be strictly log-concave. Similarly,  $(a_n)$  is log-convex if  $a_n \ge 0$  and the sequence satisfies

$$a_n^2 \le a_{n+1} a_{n-1}$$
,  $n \in \mathbb{N}_+$ . (4.2)

 $(a_n)$  is said to be strictly log-convex if (4.2) is satisfied with strict inequality. Logconvex sequences are sometimes called Kaluza sequences, since Kaluza (1928) was the first to study sequences satisfying (4.2). Karlin (1968) calls a sequence satisfying (4.1) a Pólya frequency sequence of order 2.

A probability distribution  $(p_n)$  on  $\mathbb{N}_0$  with  $p_0 > 0$  is infinitely divisible if and only if it satisfies

$$(n+1)p_{n+1} = \sum_{k=0}^{n} r_k p_{n-k} , \qquad n \in \mathbb{N}_0 , \qquad (4.3)$$

with non-negative  $r_k$  and, necessarily,  $\sum_{k=0}^{\infty} r_k/(k+1) < \infty$  (cf. Theorem 1.3.4). Theorem 2.4.1 part (ii) states that all log-convex distributions are infinitely divisible. This can be proved by induction since

$$r_n p_n p_0 = p_n p_{n+1} + \sum_{k=0}^{n-1} r_k (p_{n+1} p_{n-k-1} - p_{n-k} p_n)$$

is positive if  $(p_n)$  is strictly log-convex and noting that any log-convex sequence can be written as a limit of strictly log-convex sequences. Not all log-concave distributions are infinitely divisible since (cf. (4.3))

$$r_1 = p_0^{-2} (2p_2p_0 - p_1^2)$$

is not necessarily non-negative when  $(p_n)$  is log-concave. We also note that logconcave and log-convex distributions do in fact exist, for example the generalized logarithmic series distribution is log-convex (cf. Hansen and Willekens (1989)) and the binomial and Poisson distributions are log-concave (cf. Keilson and Gerber (1971)). Also the geometric distribution is both log-convex and log-concave.

The proofs of the two theorems in this section rely on two equations derived from (4.3). Though easily verified by (4.3), the equations were rather hard to find. Because of their importance we state them in a lemma.

k=0

LEMMA 4.2.1. Let 
$$(p_n)$$
 and  $(r_n)$  be related by (4.3) and let  $p_{-1} := 0$ . Then  

$$m (m+2) (p_{m+1}^2 - p_m p_{m+2}) = p_{m+1} (r_0 p_m - p_{m+1})$$

$$+ \sum_{l=0}^{m} \sum_{k=0}^{l} (p_{m-l} p_{m-k-1} - p_{m-k} p_{m-l-1}) (r_{k+1} r_l - r_{l+1} r_k) ;$$

$$r_{m+1} (m+2) (p_{m+1} p_{m+3} - p_{m+2}^2) = p_{m+1} (r_{m+2} p_{m+2} - r_{m+1} p_{m+3})$$

$$+ \sum_{k=0}^{m} (p_{m-k} p_{m+2} - p_{m+1} p_{m-k+1}) (r_{m+2} r_k - r_{k+1} r_{m+1}) .$$
(4.4)

Relation (4.4) is a discrete analogue of equation (10) in Yamazato (1982), whereas (4.5) is an analogue of formula (7) in de Bruijn and Erdös (1953). We will need the following lemma.

LEMMA 4.2.2. Let  $(p_n)$  and  $(r_n)$  be related by (4.3) with  $p_0 > 0$ . Then

- (i) if  $p_n^2 > p_{n-1} p_{n+1}$  for n = 1, 2, ..., m then  $r_0 p_m p_{m+1} > 0$ ;
- (ii) if  $(r_n)$  is strictly log-convex and  $r_0^2 r_1 < 0$  then  $r_{m+2} p_{m+2} r_{m+1} p_{m+3} > 0$ .

PROOF. If  $p_n^2 > p_{n-1}p_{n+1}$  for n = 1, 2, ..., m, then  $p_{n+1}/p_n$  is decreasing for n = 1, 2, ..., m, so  $r_0 = p_1/p_0 > p_{m+1}/p_m$ .

If  $(r_n)$  is strictly log-convex, then  $(r_{n+1} / r_n)$  is increasing. Hence,

$$(m+3) p_{m+3} = p_{m+2} r_0 + \sum_{k=1}^{m+2} p_{m+2-k} r_{k-1} \frac{r_k}{r_{k-1}}$$

$$< p_{m+2} r_0 + (m+2) p_{m+2} \max_{1 \le k \le m+2} \left\{ \frac{r_k}{r_{k-1}} \right\}$$

$$< p_{m+2} \frac{r_{m+2}}{r_{m+1}} + (m+2) p_{m+2} \frac{r_{m+2}}{r_{m+1}} .$$

We are now ready to prove the two main results of this section. We begin with the result on log-concave sequences.

THEOREM 4.2.3. Let  $(p_n)$  and  $(r_n)$  be related by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} r_k p_{n-k} , \qquad n \in \mathbb{N}_0 , \qquad (4.3)$$

with  $r_k \ge 0$ ,  $p_0 > 0$  and let  $(r_n)$  be log-concave. Then

 $(p_n)$  is log-concave if and only if  $r_0^2 - r_1 \ge 0$ .

**PROOF.** Suppose that  $(r_n)$  is strictly log-concave and  $r_0^2 - r_1 > 0$ , then  $(r_n)$  is positive and hence  $(p_n)$  is positive. Observe that (cf. (4.3))

$$2(p_1^2 - p_0 p_2) = p_0^2 (r_0^2 - r_1).$$
(4.6)

By using (4.6), Lemma 4.2.2 (i) and applying induction to (4.4) we see that  $(p_n)$  is strictly log-concave. The proof is completed by noting that any log-concave sequence can be written as a limit of strictly log-concave sequences.

The following theorem states the result on log-convex sequences. It is the infinitely divisible analogue of Theorem 2.3.2 (i).

THEOREM 4.2.4. Let  $(p_n)$  and  $(r_n)$  be related by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} r_k p_{n-k} , n \in \mathbb{N}_0$$

with  $r_k \ge 0$ ,  $p_0 > 0$  and let  $(r_n)$  be log-convex. Then

 $(p_n)$  is log-convex if and only if  $r_0^2 - r_1 \le 0$ .

PROOF. As in Theorem 4.2.3 except that Lemma 4.2.2 (ii) is used and induction is applied to (4.5).  $\hfill \Box$ 

It is curious to note the difference in equations (4.4) and (4.5). We were not able to find an equation of the form (4.4) to prove Theorem 4.2.4 or one of the form (4.5) to prove Theorem 4.2.3.

REMARC 4.2.5. The assumption that  $(p_n)$  is a probability distribution is not used in the proofs of Theorems 4.2.3 and 4.2.4. These theorems are thus true for arbitrary non-negative sequences related by (4.3).

#### 4.3 Absolutely continuous distributions

## 4.3 Absolutely continuous distributions

In this section infinitely divisible probability distributions F on  $\mathbb{R}_+$  with absolutely continuous canonical measures are considered. We obtain two results on the log-concavity and log-convexity of the densities of F, which are analogues to those obtained in Section 4.2. The result on log-concave densities is proved in Yamazato (1982). We here propose a proof based on applying a limiting argument to Theorem 4.2.3. This proof can easily be adapted to log-convex densities, thus giving the absolutely continuous analogue of Theorem 4.2.4.

A function f on  $\mathbb{R}$  is log-concave (log-convex) on an interval I if f is positive on I and log(f) is concave (convex) on I. The function f is said to be log-concave (log-convex) if  $I = \{x \mid f > 0\}$  is an interval and f is log-concave (log-convex) on I. As in the discrete case, f is strictly log-concave (strictly log-convex) if  $\ln(f)$  is strictly concave (strictly convex).

A distribution function F on  $(0,\infty)$  is infinitely divisible if and only if there exists a non-decreasing measure H such that

$$\int_{0}^{x} u \, dF(u) = \int_{0}^{x} F(x-u) \, dH(u) \,, \tag{4.7}$$

$$\int_{1}^{\infty} u^{-1} dH(u) < \infty , \qquad (4.8)$$

where H and F determine each other uniquely (cf. Theorem 1.3.3). If F and H have densities f and h, then

$$x f(x) = \int_{0}^{x} h(x-u) f(u) du .$$
(4.9)

Without loss of generality we assume that  $\inf\{x \mid f(x) > 0\} = 0$ . It is shown in Steutel (1970) that all absolutely continuous distributions with log-convex densities are infinitely divisible. As in the discrete case, not all distributions having log-concave densities are infinitely divisible, e.g.  $f(x)=c\exp(-x^2)$  for  $x \in (0,\infty)$ .

We begin with a lemma.

LEMMA 4.3.1. Let f and h be related by (4.9). Suppose h is monotone on  $(0,\varepsilon)$  for some  $\varepsilon > 0$  and  $0 < f(0+) < \infty$ . Then h(0+) = 1.

**PROOF.** Suppose h is non-increasing on  $(0,\varepsilon)$ , then h(0+) > 0. By (4.9) the function f is continuous. From (4.9) it follows that for  $0 < x < \varepsilon$ 

$$h(0+) \ge x f(x) / \int_0^x f(u) \, du \, ,$$

$$h(x) \le x f(x) / \int_0^x f(u) \, du \; .$$

As  $x \to 0$  the right hand sides tend to one, so h(0+)=1. Similarly if h is non-decreasing.

THEOREM 4.3.2. (Yamazato) Let F be an infinitely divisible distribution function on  $(0,\infty)$  with an absolutely continuous canonical measure H. Let f and h be the densities of F and H respectively, and assume that h is log-concave. Then

PROOF. Suppose h is log-concave and h(0+) > 1, then h must be continuous on I. Define  $(r_n(k))$  by

$$r_n(k) = h\left(\frac{n+1}{k}\right), \qquad n \in \mathbb{N}_0,$$

and any  $k \in \mathbb{N}_0$ . Then  $(r_n(k))$  is log-concave, and since h(0+) > 1 we have  $(r_0(k))^2 > r_1(k)$ , for all sufficiently large k. By (4.8) and the continuity of h we see that  $\sum r_n(k)/(n+1) < \infty$ . For fixed k define  $(p_n(k))$  by

$$(n+1)p_{n+1}(k) = \sum_{l=0}^{n} p_{n-l}(k)r_l(k), \qquad n \in \mathbb{N}_0, \qquad (4.10)$$

$$p_0(k) = k \exp(-\sum_{n=0}^{\infty} r_n(k) / (n+1)) > 0$$
, (4.11)

with  $\sum p_n(k) = k$ . By Theorem 4.2.3 and Remark 4.2.5 the sequence  $(p_n(k))$  is log-concave. Let

$$F_{k}(x) = \sum_{\substack{n \ge 0 \\ n \le kx}} k^{-1} p_{n}(k), \qquad (4.12)$$

$$H_{k}(x) = \sum_{\substack{n \ge 0 \\ n \le kx}} k^{-1} r_{n}(k) .$$
(4.13)

From (4.10) and (4.11) it follows that

$$\int_{[0,x+k^{-1}]} u \, dF_k(u) = \int_{[0,x]} F_k(x-u) \, dH_k(u) \,, \tag{4.14}$$

$$\frac{n+1}{k}p_{n+1}(k) = \int_{[0,\frac{n}{k}]} h(\frac{n+1}{k} - u) \, dF_k(u) \quad . \tag{4.15}$$

By Helly's first theorem (cf. Feller (1971)) there is a subsequence  $(F_{k(s)})$  converging weakly to some distribution function,  $F_{limit}$  say, as  $s \to \infty$ . Hence, since  $H_k \to H$ , by Helly's second theorem

f is log-concave if and only if  $h(0+) \ge 1$ .

4.3 Absolutely continuous distributions

$$\int_{[0,x]} u \, dF_{limit}(u) = \int_{[0,x]} F_{limit}(x-u) \, dH(u) \; .$$

Since H uniquely determines F in (4.7) we must have  $F = F_{limit}$ . Let

$$f_k(x) = (p_{n+1}(k))^{kx-n} \cdot (p_n(k))^{n+1-xk}, \qquad x \in [\frac{n}{k}, \frac{n+1}{k})$$
(4.16)

then  $f_k$  is a log-concave function of x. Let  $n \to \infty$  and  $k \to \infty$  in such a way that  $k^{-1}$   $(n+1) \to x$ , then it follows from (4.9), (4.15) and (4.16) that

$$x f_{limit}(x) := \lim_{\substack{k \to \infty \\ n \to \infty \\ k^{-1}(n+1) \to x}} \frac{n+1}{k} f_k(\frac{n+1}{k}) = \int_{[0,x)} h(x-u) \, dF(u) \stackrel{u.e.}{=} x f(x) \ .$$

Since log-concavity is preserved under convergence, F has a log-concave density  $f_{limit}$ . As any log-concave function with  $h(0+) \ge 1$  can be written as a limit of log-concave functions  $(h_k)$  with  $h_k(0+) > 1$ , this completes the first part of the proof.

Conversely, if f and h are log-concave then h is monotone on  $(0,\varepsilon)$  for some  $\varepsilon > 0$ . If  $0 < f(0+) < \infty$  then h(0+)=1 by Lemma 4.3.1. If f is log-concave then f(0+) cannot be infinite. If f(0+)=0, then f is non-decreasing on  $(0,\varepsilon)$  and

$$x f(x) \le f(x) \int_{0}^{x} h(u) du$$

Letting  $x \rightarrow 0$  yields  $h(0+) \ge 1$ .

The proof of Theorem 4.3.2 can easily be adapted to log-convex densities by using Theorem 4.2.4 instead of Theorem 4.2.3. We then obtain

THEOREM 4.3.3. Let F be an infinitely divisible distribution function with an absolutely continuous canonical measure H. Let f and h be the densities of F and H respectively, and assume that h is log-convex. Then

f is log-convex if and only if  $h(0+) \le 1$ .

#### 4.4 Applications and counterexamples

In this section we define a set of infinitely divisible distributions in terms of their canonical measures and determine under what conditions a distribution in this set is log-concave or log-convex. An application of this result shows that if  $(p_n)$  is log-concave (log-convex) then  $(r_n)$  is not necessarily log-concave (log-convex). Finally, we characterize the log-convex discrete stable distributions.

Let D denote the set of distributions having canonical measures  $(r_n)$  of the form

$$r_{n} = (n+1) \left\{ \int_{0}^{b} y^{n} dm(y) + \int_{c}^{a} y^{n} dy \right\}, \quad n \in \mathbb{N}_{0}, \qquad (4.17)$$

for fixed a, b and c with  $0 \le b \le 1$ ,  $0 \le c \le a \le 1$ , m bounded by Lebesgue measure and

$$\int_{a}^{c} dm(y) < b - a, \quad \text{if } b > a,$$

$$\int_{0}^{c} dm(y) < c, \quad \text{if } c > 0.$$

٢

The proof of Theorem 4.2 in Yamazato (1982) can be adapted to prove the following theorem if Theorem 3.3.8 of Chapter 3 and Corollary 2 to Theorem 3.3.8 is used in the same fashion as Lemma 4.2 in Yamazato (1982).

THEOREM 4.4.1. Let  $(p_n)$  and  $(r_n)$  be related by

$$(n+1)p_{n+1} = \sum_{k=0}^{n} r_k p_{n-k} , \quad n \in \mathbb{N}_0 ,$$

with non-negative  $r_k$  and  $p_0 > 0$ . Let  $(p_n) \in D$ , then

(i) if c = 0 and  $a \ge b$  then  $(p_n)$  is log-concave;

if  $c \ge 0$  and a < b then  $(p_n)$  is not log-concave;

(ii) if  $c \ge 0$  and  $a \ge c \ge b$  then  $(p_n)$  is log-convex;

if  $c \ge 0$  and  $a \ge b > c$  then  $(p_n)$  is not log-convex;

if  $c \ge 0$  and b > a > c then  $(p_n)$  is not log-convex;

if  $c \ge 0$  and b < a = c then  $(p_n)$  is log-convex.

**REMARK** 4.4.2. The absolutely continuous analogue of Theorem 4.4.1 can be obtained by applying the same type of limiting argument as in the proof of Theorem 4.3.2.  $\Box$ 

REMARK 4.4.3. Let *m* in (4.11) be Lebesgue measure on (d,b), and zero otherwise. Then  $r_n = b^n - d^n + a^n$  and  $r_n^2 - r_{n+1} r_{n-1} < 0$  for large *n* if a > b > d > 0, whereas  $(p_n)$  is log-concave by Theorem 4.4.1 (i). Similarly  $(r_n)$  is asymptotically log-concave if 0 = d < b < c < a, whereas  $(p_n)$  is log-convex by Theorem 4.4.1 (ii). Hence, if  $(p_n)$  is log-concave (log-convex) then  $(r_n)$  is not necessarily log-concave (log-convex)

(compare with Theorems 4.2.3 and 4.2.4).

REMARK 4.4.4. Theorem 4.4.1 characterizes D in terms of log-convexity and almost completely in terms of log-concavity. We were not able to prove that if  $a \ge b > c > 0$ , then  $(p_n)$  is log-concave (cf. part (i) of Theorem 4.4.1).

A discrete analogue of an absolutely continuous stable distribution was introduced by Steutel and van Harn (1979) and discussed in Section 1.4. Steutel and van Harn (1979) proved that a distribution  $(p_n)$  is discrete stable with exponent  $\delta$  if and only if its generating function is of the form (cf. Theorem 1.4.3)

 $P(z) = \exp(-\lambda(1-z)^{\delta}), \quad \delta \in (0,1], \ \lambda \ge 0$ .

Taking generating functions on both sides of (4.3) and comparing with the Taylor series expansion of  $-\lambda(1-z)^{\delta}$  one sees that  $(r_n)$  is strictly log-convex and that  $r_0^2 - r_1 < 0$  if and only if  $\delta < 1 - r_0$ . Applying Theorem 4.2.4 to these observations gives

THEOREM 4.4.5. Let  $(p_n)$  be discrete stable with exponent  $\delta$ . Then

 $(p_n)$  is strictly log-convex if and only if  $\lambda < \delta^{-1} - 1$ .

The canonical density h of an absolutely continuous stable distribution on  $(0,\infty)$  is of the form  $cx^{-\delta}$ ,  $\delta \in (0, 1]$ , hence h is log-convex and  $h(0+) = \infty$ . Applying Theorem 4.3.3 we have, rather unexpectedly, that there are no log-convex stable densities on  $(0,\infty)$ .

**Chapter 5** 

# A GENERALIZED SELF-DECOMPOSABILITY

### **5.1 Introduction**

Functional equations have been a helpful tool in representing subsets of the set of infinitely divisible distributions. The definitions of self-decomposable and stable distributions in terms of a functional equation for their characteristic function or probability generating function (cf. Section 1.4) are well-known examples; for other examples see van Harn (1978). O'Connor (1979a) shows that membership in the set of infinitely divisible distributions with unimodal Lévy spectral functions is related with the solutions of the functional equation (cf. equation (1.6))

$$\phi(t) = \phi^{\mathsf{p}}(ct) \phi_{c}(t) \qquad , c \in (0,1), t \in \mathbb{R},$$
(5.1)

with  $\beta = c$  and where  $\phi$  and  $\phi_c$  are characteristic functions. Jurek (1985) calls such characteristic functions shrinking-self-decomposable, or s-self-decomposable for short. All self-decomposable characteristic functions are s-self-decomposable, as follows easily from the fact that self-decomposable characteristic functions are infinitely divisible. Interpolating between (5.1) with  $\beta = c$  and (1.6), O'Connor (1979b) studies equation (5.1) with  $\beta = c^{1-\alpha}$ ,  $\alpha \in (0, 1)$ . In O'Connor (1981) the case  $\alpha \in (1, 3)$  is considered. In this chapter we study the case  $\beta = c^{\alpha}$ ,  $\alpha \in \mathbb{R}$ . This case has also been studied in Jurek (1988) and (1989), where some of the results of this chapter are also proved. For a detailed comparison we refer to Remarks 5.4.11 and 5.4.12.

We consider random variables X on  $\mathbb{R}$ , on  $\mathbb{R}_+$  and on  $\mathbb{N}_0$ . We introduce a one parameter family of functional equations, of the form (5.1), satisfied by the characteristic function, Laplace-Stieltjes transform, or probability generating function of X, depending on whether the random variable has support on  $\mathbb{R}$ , on  $\mathbb{R}_+$  or on  $\mathbb{N}_0$ . Our equations for random variables on  $\mathbb{R}$  include O'Connor's and have as special

## 5.1 Introduction

cases the functional equation defining self-decomposable distributions and s-selfdecomposable distributions. We establish a canonical form for the integral transform satisfying these functional equations, show that these integral transforms are infinitely divisible, and have Lévy spectral functions that are, in an extended sense,  $\alpha$ -unimodal. These results include those of O'Connor and Jurek. It is also shown that this one parameter family of functional equations provides a classification of the set of infinitely divisible random variables.

## 5.2 $\alpha$ -Unimodality

A random variable X with distribution function F and density f is said to be unimodal, with mode at  $x_0$  (not necessarily unique), if f(x) is non-decreasing for  $x < x_0$  and non-increasing for  $x > x_0$ . Throughout this chapter we assume that  $x_0=0$ , i.e., if a function is said to be unimodal (or  $\alpha$ -unimodal) it is understood that its mode is at the origin. Khintchine (1938) showed that X is unimodal (at zero) if X = UY, with U and Y independent and U uniform on (0,1). Olshen and Savage (1970) generalized this concept; a random variable is said to be  $\alpha$ -unimodal (at zero) if it is of the form  $U^{1/\alpha}Y$ , with U and Y independent and U uniformly distributed on (0,1) and  $\alpha \neq 0$ . If Y has distribution function G, then

$$f(x) = \alpha x^{\alpha - 1} \int_{x}^{\infty} v^{-\alpha} dG(v), x \in \mathbb{R}_{+},$$
  
$$f(x) = \alpha |x|^{\alpha - 1} \int_{-\infty}^{x} |v|^{-\alpha} dG(v), x \in \mathbb{R}_{-}$$

This result corresponds to Corollary 2 p. 28 in Olshen and Savage (1970). Hence, f is  $\alpha$ -unimodal if and only if  $|x|^{1-\alpha} f(x)$  is non-decreasing on  $(-\infty, 0)$  and non-increasing on  $(0,\infty)$ . We will use  $\alpha$ -unimodality in connection with Lévy spectral functions, so a more general definition is needed.

DEFINITION 5.2.1. A function C (not necessarily non-negative) is said to be  $\alpha$ unimodal for some  $\alpha \in \mathbb{R}$ , if  $|x|^{1-\alpha} C(x)$  is bounded from below and non-decreasing on  $(-\infty, 0)$  and non-increasing on  $(0, \infty)$  or equivalently if there exists constants  $\lambda_1, \lambda_2 \in \mathbb{R}$  and a function N, non-decreasing on  $(-\infty, 0)$  and  $(0, \infty)$  such that

$$C(x) = \begin{cases} x^{\alpha-1} \left( \int_{x}^{\infty} v^{-\alpha} dN(v) + \lambda_{1} \right), x > 0 \\ x & , \\ |x|^{\alpha-1} \left( \int_{-\infty}^{x} |v|^{-\alpha} dN(v) + \lambda_{2} \right), x < 0 \end{cases}$$
(5.2)

and such that the integrals converge for every  $x \in \mathbb{R} \setminus \{0\}$ .

An analogue of  $\alpha$ -unimodality for discrete distributions on  $\mathbb{N}_0$  is introduced in Abouammoh (1987). An equivalent, but different, definition was given by Steutel (1988). Steutel (1988) calls a random variable discrete  $\alpha$ -unimodal (at zero) if X = U<sup>1/ $\alpha$ </sup>  $\otimes$  Y with U and Y independent and U uniform on (0,1). The multiplication operator  $\otimes$  is defined in Section 1.4. Since we only consider discrete random variables on  $\mathbb{N}_0$ ,  $\alpha$ -unimodality at zero is equivalent to  $\alpha$ -monotonicity. If X and Y are  $\mathbb{N}_0$  valued random variables with probability generating functions G and S, respectively, then

$$G(z) = \alpha \int_{0}^{1} S(1-v(1-z)) v^{\alpha-1} dv .$$

Expanding the integral on the right hand side Steutel (1988) shows that  $(g_n)$  is  $\alpha$ -monotone if and only if  $(n! \Gamma(n+\alpha)^{-1} g_n)$  is non-increasing or equivalently if  $(n+\alpha)g_n \ge (n+1)g_{n+1}$ . We generalize this definition to

DEFINITION 5.2.2. A sequence  $(r_n)$  is said to be  $\alpha$ -monotone, for  $\alpha > 0$ , if  $(n! \Gamma(n+\alpha)^{-1} r_n)$  is bounded from below and is non-increasing, or equivalently if there exists  $\lambda_1 \in \mathbb{R}$  and a non-negative sequence  $(h_n)$  such that

$$r_n = \alpha \frac{\Gamma(n+\alpha)}{n!} \left[ \sum_{k=n}^{\infty} \frac{k!}{\Gamma(k+1+\alpha)} h_k + \lambda_1 \right] \quad , n \in \mathbb{N}_0 .$$
(5.3)

A sequence  $(r_n)$  is said to be zero-monotone if  $r_0 = \lambda$  and  $r_n = 0$  for  $n \ge 1$  (cf. (5.3)), i.e.,  $(r_n)$  is the canonical measure of the Poisson distribution.

REMARK 5.2.3. It is immediate from Definitions 5.2.1 and 5.2.2 that if a function f or a sequence  $(a_n)$  is  $\alpha_0$ -unimodal then it is  $\alpha$ -unimodal for every  $\alpha \ge \alpha_0$ .

## 5.3 Distributions on $\mathbb{N}_0$

The starting point of this chapter was the work on random variables supported by  $\mathbb{N}_0$ . In this special class of random variables the probabilities themselves can be found explicitly and the corresponding Lévy spectral functions are easily computed. This provided a good source of intuition and insight which was very helpful in establishing the results of Section 5.4. In this section this discrete analogue is discussed. We begin with a definition.

## 5.3 Distributions on $\mathbb{N}_0$

DEFINITION 5.3.1 Let the random variable X on  $\mathbb{N}_0$  have distribution  $(p_n)$  and probability generating function P. The function P is said to be  $\alpha$ -self-decomposable for some  $\alpha \in \mathbb{R}$  and to belong to the set  $S_{\alpha}(\mathbb{N}_0)$ , if for every  $c \in (0,1)$  there exists a probability generating function  $P_c$  such that

$$P(z) = P^{c^{*}}(1 - c(1 - z))P_{c}(z) , |z| \le 1.$$
(5.4)

If  $\alpha = 0$  in Definition 5.3.1, then (5.4) provides the functional equation defining discrete self-decomposable probability generating functions as given in Section 1.4.

Let  $(r_n)$  be  $(\alpha+1)$ -monotone,  $\alpha > -1$ ,  $\alpha \neq 0$  and let  $\tilde{H}(z) = \sum_{n=0}^{\infty} h_n / (n+1) z^{n+1}$ . By using

$$\int_{0}^{1} v^{\beta-1} (1-v)^{\gamma-1} dv = \Gamma(\beta) \Gamma(\gamma) \Gamma(\beta+\gamma)^{-1} ,$$

we obtain

$$\tilde{R}(z) := \sum_{n=0}^{\infty} r^n / (n+1) z^{n+1} = \int_0^1 (\tilde{H}(1-\nu(1-z)) - \tilde{H}(1-\nu)) \nu^{\alpha-1} d\nu + \lambda_2 - \lambda_2 (1-z)^{-\alpha},$$
(5.5)

where  $\lambda_2 = \lambda_1 (1+\alpha) \Gamma(1+\alpha) (-\alpha)^{-1}$ . Hence  $(r_n)$  in (5.3) is a canonical measure, i.e.,  $r_n \ge 0$  and  $\sum r_n / (n+1) < \infty$  (cf. Theorem 1.3.4), if and only if  $(h_n)$  is, with  $\lambda_1 \ge 0$ ,  $\lambda_1 = 0$  for  $\alpha \ge 0$ . Similarly if  $\alpha = 0$ .

Suppose  $(p_n)$  has canonical measure  $(r_n)$  with  $(r_n)$   $(\alpha+1)$ -monotone,  $\alpha > -1$  and suppose  $(q_n)$  is an infinitely divisible distribution with canonical measure  $(h_n)$ . By (5.5) and Theorem 1.3.4,

$$\ln P(z) = R(z) + \ln P(0) = R(z) - R(1)$$
  
=  $\int_{0}^{1} (\tilde{H}(1 - v(1 - z)) - \tilde{H}(1)) v^{\alpha - 1} dv - \lambda_2 (1 - z)^{-\alpha}$   
=  $\int_{0}^{1} \ln Q (1 - v(1 - z)) v^{\alpha - 1} dv - \lambda_2 (1 - z)^{-\alpha}$ .

We now prove

~

THEOREM 5.3.2. Let  $\alpha \in (-1,\infty)$  and let P be a probability generating function. The following statements are equivalent.

(i) *P* is  $\alpha$ -self-decomposable;

- (ii)  $\ln P(z) = \int_{0}^{1} \ln Q(1-v(1-z))v^{\alpha-1} dv \lambda(1-z)^{-\alpha}, \ \lambda \ge 0 \text{ and } Q \text{ is a unique infinitely divisible probability generating function;}$
- (iii)  $(p_n)$  is infinitely divisible and its canonical measure  $(r_n)$  is  $(\alpha+1)$ -monotone.

Furthermore,  $\lambda$  in (ii) is zero if  $\alpha \ge 0$  and  $P_c$  is infinitely divisible for every  $c \in (0,1)$  if P is  $\alpha$ -self-decomposable.

PROOF. (ii)  $\Leftrightarrow$  (iii) has been proved above. We now prove (i)  $\Leftrightarrow$  (ii). Let *P* satisfy (5.4), let r > 0 and  $c_n \in (0,1)$  for  $n \in \mathbb{N}_+$  such that  $r (1-c_n)^{-1} \in \mathbb{N}_0$ , then

$$\ln Q_{r,n}(z) := \ln (P_{c_n}(z))^{r(1-c_n)^{-1}}$$
  
=  $r \left\{ \frac{\ln P(z) - \ln P(1-c_n(1-z))}{1-c_n} + \frac{1-c_n^{\alpha}}{1-c_n} \ln P(1-c_n(1-z)) \right\}$ 

is the logarithm of a probability generating function. Let  $c_n$  be such that  $c_n \uparrow 1$  as  $n \to \infty$ , then

$$\ln Q_r(z) := \lim_{n \to \infty} \ln Q_{r,n}(z) = r \left( -(1-z) P'(z) / P(z) + \alpha \ln P(z) \right).$$
(5.6)

Since  $Q_r(z) \rightarrow 1$  as  $z \rightarrow 1$  (cf. Lemma 1, Steutel and van Harn (1979)), by Theorem 1.3.7,  $Q_r$  is a probability generating function for every r > 0, and thus  $Q \coloneqq Q_1$  is an infinitely divisible probability generating function. Equation (5.6) gives rise to the following differential equation,

$$\ln Q(z) = -(1-z) \frac{d}{dz} \ln P(z) + \alpha \ln P(z),$$

which has (ii) as unique solution.

Conversely, if P is as in (ii), then P satisfies (5.4) with

$$\ln P_c(z) = \int_c^1 \ln Q(1 - v(1 - z)) v^{\alpha - 1} dv \quad .$$
(5.7)

Let  $(p_n(c))$  be the probability distribution corresponding to  $P_c$  and define  $(r_n(c))$  using  $(p_n(c))$  in (1.2). Then

$$\sum_{n=0}^{\infty} r_n(c) z^n \coloneqq R_c(z) = \frac{d}{dz} \ln P_c(z)$$
$$= \int_c^1 H(1 - v(1 - z)) v^\alpha dv$$

which has positive coefficients. Hence  $P_c$  is infinitely divisible.

COROLLARY 1. *P* is discrete stable with exponent  $\delta$  (cf. Theorem 1.4.3), then  $P \in S_{\alpha}(\mathbb{N}_0)$  for every  $\alpha \ge -\delta$ .

COROLLARY 2. The Poisson distribution is the only distribution in  $S_{-1}(\mathbb{IN}_0)$ , and its canonical measure is  $\alpha$ -monotone with  $\alpha = 0$ .

Note that for  $\alpha > 0$  the sequence  $(r_n)$  is  $(\alpha+1)$ -monotone if and only if  $(r_n / (n+1))$  is  $\alpha$ -monotone. Since all log-convex sequences and all Hausdorff moment sequences are non-increasing we have (cf. Theorem 3.3.8 and Theorem 4.2.4)

COROLLARY 3. If  $(r_n)$  is log-convex, then  $(p_n)$  is 0-self-decomposable.

COROLLARY 4. If  $(p_n)$  is a mixture of geometric distributions, then  $(p_n)$  is 1-self-decomposable.

REMARK 5.3.3. It follows from the proof of Theorem 5.3.2 that a necessary and sufficient condition for the convergence of the integral in Theorem 5.3.2 (ii) is that the canonical measure  $(h_n)$  of the infinitely divisible probability generating function Q satisfies

$$\sum_{k=0}^{\infty} \frac{(k+1)!}{\Gamma(k+2+\alpha)} \frac{h_k}{k+1} < \infty,$$
(5.8)

for  $\alpha \in (-1,0)$  (recall that  $(r_n)$  is  $(\alpha+1)$ -monotone). By Stirling's formula (5.8) is equivalent to

$$\sum_{k=0}^{\infty} (k+1)^{-\alpha} \frac{h_k}{k+1} < \infty.$$

If  $\alpha = 0$  then it is necessary and sufficient that

$$\sum_{k=0}^{\infty} \sum_{n=0}^{k} \frac{1}{n+1} \frac{h_k}{k+1} < \infty,$$

or equivalently

$$\sum_{k=0}^{\infty} \ln (k+1) \frac{h_k}{k+1} < \infty.$$

Berg and Forst (1983) introduced the set of *n*-times self-decomposable probability generating functions  $L_n(\mathbb{N}_0)$  inductively by letting  $L_0(\mathbb{N}_0) = S_0(\mathbb{N}_0)$  and (cf. (5.4))

 $L_{n+1}(\mathbb{N}_0) = \{ P \in L_n(\mathbb{N}_0) \mid P_c \in L_n(\mathbb{N}_0), \ c \in (0,1) \} \,.$ 

Hence  $L_1(\mathbb{N}_0)$  is the subset of  $S_0(\mathbb{N}_0)$ , where  $P_c$  is 0-self-decomposable for every  $c \in (0, 1)$ . Similarly, let  $S_{\alpha}^0(\mathbb{N}_0) := S_{\alpha}(\mathbb{N}_0)$  and let

$$S_{\alpha}^{n+1}(\mathbb{N}_{0}) = \{ P \in S_{\alpha}^{n}(\mathbb{N}_{0}) \mid P_{c} \in S_{\alpha}^{n}(\mathbb{N}_{0}), c \in (0,1) \} .$$

We conclude this section with a characterization of the probability generating functions in  $S^n_{\alpha}(\mathbb{N}_0)$ .

THEOREM 5.3.4. Let  $P \in S_{\alpha}(\mathbb{N}_0)$  with  $P_c$  given by (5.4) and Q by Theorem 5.3.2. Then  $Q \in S^n_{\alpha}(\mathbb{N}_0)$  if and only if  $P \in S^{n+1}_{\alpha}(\mathbb{N}_0)$ ,  $n \in \mathbb{N}_0$ .

**PROOF.** Suppose  $Q \in S^n_{\alpha}(\mathbb{N}_0)$ . Observe that

$$\ln P_{c}(z) = \ln P(z) - c^{\alpha} \ln P(1 - c(1 - z))$$
  
= 
$$\int_{0}^{1} \left[ \ln Q(1 - v(1 - z)) - c^{\alpha} \ln Q(1 - cv(1 - z)) \right] v^{\alpha - 1} dv$$
  
= 
$$\int_{0}^{1} \ln Q_{c}(1 - v(1 - z)) v^{\alpha - 1} dv ,$$

for some probability generating function  $Q_c$ . Obviously, if Q can be decomposed n+1-times in this way, then so can P. Hence  $P \in S^{n+1}_{\alpha}(\mathbb{N}_0)$ .

Conversely, if  $P \in S_{\alpha}^{n+1}(\mathbb{N}_0)$ , then  $P_c \in S_{\alpha}^n(\mathbb{N}_0)$  for every  $c \in (0, 1)$ . By the proof of Theorem 5.3.2

$$\ln Q(z) = \lim_{c \downarrow 1} (1-c)^{-1} \ln P_c(z).$$

Obviously  $S^n_{\alpha}(\mathbb{N}_0)$  is closed under limits and so  $Q \in S^n_{\alpha}(\mathbb{N}_0)$ .

## 5.4 Distributions on IR

Throughout this section we will understand by the derivative of a Lévy spectral function M a right continuous function M' defined by

$$M'(x+)=M_+'(x),$$
  
 $M'(x-)=M_-'(x),$ 

.

.

where  $M_+$  and  $M_-$  are respectively the right and left derivatives,  $f(x+)=\lim_{y \downarrow x} f(y)$  and  $f(x-)=\lim_{y \uparrow x} f(y)$ . The assumption of right continuity of M' is non-essential and is assumed for uniqueness only. The results of this section are also true if we were to make M' left continuous or set M' equal to any (linear) combination of  $M_+$  and  $M_-$  at points where  $M_+ \neq M_-$ . 5.4 Distributions on  $\mathbb R$ 

We begin with a definition

DEFINITION 5.4.1. Let the random variable X on  $\mathbb{R}$  have distribution function F and characteristic function  $\phi$ . The function  $\phi$  is said to be  $\alpha$ -self-decomposable for some  $\alpha \in \mathbb{R}$ , if for every  $c \in (0, 1)$  there exists a characteristic function  $\phi_c$  such that

$$\phi(t) = \phi^{c^{*}}(ct) \phi_{c}(t) \qquad , t \in \mathbb{R}.$$
(5.8)

For  $\alpha = 0$  we have the functional equation defining self-decomposable characteristic functions and for  $-2 < \alpha \le 1$  the equations corresponding to O'Connors class  $L_{1-\alpha}$ . Our results include those of O'Connor, but our proofs differ. For a detailed comparison we refer to Remark 5.4.11.

We will use the following two lemmas; the proof of the first lemma is similar to the first part of the proof of Theorem 5.11.1 in Lukacs (1970). The discrete counterpart of the second lemma is discussed in the second paragraph of Section 5.3.

LEMMA 5.4.2. If  $\phi$  satisfies (5.8), then  $\phi$  has no real zeros.

PROOF. If  $\phi$  satisfies (5.8), then  $\psi = |\phi|^2$  is a characteristic function which also satisfies (5.8). Suppose  $\psi$  has zeros. Since  $\psi$  is continuous and  $\psi(0) = 1$ , there exists a  $t_0$  such that  $\psi(t_0) = 0$  while  $\psi(t) \neq 0$  for  $|t| < t_0$ . It follows from (5.8) that  $\psi_c(t_0) = 0$  while  $\psi(t) \neq 0$  for  $|t| < t_0$ . By Theorem 4.1.2 on p. 69 of Lukacs (1970), with n = 1 and  $t = t_0/2$ , we have

 $4(1-\psi_c(t_0/2)) \ge 1-\psi_c(t_0)=1$ .

Since  $\psi_c(t_0/2) = \psi(t_0/2) / \psi^{c^{\alpha}}(ct_0/2)$  is continuous in *c*, we obtain a contradiction by choosing *c* sufficiently close to 1. So  $\psi$  and  $\psi_c$ , and hence also  $\phi$  and  $\phi_c$ , have no real zeros.

LEMMA 5.4.3. Let  $\alpha > -2$  and let M be such that M'(x) is  $\alpha$ -unimodal, i.e.,

$$M'(x) = \begin{cases} x^{\alpha-1} \left( \int_{x}^{x} v^{-\alpha} dN(v) + \lambda_{1} \right), x > 0 \\ x + 1 \left( \int_{x}^{x} |v|^{-\alpha} dN(v) + \lambda_{2} \right), x < 0 \end{cases}$$
(5.9)

with the integrals converging for every  $x \in \mathbb{R} \setminus \{0\}$  and  $\int_{1}^{\infty} \ln v \, dN(v) < \infty$  if  $\alpha = 0$ . Then M is a Lévy spectral function if and only if N can be chosen to be a Lévy spectral function and  $\lambda_1$ ,  $\lambda_2 \in [0,\infty)$  and  $\lambda_1 = \lambda_2 = 0$  for  $\alpha \ge 0$ . **PROOF.** First let *M* be a Lévy spectral function satisfying (5.9). We have to show that *N* satisfies the requirements (ii) and (iii) of Theorem 1.3.2 and that  $\lambda_1$  and  $\lambda_2$  satisfy the requirements of the lemma.

By Theorem 1.3.2 (i) and (ii),  $M'(x) \ge 0$  and  $M(\infty) = 0$ . Since  $\lim_{x\to\infty} x^{1-\alpha} M'(x) = \lambda_1$  we have  $\lambda_1 \in [0,\infty)$ . Similarly  $\lambda_2 \in [0,-\infty)$ . From (5.9) it follows that for y > 0

$$-M(y) = \int_{y}^{\infty} M'(x) \, dx = \int_{y}^{\infty} x^{\alpha - 1} \left( \int_{x}^{\infty} v^{-\alpha} \, dN(v) + \lambda_1 \right) \, dx \,. \tag{5.10}$$

Since  $0 \le -M(y) < \infty$ , we must have that  $\int_{1}^{\infty} \lambda_1 x^{\alpha-1} dx < \infty$  and hence that  $\lambda_1 = 0$  for  $\alpha \ge 0$ . A similar reasoning for y < 0 yields  $\lambda_2 = 0$  for  $\alpha \ge 0$ .

Requirement (ii) of Theorem 1.3.2. Let  $\alpha \leq 0$ . By (5.9)

$$\infty > |M'(x)| \ge |x^{-1} \int_{x}^{\infty} dN(v) + \lambda_1 x^{\alpha - 1} |,$$

hence  $|N(\infty)| < \infty$ . Similarly  $|N(-\infty)| < \infty$ . Let  $\alpha > 0$ . Observe that for y > 0 (cf. (5.10))

$$-M(y) = \alpha^{-1} \int_{y}^{\infty} (1 - (y/v)^{\alpha}) dN(v) .$$

By (5.9)  $\int_{v}^{\infty} (y/v)^{\alpha} dN(v) < \infty$ , so  $\int_{y}^{\infty} dN(v)$  converges and hence  $|N(\infty)| < \infty$ . Similarly  $|N(-\infty)| < \infty$ .

Requirement (iii) of Theorem 1.3.2. Let  $0 < \delta < \epsilon$ . From (5.9) it follows for  $\alpha > -2$ , that

$$\int_{\delta}^{\varepsilon} u^2 dM(u) = (\alpha + 2)^{-1} \left[ \left( \varepsilon^{\alpha + 2} - \delta^{\alpha + 2} \right) \left( \int_{\varepsilon}^{\infty} v^{-\alpha} dN(v) + \lambda_1 \right) - \delta^{\alpha + 2} \int_{\delta}^{\varepsilon} v^{-\alpha} dN(v) + \int_{\delta}^{\varepsilon} v^2 dN(v) \right].$$
(5.11)

Since  $\delta^{\alpha+2} \le v^{\alpha+2}$  for  $v \in (\delta, \varepsilon)$ , the right hand side of (5.11) is bounded from below by

$$(\alpha+2)^{-1} \left[ \left( \varepsilon^{\alpha+2} - \delta^{\alpha+2} \right) \left( \int_{\varepsilon}^{\infty} v^{-\alpha} dN(v) + \lambda_1 \right) \right].$$
 (5.12)

Let  $\delta = 0$ , then the left hand side of (5.11) tends to zero as  $\varepsilon \to 0$  and so (5.12) tends to zero as  $\varepsilon \to 0$ . By letting  $\delta \to 0$  in (5.11) it follows that N satisfies condition (iii) of Theorem 1.3.2. Similarly for  $0 > \delta > \varepsilon$ .

Conversely, suppose N is a Lévy spectral function such that the integrals in (5.9) converge. We have to show that M satisfies requirements (ii) and (iii) of Theorem

#### 5.4 Distributions on $\mathbb{R}$

1.3.2. Assume with out loss of generality that  $\lambda_1 = \lambda_2 = 0$ .

Requirement (ii) of Theorem 1.3.2. Let z > y > 0 and let  $\alpha \neq 0$ . By (5.9) we have that

$$0 \le M(z) - M(y) = \int_{y}^{z} M'(x) dx$$
  
=  $\alpha^{-1} [z^{\alpha} \int_{z}^{\infty} v^{-\alpha} dN(v) - y^{\alpha} \int_{y}^{\infty} v^{-\alpha} dN(v) + N(z) - N(y)]$   
 $\le \alpha^{-1} [z^{\alpha} \int_{z}^{\infty} v^{-\alpha} dN(v) - N(y)].$ 

For  $\alpha < 0$ ,  $\lim_{z \to \infty} z^{\alpha} \int_{z}^{\infty} v^{-\alpha} dN(v) = 0$ , and so  $|M(\infty)| < \infty$ . If  $\alpha > 0$ , then  $\lim_{z \to \infty} z^{\alpha} \int_{z}^{\infty} v^{-\alpha} dN(v) = \lim_{z \to \infty} z^{\alpha} \int_{z}^{\infty} (z/v)^{\alpha} dN(v) \le \lim_{z \to \infty} N(z) = 0$ ,

and hence  $|M(\infty)| < \infty$ . For  $\alpha = 0$ , observe that

$$0 \le M(z) - M(y) = \int_{y}^{z} \ln v \, dN(v) + N(y) \ln y - N(z) \ln z \; .$$

By the condition on N in the lemma for  $\alpha = 0$ ,  $\lim_{z\to\infty} N(z) \ln z = 0$  and hence  $|N(\infty)| < \infty$ . A similar argument for z < y < 0 shows that  $|M(-\infty)| < \infty$ .

Requirement (iii) of Theorem 1.3.2. Let  $0 < \delta < \varepsilon$ . The right hand side of (5.11) is bounded from above by

$$(\alpha+2)^{-1} \left[ \varepsilon^{\alpha+2} \int_{\varepsilon}^{\infty} v^{-\alpha} dN(v) + \int_{\delta}^{\varepsilon} v^2 dN(v) \right].$$
(5.13)

Letting  $\delta \to 0$  in (5.13) and (5.11) we see that  $\int_0^{\varepsilon} u^2 dM(v) < \infty$ . A similar argument holds for  $0 > \delta > \varepsilon$ .

COROLLARY 1. Let N be a Lévy spectral function such that the integrals in (5.9) converge for  $\alpha \neq 0$  and  $\int_{1}^{\infty} \ln v \, dN(v) < \infty$  if  $\alpha = 0$ . Then,

(i) 
$$\lim_{x \to \infty} x^{\alpha} \int_{x}^{\infty} v^{-\alpha} dN(v) = 0 = \lim_{x \to \infty} |x|^{\alpha} \int_{-\infty}^{x} |v|^{-\alpha} dN(v), \ \alpha \in (-\infty, \infty);$$
  
(ii) 
$$\lim_{x \to 0^{+}} x^{\alpha+2} \int_{x}^{\infty} v^{-\alpha} dN(v) = 0 = \lim_{x \to 0^{-}} |x|^{\alpha+2} \int_{-\infty}^{x} |v|^{-\alpha} dN(v), \ \alpha \in (-2, \infty).$$

**PROOF.** This corollary is proved for  $\alpha \in (0, 1]$  in O'Connor (1979b). Part (ii) follows from the proof of Lemma 5.4.3. Part (i) is trivial for  $\alpha < 0$  since the integrals always

converge. For  $\alpha \ge 0$  part (i) is evident from

$$|x^{\alpha}\int_{x}^{\infty}v^{-\alpha} dN(v)| \leq |N(x)|.$$

REMARK 5.4.4. If M'(x) is  $\alpha$ -unimodal then for  $0 < x \le \varepsilon$ ,

$$x^2 M'(x) \ge x^{\alpha+1} \int_{\varepsilon}^{\infty} v^{-\alpha} dN(v)$$
.

Integrating over  $(0,\varepsilon)$  it follows, by Theorem 1.3.2 (iii), that there are no  $\alpha$ -unimodal Lévy spectral functions with  $\alpha \leq -2$ .

In Section 1.4 we defined the linear operator  $T_r$  on  $\mathbb{R}$  by  $T_r(x)=rx$  and showed that it was closely related with the random variables having 0-self-decomposable characteristic functions. Let  $T_r$  operate on set functions by

$$T_r M(B) := M(T_r^{-1}B) = M(\{x \in \mathbb{R} \mid T_c(x) \in B\}),$$

for any Borel set B. The following lemma gives a connection between  $T_r$  and the notion of  $\alpha$ -unimodality.

LEMMA 5.4.5. Let  $\alpha \in \mathbb{R}$ ,  $\beta_{\varepsilon}$  be the set of Borel sets on  $\mathbb{R} \setminus (-\varepsilon, \varepsilon)$ , for any  $\varepsilon > 0$ , and M a Lévy spectral function. The following statements are equivalent.

- (i)  $M(B) \ge c^{\alpha} T_c M(B)$ ,  $c \in (0,1)$ ,  $B \in \beta_{\varepsilon}$ ;
- (ii) M has left and right derivatives on  $\mathbb{R}\setminus\{0\}$  and M'(x) is  $\alpha$ -unimodal.

**PROOF.** First we prove (i)  $\Rightarrow$  (ii). Let  $\alpha \in \mathbb{R}$ , fix  $\varepsilon > 0$  and let  $B = (a, b), 0 < \varepsilon \le a < b$ . Then (i) is equivalent to

$$c^{\alpha}(M(b/c) - M(a/c)) \le M(b) - M(a).$$
 (5.14)

If  $\alpha \le 0$ , then *M* is convex and hence by Theorem A, p. 4, Roberts and Varberg (1973), *M* is absolutely continuous on  $(\varepsilon, \infty)$ . Suppose  $\alpha > 0$  and let  $w(x) := M(x^{1/\alpha})$ . From (5.14) with  $x = (b/c)^{\alpha}$ ,  $y = (a/c)^{\alpha}$ ,  $x' = b^{\alpha}$  and  $y' = a^{\alpha}$  it follows that

$$\frac{w(x)-w(y)}{x-y} \le \frac{w(x')-w(y')}{x'-y'}$$

Hence w is convex. By Theorem A, p. 4, Roberts and Varberg (1973), w and hence also M is absolutely continuous on  $(\varepsilon, \infty)$ . Observe that

$$\int_{B} M'(x) \, dx = M(B) \ge c^{\alpha} T_c \, M(B) = c^{\alpha - 1} \, \int_{B} T_c M'(x) \, dx \,, \tag{5.15}$$

with B = (a, b). Differentiating (5.15) with respect to a and multiplying both sides by  $a^{1-\alpha}$ , we see that  $a^{1-\alpha}M'(a)$  is non-increasing on  $\mathbb{R}_+$ . Similarly for  $b < a \le \varepsilon < 0$  we obtain that  $a^{1-\alpha}M'(a)$  is non-decreasing on  $\mathbb{R}_-$ 

The converse is proved for  $B = (a, b), 0 < \varepsilon \le a < b$ , by observing that (cf. (5.9))

$$M(B) = \int_{B} M'(x) dx = \int_{B} x^{\alpha - 1} \left( \int_{x}^{\infty} v^{-\alpha} dN(v) \right) dx$$
$$= c^{\alpha} \int_{T_{c}^{-1}B} x^{\alpha - 1} \left( \int_{cx}^{\infty} v^{-\alpha} dN(v) \right) dx$$
$$\ge c^{\alpha} \int_{T_{c}^{-1}B} M'(x) dx = c^{\alpha} T_{c} M(B) .$$

If  $M(\mathbb{R}) < \infty$ , then by Lemma 5.4.5,  $M(\mathbb{R}) \ge c^{\alpha} M(\mathbb{R}), c \in (0, 1)$ . Hence

COROLLARY 1. If M is a Lévy spectral function having left and right derivatives on  $\mathbb{R}\setminus\{0\}$  and where M'(x) is  $\alpha$ -unimodal,  $\alpha < 0$ , then M must be unbounded.

Before proving a representation theorem for  $\alpha$ -self-decomposable characteristic functions we will prove a preparatory lemma, whose counterpart for discrete distributions is discussed in the third paragraph of Section 5.3.

LEMMA 5.4.6. Let  $\phi$  be an infinitely divisible characteristic function with Lévy spectral function M having left and right derivatives on  $\mathbb{R} \setminus \{0\}$  and such that M'(x) is  $\alpha$ -unimodal. Then,

(i) if  $\alpha \in [0, \infty)$  there exists an infinitely divisible characteristic function  $\gamma$  such that

$$\ln\phi(t) = \int_0^1 \ln\gamma(vt) v^{\alpha-1} dv;$$

(ii) if  $\alpha \in (-2, 0)$  there exists an infinitely divisible characteristic function  $\gamma$  and a stable characteristic function  $\phi_{\text{STABLE}(-\alpha)}(t)$ , possibly degenerate, such that

$$\ln \phi(t) = \int_{0}^{1} \ln \gamma(vt) v^{\alpha-1} dv + \ln \phi_{\text{STABLE}(-\alpha)}(t) .$$

PROOF. We partly follow the proof of Theorem 2 in Alf and O'Connor (1977). The proof of (i) is very similar to that of (ii) for  $\alpha \in (-1,0)$ , so we only prove (ii). For  $\phi$  as given in the lemma, there exists a Lévy spectral function N and  $\lambda_1$ ,  $\lambda_2 \in [0, \infty)$  such that N and M are related by (5.9). Define the Lévy spectral function  $M_1$  by

$$M_{1}(x) = \begin{cases} M(x) - \alpha^{-1} \lambda_{1} x^{\alpha} , x > 0 \\ M(x) + \alpha^{-1} \lambda_{2} |x|^{\alpha} , x < 0 \end{cases}$$
(5.16)

Hence  $M_1$  is an  $\alpha$ -unimodal Lévy spectral function with the same N in (5.9) as M, but with  $\lambda_1 = \lambda_2 = 0$ . Define an infinitely divisible characteristic function  $\gamma$  using N for M in Theorem 1.3.2. We now wish to evaluate the integral in (ii) and show, by choosing  $a_{\gamma}$  and  $\sigma_{\gamma}$  appropriately, that it is equal to  $\ln \phi(t)$ . The manner in which  $a_{\gamma}$  must be selected is closely related to the proof of Theorem 5.7.3 in Lukacs (1970). We consider two cases.

CASE I,  $\alpha \in (-1,0)$ . For ease of notation we define for x > 0

$$L(t, x) = x^{\alpha} \int_{0}^{1} v^{\alpha - 1} k(vt, x) dv = \int_{0}^{x} (e^{itv} - 1)v^{\alpha - 1} dv - \frac{ix^{\alpha + 1}t}{(\alpha + 1)(1 + x^{2})}.$$
 (5.17)

From the first equality it follows that  $L(t,x)/x^{\alpha+2}$  is bounded as  $x \to 0$ . Note that

$$\frac{\partial L(t,x)}{\partial x} = x^{\alpha-1} \left[ k(t,x) + 2ix^3 t/(\alpha+1)(1+x^2)^2 \right].$$

Observe that

$$\int_{0}^{1} \ln \gamma(vt) v^{\alpha-1} dv - i(\alpha+1)^{-1} a_{\gamma} t + \frac{1}{2}(\alpha+2)^{-1} \sigma_{\gamma}^{2} t^{2}$$

$$= \int_{-\infty}^{\infty} \int_{0}^{1} v^{\alpha-1} k(vt, x) dv dN(x)$$

$$= \int_{0}^{\infty} L(t, x) x^{-\alpha} dN(x) + \int_{0}^{\infty} L(-t, x) x^{-\alpha} d-N(-x)$$

$$= \lim_{\epsilon \to 0} \left[ \frac{-L(t, x)}{x^{\alpha+2}} x^{\alpha+2} \int_{x}^{\infty} v^{-\alpha} dN(v) \int_{x=\epsilon}^{\alpha} + \int_{\epsilon}^{\infty} \frac{\partial L(t, x)}{\partial x} \int_{x}^{\infty} v^{-\alpha} dN(v) dx \right]$$

$$+ \lim_{\epsilon \to 0} \left[ \frac{-L(-t, x)}{x^{\alpha+2}} x^{\alpha+2} \int_{x}^{\infty} v^{-\alpha} d-N(-v) \int_{x=\epsilon}^{\infty} + \int_{\epsilon}^{\infty} \frac{\partial L(-t, x)}{\partial x} \int_{x}^{\infty} v^{-\alpha} d-N(-v) dx \right]$$

$$= \int_{\mathbb{R}\setminus\{0\}} k(t, x) dM_{1}(x) + 2(\alpha+1)^{-1} \int_{\mathbb{R}\setminus\{0\}} itx^{3}/(1+x^{2})^{2} dM_{1}(x) , \qquad (5.18)$$

where the second equality is obtained by integrating by parts, and the final expression by observing that  $L(t, x)/x^{\alpha+2}$  is bounded as  $x \to 0$  and using Corollary 1 to Lemma 5.4.3. Let  $a_{\gamma}$  and  $\sigma_{\gamma}$  be defined by

5.4 Distributions on  $\mathbb{R}$ 

$$(\alpha+1)^{-1} a_{\gamma} + 2 (\alpha+1)^{-1} \int_{\mathbb{R}\setminus\{0\}} x^3 / (1+x^2)^2 dM_1(x) = a_{\phi}, \qquad (5.19a)$$

$$\frac{1}{2}(\alpha+2)^{-2}\sigma_{\gamma}^2 = \frac{1}{2}\sigma_{\phi}^2$$
. (5.19b)

From (5.18) we obtain (cf. Theorem 1.4.1 and (5.16))

$$\int_{0}^{1} \ln \gamma(vt) v^{\alpha-1} dv = i (\alpha+1)^{-1} a_{\gamma} t + \frac{1}{2} (\alpha+2)^{-2} \sigma_{\gamma}^{2} t^{2} + 2(\alpha+1)^{-1} \int_{\mathbb{R}\setminus\{0\}} itx^{3}/(1+x^{2})^{2} dM_{1}(x) + \int_{\mathbb{R}\setminus\{0\}} k(t,x) dM_{1}(x) = ia_{\phi} t + \frac{1}{2} \sigma_{\phi}^{2} t^{2} + \int_{\mathbb{R}\setminus\{0\}} k(t,x) dM(x) - \ln \phi_{\text{STABLE}(-\alpha)}(t) .$$

Hence  $\phi$  is of the desired form.

CASE II,  $\alpha \in (-2, -1]$ . For ease of notation we define

$$L^{*}(t, x) = \int_{0}^{x} (e^{itv} - 1 - itv)v^{\alpha - 1} dv \quad .$$
 (5.20)

Note that  $L^*(t, x) / x^{\alpha+2}$  is bounded as  $x \to \infty$  and that

$$\frac{\partial L^*(t,x)}{\partial x} = x^{\alpha-1} \left[ k(t,x) - itx^3/(1+x^2) \right].$$

Since the integrals in (5.9) converge we can let

$$a_{\gamma} = -\int_{\mathbb{R}\setminus\{0\}} v^3 / (1 + v^2) dN(v) \quad . \tag{5.21}$$

Also define  $\sigma_{\gamma}^2$  by (5.19b). Analogous to Case I, we have that

$$\int_{0}^{1} \ln \gamma(vt) v^{\alpha-1} dv + \frac{1}{2} \sigma_{\phi}^{2} t^{2}$$

$$= \int_{\mathbb{R}\setminus\{0\}} \int_{0}^{1} (e^{itvx} - 1 - itvx) v^{\alpha-1} dv dN(x)$$

$$= \int_{0}^{\infty} L^{*}(t, x) x^{-\alpha} dN(x) + \int_{0}^{\infty} L^{*}(-t, x) x^{-\alpha} d-N(-x)$$

$$= \lim_{\varepsilon \to 0} \left[ \frac{-L^{*}(t, x)}{x^{\alpha+2}} x^{\alpha+2} \int_{x}^{\infty} v^{-\alpha} dN(v) \right]_{x=\varepsilon} + \int_{\varepsilon}^{\infty} \frac{\partial L^{*}(t, x)}{\partial x} \int_{x}^{\infty} v^{-\alpha} dN(v) dx \right]$$

$$+ \lim_{\varepsilon \to 0} \left[ \frac{-L^{*}(-t, x)}{x^{\alpha+2}} x^{\alpha+2} \int_{x}^{\infty} v^{-\alpha} d-N(-v) \int_{x=\varepsilon}^{\infty} + \int_{\varepsilon}^{\infty} \frac{\partial L^{*}(-t, x)}{\partial x} \int_{x}^{\infty} v^{-\alpha} d-N(-v) dx \right]$$

$$= \int_{\mathbb{R}\setminus\{0\}} k(t, x) dM_{1}(x) - \int_{\mathbb{R}\setminus\{0\}} itx^{3}/(1+x^{2}) dM_{1}(x) .$$

Rewriting as in Case I, we see that  $\phi$  has the desired form.

We are now ready to prove the two main theorems of this section (and in the author's opinion also two of the nicest results in this monograph).

THEOREM 5.4.7. Let  $\alpha \ge 0$  and let  $\phi$  be a characteristic function. The following statements are equivalent.

- (i)  $\phi$  is  $\alpha$ -self-decomposable, and  $\phi'$  exists on  $\mathbb{R} \setminus \{0\}$  with  $\lim_{t \to 0} t \phi'(t) = 0$ ;
- (ii) There exists a unique infinitely divisible characteristic function  $\gamma$  such that

$$\ln\phi(t) = \int_0^1 \ln\gamma(vt) v^{\alpha-1} dv ;$$

- (iii)  $\phi$  is infinitely divisible with Lévy spectral function M having left and right derivatives and such that M'(x) is  $\alpha$ -unimodal;
- (iv)  $\phi$  is  $\alpha$ -self-decomposable and  $\phi_c$  is an infinitely divisible characteristic function for every  $c \in (0, 1)$ .

PROOF. First we prove (i)  $\iff$  (ii). Let r > 0 and  $c_n \in (0, 1)$  for  $n \in \mathbb{N}_+$  such that  $r(1-c_n)^{-1} \in \mathbb{N}_0$ . By (5.8)

$$\ln \gamma_{r,n}(t) := \ln \left(\phi_{c_n}(t)\right)^{r(1-c_n)^{-1}} = r \left\{ \frac{\ln \phi(t) - \ln \phi(c_n t)}{1-c_n} + \frac{1-c_n^{\alpha}}{1-c_n} \ln \phi(c_n t) \right\}, \quad (5.22)$$

is the logarithm of a characteristic function. Let  $c_n$  be such that  $c_n \uparrow 1$  as  $n \to \infty$ , then

$$\ln \gamma_r(t) := \lim_{n \to \infty} \ln \gamma_{r,n}(t) = r \left( t \phi'(t) / \phi(t) + \alpha \ln \phi(t) \right).$$
(5.23)

Since  $\gamma_r(t) \to 1$  as  $t \to 0$ , by Theorem 1.3.7,  $\gamma_r$  is a characteristic function for every r > 0, and thus  $\gamma := \gamma_1$  is an infinitely divisible characteristic function. Equation (5.23) gives rise to the following differential equation:

$$t^{\alpha-1}\ln\gamma(t) = t^{\alpha} \frac{d}{dt}\ln\phi(t) + \alpha t^{\alpha-1}\ln\phi(t) = \frac{d}{dt}t^{\alpha}\ln\phi(t).$$
(5.24)

Hence  $\phi$  is given by (ii). Conversely, if  $\phi$  is as in (ii), then  $\phi'$  exists on  $\mathbb{R}\setminus\{0\}$  with  $t\phi'(t) \to 0$  as  $t \to 0$  and  $\phi$  satisfies (5.8) with

$$\ln \phi_c(t) = \int_{c}^{1} \ln \gamma(vt) \, v^{\alpha - 1} \, dv \quad . \tag{5.25}$$

The characteristic function  $\phi_c$  is infinitely divisible and so (ii)  $\Rightarrow$  (iv) is also proved.

Suppose (iv) is satisfied. Since  $\phi = \lim_{c \to 0} \phi_c$ , by Theorem 1.3.7,  $\phi$  is infinitely divisible with some Lévy spectral function M. Since  $\phi$  and  $\phi_c$  are related by (5.8), the Lévy spectral function  $M_c$  of  $\phi_c$  is given by

5.4 Distributions on  $\mathbb{R}$ 

$$M_c(x) = M(x) - c^{\alpha} M(x/c)$$

(cf. Theorem 1.3.2 and Lukacs (1970), p. 163).  $M_c$  is non-decreasing by Theorem 1.3.2 (i), hence from Lemma 5.4.5 we see that (iii) is satisfied.

The proof is completed by applying Lemma 5.4.6 to get (iii)  $\Rightarrow$  (ii).

THEOREM 5.4.8. Let  $\alpha \in (-2, 0)$  and let  $\phi$  be a characteristic function. The following statements are equivalent.

- (i)  $\phi$  is  $\alpha$ -self-decomposable,  $\phi'$  exists on  $\mathbb{R} \setminus \{0\}$  with  $\lim_{t \to 0} t \phi'(t) = 0$ ;
- (ii) There exists a unique infinitely divisible characteristic function  $\gamma$  and a stable, possibly degenerate, characteristic function  $\phi_{\text{STABLE}(-\alpha)}(t)$  such that

$$\ln \phi(t) = \int_0^1 \ln \gamma(vt) v^{\alpha-1} dv + \ln \phi_{\text{STABLE}(-\alpha)}(t);$$

- (iii)  $\phi$  is infinitely divisible with Lévy spectral function M having left and right derivatives and such that M'(x) is  $\alpha$ -unimodal;
- (iv)  $\phi$  is  $\alpha$ -self-decomposable and  $\phi_c$  is an infinitely divisible characteristic function for every  $c \in (0, 1)$ .

PROOF. Lemma 5.4.6 proves that (iii)  $\Rightarrow$  (ii) and (ii)  $\Rightarrow$  (i) is shown as in Theorem 5.4.7. It remains to prove (i)  $\Rightarrow$  (iv)  $\Rightarrow$  (iii). As in the proof of Theorem 5.4.7, equation (5.8) gives rise to the differential equation (5.24). Integrating on both sides of (5.24) over  $(t_0, t), t_0 > 0$  yields

$$\int_{t_0}^t u^{\alpha-1} \ln \gamma(u) du = t^{\alpha} \ln \phi(t) - t_0^{\alpha} \ln \phi(t_0) ,$$

and hence

$$\ln \phi(t) = \int_{t_0}^t \ln \gamma^{1/\alpha}(u) d(u/t)^{\alpha} + t^{-\alpha} t_0^{\alpha} \ln \phi(t_0) ,$$

for all  $t \ge t_0$ . By (5.8) we have

$$\ln \phi_c(t) = \int_c^1 \ln \gamma(vt) v^{\alpha - 1} dv , \qquad (5.26)$$

for all  $t \ge 0$ . Similarly, we obtain that  $\phi_c$  is of the form (5.26) for  $t \le 0$ . So  $\phi_c$  is infinitely divisible and condition (iv) is satisfied.

By Theorem 1.3.8 (i),  $\phi_c^{c^{-\alpha}}$  is a characteristic function for every c such that  $c^{\alpha} \in \mathbb{N}_+$ . From (5.8) it follows that

$$\phi^{c^{-\alpha}}(t) = \phi(ct) \phi^{c^{-\alpha}}_c(t)$$

So  $\phi^{c^{-\alpha}}(t)$  is a characteristic function for every c such that  $c^{\alpha} \in \mathbb{N}_+$  and therefore, by Theorem 1.3.8 (i),  $\phi$  is infinitely divisible. Since  $\phi$  and  $\phi_c$  are related by (5.8) the Lévy spectral function  $M_c$  of  $\phi_c$  is given by

$$M_c(x) = M(x) - c^{\alpha}M(x/c)$$

(cf. Theorem 1.3.2 and Lukacs (1970), p. 163).  $M_c$  is non-increasing by Theorem 1.3.2 (i), hence from Lemma 5.4.5 we see that (iii) is satisfied.

It is interresting to note (as was done to me by L. de Haan) the exceptional role that the stable characteristic function (with exponent  $\alpha$ ) plays in Theorem 5.4.8 (ii). This resembles the exceptional role that the normal characteristic function plays in Theorem 1.3.2. The stable characteristic function with exponent  $\alpha$  has an 'extreme' Lévy spectral function M in the sense that  $x^{1-\alpha}M'(x)$  is constant and that it is the only Lévy spectral function that satisfies Lemma 5.4.5 (i) with equality. Contrary to the case of Theorem 1.3.2, the tail behaviour of the distribution function of  $\phi_{\text{STABLE}(-\alpha)}$  is not much different than that of the integral in Theorem 5.4.8 (ii) (cf. Steutel (1974) and his reference list).

NOTATION 5.4.9. A characteristic function  $\phi$  belongs to the set  $S_{\alpha}(\mathbb{R})$  if  $\phi$  satisfies any of the conditions (i) - (iii) of Theorems 5.4.7 and 5.4.8.

If  $\phi$  is stable with exponent  $\delta$ , then, by Theorem 1.4.1,  $\phi$  has a Lévy spectral function M such that M'(x) is  $(-\delta)$ -unimodal and hence (cf. Remark 5.2.3)  $\alpha$ -unimodal for all  $\alpha \ge -\delta$ . We thus have the following corollary to Theorem 5.4.8 (iii).

COROLLARY 1.  $\phi$  is stable with exponent  $\delta$ , then  $\phi \in S_{\alpha}(\mathbb{R})$  for every  $\alpha \geq -\delta$ .

The proof of (i)  $\Rightarrow$  (iv)  $\Rightarrow$  (iii) is also valid for  $\alpha \leq -2$ . By Remark 5.4.4 and observing that the normal characteristic function is  $\alpha$ -self-decomposable if and only if  $\alpha \geq -2$ , we obtain

COROLLARY 2. The normal characteristic function is the only characteristic function in  $S_{-2}(\mathbb{R})$  and in the case  $\alpha < -2$ ,  $S_{\alpha}(\mathbb{R})$  contains only degenerate characteristic functions.

By (5.26) and Theorem 5.4.8 (iii) we see that

$$c^{\alpha} \ln \phi(ct) = \ln \phi(t) - \ln \phi_c(t) = \int_0^{\infty} \ln \gamma(vt) v^{\alpha - 1} dv + \ln \phi_{\text{STABLE}(-\alpha)}(t)$$

Let  $c = (n+1)^{\alpha^{-1}}$ . Letting  $n \to \infty$  we have (cf. Definition 1.4.7)

COROLLARY 3. Let  $-2 \le \alpha < 0$ . If  $\phi \in S_{\alpha}(\mathbb{R})$ , then  $\phi$  is in the domain of normal attraction of a stable characteristic function of order  $-\alpha$ .

REMARK 5.4.10. It follows from the proof of Theorem 5.4.8 that the the integral in Theorem 5.4.8 (ii) converges if and only if the Lévy spectral function N of the (unique) infinitely divisible characteristic function  $\gamma$  satisfies

$$\int_{|v|\geq 1} v^{-\alpha} dN(v) < \infty , \alpha \in (-2,0),$$

and  $\gamma$  has no degenerate component if  $\alpha \in (-2, -1]$ , i.e.,  $a_{\gamma}$  is of the form (5.21). For  $\alpha = 0$  it is necessary and sufficient that the Lévy spectral function N satisfies

$$\int_{|v|\geq 1} |v|^{-1} N(v) \, dv = \int_{|x|\geq 1} \ln |x| \, dN(x) < \infty,$$

and hence

$$\int_{|x|\geq 1} \ln |x| \, dN(x) < \infty.$$

Finally, the integral in Theorem 5.4.7 (ii) converges for any characteristic function  $\gamma$ , provided  $\alpha > 0$ .

REMARK 5.4.11. O'Connor (1979b) proves Theorem 5.4.7 for  $\alpha \in (0,1)$ . In his proof he uses the results of O'Connor (1979a), which proves Theorem 5.4.7 for  $\alpha = 1$ . We prove Theorem 5.4.7 for any fixed  $\alpha \ge 0$  and we do not use results for other known fixed  $\alpha$ . O'Connor (1981) does not consider condition (i) of Theorem 5.4.8, but the condition

#### (i') $\phi$ is $\alpha$ -self-decomposable and infinitely divisible,

and proceeds to prove (i')  $\Leftrightarrow$  (iii) for  $\alpha \in (-2,0)$  and (ii)  $\Leftrightarrow$  (iii) for  $\alpha \in (-1,0)$ . We prefer (i) to (i') as it is usually easier to verify that  $\phi'$  exists and  $t \phi'(t) \rightarrow 0$  as  $t \rightarrow 0$  rather than  $\phi$  is infinitely divisible. The proofs are also simpler if condition (i') is used instead of (i). O'Connor tried but was unable to prove (ii)  $\Leftrightarrow$  (iii) for  $\alpha \in (-2,-1]$ . We were able to do so by making the appropriate choices of  $a_{\gamma}$  and  $\sigma_{\gamma}$  in the proof of Lemma 5.4.6.

REMARK 5.4.12. Jurek (1988) and (1989) obtains some of the results of this section. Our work, however, was done independently and almost simultaneously of Jurek's. Jurek (1988) and (1989) defines  $S_{\alpha}(\mathbb{R})$  as the set of characteristic functions  $\phi$  which can be obtained as limits of the following kind (cf. Theorem 6.2.3 (ii));

$$\ln \phi(t) = \lim_{n \to \infty} \sum_{j=1}^{n} (1/n)^{\alpha} \ln \phi_j(t/n) ,$$

where  $(\phi_j)$  is a sequence of infinitely divisible characteristic functions. He then proves that  $\phi \in S_{\alpha}(\mathbb{R}) \iff (i')$  and that  $(i') \iff (ii)$  of Theorems 5.4.7 and 5.4.8 for  $\alpha > -2$ . Jurek however assumes that  $\phi$  stems from a symmetric random variable in the case that  $\alpha \in (-2, -1]$ . The key to understanding why this assumption is not necessary lays in equation (5.21) in the proof of Lemma 5.4.6. By choosing  $a_{\gamma}$  as in (5.21) we ensure that the integral in Theorem 5.4.8 (ii) converges. If  $\phi$  stems from a symmetric random variable, then so does  $\gamma$  and hence N (the Lévy spectral function of  $\gamma$ ) is symmetric. If N is symmetric, then  $a_{\gamma}=0$  in (5.21) and the integral in Theorem 5.4.8 (ii) will always converge. If  $N \equiv 0$ , then by (5.21)  $a_{\gamma}=0$  if the integral in Theorem 5.4.8 (ii) is to converge. Furthermore, if we assume that  $a_{\gamma}=0$  and require that the integral in Theorem 5.4.8 (ii) converges, then necessarily

$$\int_{\mathbb{R}\setminus\{0\}} v^3/(1+v^2) dN(v) = 0,$$

which (for example) symmetry of N ensures.

REMARK 5.4.13. The proof of Theorem 5.3.4 can easily be adapted to prove the analogue of Theorem 5.3.4 for distributions on  $\mathbb{R}$  (and  $\mathbb{R}_+$ ).

## 5.5 Distributions on $\mathbb{R}_{\perp}$

In this section we use the results of Section 5.3, to prove the analogue of Theorems 5.3.2, 5.4.7 and 5.4.8 for random variables supported by  $\mathbb{R}_+$ . The theorem is easily proved by Theorems 5.4.7 and 5.4.8. It is stated separately because it takes a simpler form (as in Theorem 5.3.2), and, as will be shown, the  $\alpha$ -self-decomposable distributions on  $\mathbb{R}_+$  and on  $\mathbb{N}_0$  are closely related.

DEFINITION 5.5.1. Let the random variable X on  $\mathbb{R}_+$  have distribution function f and Laplace-Stieltjes transform  $\hat{f}$ . The function  $\hat{f}$  is said to be  $\alpha$ -self-decomposable and belong to  $S_{\alpha}(\mathbb{R}_+)$ , for some  $\alpha \in \mathbb{R}$ , if for every  $c \in (0,1)$  there exists a Laplace-Stieltjes transform  $\hat{f}_c$  such that

$$\hat{f}(\tau) = \hat{f}^{c^{\alpha}}(c\tau)\hat{f}_{c}(\tau) \quad , \tau \in \mathbb{R}_{+}.$$
(5.27)
5.5 Distributions on  $\mathbb{R}_{\perp}$ 

It can be shown (cf. Theorem 2.7.1 of van Harn (1978)) that X is infinitely divisible and supported by  $\mathbb{R}_+$  (cf. Theorem 1.3.3) if and only if the characteristic function  $\phi$  of X is given by Theorem 1.3.2 with  $M \equiv 0$  for x < 0,  $a_{\phi} = \sigma_{\phi}^2 = 0$  and x M'(x) = H'(x) for x > 0. Furthermore  $\hat{f}$  is analytic and  $\tau \hat{f}'(\tau) \to 0$  as  $\tau \to 0$  (cf. Steutel and van Harn (1979)). We have thus proved

THEOREM 5.5.2. Let  $\alpha \in (-1,\infty)$  and let  $\hat{f}$  be a Laplace-Stieltjes transform. The following statements are equivalent.

- (i)  $\hat{f}$  is  $\alpha$ -self-decomposable;
- (ii)  $\ln \hat{f}(\tau) = \int_{0}^{1} \ln \hat{q}(v\tau) v^{\alpha-1} dv \lambda \tau^{-\alpha}, \lambda \ge 0 \text{ and } \hat{q} \text{ a unique infinitely divisible}$ Laplace-Stieltjes transform;
- (iii)  $\hat{f}$  is infinitely divisible with canonical measure H having left and right derivatives and such that H'(x)/x is  $\alpha$ -unimodal on  $\mathbb{R}_+$ .

Furthermore  $\lambda$  in (ii) is zero if  $\alpha \ge 0$  and  $\hat{f}_c$  is infinitely divisible for every  $c \in (0,1)$  if  $\hat{f}$  is  $\alpha$ -self-decomposable.

Since all log-convex functions and all completely monotone functions are nonincreasing we have (cf. Theorems 4.3.3 and 3.5.3)

COROLLARY 1. If H'(x) / x is log-convex, then  $\hat{f} \in S_0(\mathbb{R}_+)$ .

COROLLARY 2. If F is a mixture of exponential distributions, then  $\hat{f} \in S_1(\mathbb{R}_+)$ .

Goldie (1967) proved that there is a correspondence between probability generating functions in  $ID(\mathbb{IN}_0)$  and Laplace-Stieltjes transforms in  $ID(\mathbb{IR}_+)$ . A similar result was proved in Forst (1979) between  $S_0(\mathbb{IN}_0)$  and  $S_0(\mathbb{IR}_+)$ . We now prove a lemma which generalizes these results by giving a correspondence between probability generating functions in  $S_{\alpha}(\mathbb{IN}_0)$  and Laplace-Stieltjes transforms in  $S_{\alpha}(\mathbb{IR}_+)$ .

LEMMA 5.5.3. Let F be a distribution function on  $\mathbb{R}_+$  and define the probability distribution  $(p_n(s))$  on  $\mathbb{N}_0$  by

$$p_n(s) = (n!)^{-1} \int_0^\infty e^{-st} (st)^n dF(t)$$
.

The following statements are equivalent

- (i) the probability generating function of  $(p_n(s))$  is in  $S_{\alpha}(\mathbb{N}_0)$  for all  $s \in \mathbb{R}_+$ ;
- (ii) the Laplace-Stieltjes transform of F is in  $S_{\alpha}(\mathbb{R}_+)$ .

**PROOF.** The probability generating function  $P_s$  of  $(p_n(s))$  and the Laplace-Stieltjes transform  $\hat{f}$  of F are related by

 $\hat{f}(s(1-z)) = P_s(z).$ 

The lemma now follows from (5.4) and (5.27).

REMARK 5.5.4. Theorem 5.5.2 can also be proved by using Theorem 5.3.2, Lemma 5.5.3, Lemma 4.2.1 of Steutel (1970) and Theorem 1, p. 439 of Feller (1971).

REMARK 5.5.5. Multiple  $\alpha$ -self-decomposable characteristic functions are studied in Hansen (1989b) and (a generalization and a unification of) multiple  $\alpha$ -self-decomposable distributions on  $\mathbb{R}_+$  and  $\mathbb{N}_0$  is considered in Hansen (1989a). In Hansen (1989c) characteristic functions satisfying

$$\phi(t) = \phi^{c^{\alpha}}(c^{-1}t)\phi_{c}(t) \qquad , t \in \mathbb{R},$$

are characterized.

## **5.6** A classification of ID(I)

From Theorems 5.4.7 and 5.4.8 it is clear that  $S_{\alpha}(\mathbb{R}) \subseteq ID(\mathbb{R})$ , and that  $S_{\alpha}(\mathbb{R})$  is closed under multiplication and limits. If  $\alpha(1) > \alpha(2)$  and  $\phi \in S_{\alpha(2)}(\mathbb{R})$ , then (cf. (5.8))  $\phi(t) = \phi^{c^{\alpha(1)}}(ct) \phi^{c^{\alpha(2)}} - c^{\alpha(1)}(ct) \phi_c(t)$ .

Since  $\phi$  is infinitely divisible,  $\phi^{c^{\alpha(2)}-c^{\alpha(1)}}$  is a characteristic function and hence  $\phi \in S_{\alpha(1)}(\mathbb{R})$ . From Corollary 2 to Theorem 5.4.8 it follows that  $S_{\alpha}(\mathbb{R})$ ,  $\alpha < -2$ , contains only degenerate characteristic functions. Theorem 5.4.7 (ii) implies that for any  $\phi \in ID(\mathbb{R})$ ,

$$\phi_{\alpha}(t) := \exp\left(\int_{0}^{1} \ln \phi^{\alpha}(vt) v^{\alpha-1} dv\right) = \exp\left(\int_{0}^{1} \ln \phi(vt) dv^{\alpha}\right),$$

is in  $S_{\alpha}(\mathbb{R})$  for all  $\alpha > 0$ , with the integral converging by Remark 5.4.10. By Helly's second theorem  $\phi_{\alpha} \rightarrow \phi$  as  $\alpha \rightarrow \infty$ . Hence

$$\bigcup_{\alpha \in \mathbb{R}} S_{\alpha}(\mathbb{R}) = ID(\mathbb{R}) .$$

66

Similarly for  $S_{\alpha}(\mathbb{R}_+)$  and  $S_{\alpha}(\mathbb{N}_0)$ . We collect these results in the following theorem.

THEOREM 5.6.1. For  $I = \mathbb{R}$ ,  $\mathbb{R}_+$  or  $\mathbb{N}_0$ , the sets  $S_{\alpha}(I)$  are multiplication semigroups, closed under limits and provide a classification of ID(I), i.e.,

- (i) If  $\alpha_2 < \alpha_1$  then  $S_{\alpha_2}(I) \subset S_{\alpha_1}(I)$ ;
- (ii)  $ID(I) = \bigcup_{\alpha \in \mathbb{R}} S_{\alpha}(I);$
- (iii)  $S_{\alpha}(I)$  is closed under limits and multiplication for every  $\alpha$ .

## Chapter 6

# α-SELF-DECOMPOSABILITY AND LIMIT LAWS

## **6.1 Introduction**

Limit distributions of sums of independent random variables has been a central topic in probability theory and statistics for many years. The classical central limit problem and its successive generalizations to stable and self-decomposable random variables are well-known examples. Also, the set of infinitely divisible random variables can be described as the solution of the so-called general central limit problem (cf. Loève (1977), compare Theorem 1.3.11). Recently, Jurek (1981) introduced the set of s-self-decomposable random variables, which are defined as limits of sums of 'shrunken' random variables. Jurek (1985) showed that this set coincides with the set of random variables having characteristic functions in  $S_1(\mathbb{R})$  (see Notation 5.4.9). The set of 1-self-decomposable random variables satisfying (i) of Theorem 5.4.8 is included in the set of infinitely divisible random variables and in turn includes the sets of self-decomposable and stable random variables.

In contrast to the previous chapters we only study random variables on  $\mathbb{R}$  in this chapter. We consider random variables with characteristic functions in  $S_{\alpha}(\mathbb{R})$  as limits of sums of independent normed or 'shrunken' random variables. Using the representations of  $S_{\alpha}(\mathbb{R})$  obtained in the previous chapter, we present in Section 6.2 four limit forms of  $\alpha$ -self-decomposable random variables. One of these limit forms is the one used by Jurek (1988) and (1989) to define  $S_{\alpha}(\mathbb{R})$ . The method used is easily adapted to  $S_{\alpha}(\mathbb{N}_0)$ . In Section 6.3 we introduce two shrinking operators. Both operators are closely connected with  $\alpha$ -unimodality. The first operator is a generalization of Jurek's (1981) shrinking operator and the second operator, a stochastic shrinking operator, is a generalization of the well-known linear operator  $T_t$ 

## 6.1 Introduction

(cf. Section 1.4). In the same vein as Jurek (1981) we consider limit distributions of sums of 'generalized shrunken' independent random variables and show that the set of these limit distributions is equal to  $S_{\alpha}(\mathbb{R})$ ,  $\alpha \ge 0$ . In Section 6.3 we also characterize the set of distributions obtained as limits of sums of 'generalized shrunken' independent identically distributed random variables. In Section 6.4 we discuss  $S_{\alpha}(\mathbb{R})$  for  $\alpha \in (-2,0]$  in the context of limit distributions in detail. Among other results we show that  $S_{\alpha}(\mathbb{R})$ ,  $\alpha \in (-2,0]$ , contains limit distributions of  $T_c$ -normed sums of block wise identically distributed random variables. We conclude in Section 6.5 with a few comments and remarks.

#### 6.2 Sums of uan triangular arrays

In this section we use equation (5.8) and the results of Section 5.4 to describe random variables X with characteristic functions  $\phi$  in  $S_{\alpha}(\mathbb{R})$  as limits of sums of uan random variables. Our approach is similar to that in Lukacs (1970), Theorem 5.11.1 and its corollary.

Let  $\phi_{\text{STABLE}(-\alpha)}(t) = 1$  for  $\alpha \ge 0$ . Writing the integrals in Theorems 5.4.7 and 5.4.8 as limits of Riemann sums we have (cf. Notation 5.4.9)

THEOREM 6.2.1. Let  $\alpha > 0$  and let  $\phi$  be a characteristic function. The following statements are equivalent.

- (i)  $\phi \in S_{\alpha}(\mathbb{R});$
- (ii) There exists an infinitely divisible characteristic function  $\gamma$  such that

$$\ln \phi(t) = \alpha^{-1} \lim_{n \to \infty} \sum_{j=1}^{n} (j/n) \ln \gamma^{1/j} ((j/n)^{1/\alpha} t);$$

(iii) There exists an infinitely divisible characteristic function  $\gamma$  and a stable characteristic function  $\phi_{\text{STABLE}(-\alpha)}(t)$  such that

$$\ln \phi(t) = \ln \phi_{\text{STABLE}(-\alpha)}(t) + \lim_{n \to \infty} \sum_{j=1}^{n} (j/n)^{\alpha} \ln \gamma^{1/j}((j/n)t).$$

Conditions (i) and (iii) are also equivalent for  $\alpha \leq 0$ .

In Theorem 6.2.1 we see that the logarithm of any characteristic function in  $S_{\alpha}(\mathbb{R})$  can be written as a limit of a weighted sum of logarithms of identical characteristic functions. In the next two theorems we obtain a less restrictive limiting form of  $\phi$  than in Theorem 6.2.1. The proof of the first theorem is simpler and almost identical with the proof of the second and is therefore omitted.

THEOREM 6.2.2. Let  $\alpha > 0$  and let  $\phi$  be a characteristic function. The following statements are equivalent.

- (i)  $\phi \in S_{\alpha}(\mathbb{R});$
- (ii) There exist infinitely divisible characteristic functions  $(\phi_i)_1^{\infty}$  such that

$$\ln \phi(t) = \lim_{n \to \infty} \sum_{j=1}^{n} (1/n) \ln \phi_j((1/n)^{1/\alpha} t);$$

(iii) There exist infinitely divisible characteristic function  $(\phi_i)_1^{\infty}$  such that

$$\ln \phi(t) = \lim_{n \to \infty} \sum_{j=1}^{n} (1/n)^{\alpha} \ln \phi_j(t/n).$$

Conditions (i) and (iii) are also equivalent for  $\alpha = 0$ .

THEOREM 6.2.3. Let  $\alpha \in (-2, 0)$  and let  $\phi$  be a characteristic function. The following statements are equivalent.

- (i)  $\phi \in S_{\alpha}(\mathbb{R});$
- (ii) There exist infinitely divisible characteristic functions  $(\phi_j)_1^{\infty}$  such that

$$\ln \phi(t) = \lim_{n \to \infty} \sum_{j=1}^{n} (1/n)^{\alpha} \ln \phi_j(t/n)$$

PROOF. Suppose  $\phi \in S_{\alpha}(\mathbb{R})$  and let  $\alpha \neq -1$ . Letting  $j^{\alpha-1} \ln \gamma(jt) = \ln \phi_j(t)$ ,  $j \ge 2$  and  $\phi_1 = \gamma \cdot \phi_{\text{STABLE}(-\alpha)}$  in Theorem 6.2.1 (iii), we see that (i) implies (ii). A similar argument holds for  $\alpha = -1$ . We now prove (ii)  $\Rightarrow$  (i). Let  $(k_n)$  be a sequence of real numbers such that  $k_n < n$  and  $(k_n/n) \rightarrow c \in (0, 1)$  as  $n \rightarrow \infty$ . Observe that

$$\sum_{j=1}^{n} (1/n)^{\alpha} \ln \phi_j(t/n) = \sum_{j=1}^{k_n} (1/k_n)^{\alpha} (k_n/n)^{\alpha} \ln \phi_j((t/k_n) (k_n/n)) + \sum_{j=k_n+1}^{n} (1/n)^{\alpha} \ln \phi_j(t/n).$$
(6.1)

Letting  $n \to \infty$  in (6.1), the left hand side tends, by definition, to  $\ln \phi(t)$ , the first term on the right hand side tends, by definition, to  $c^{\alpha} \ln \phi(ct)$  and so the second term tends to a limit, which by Theorems 1.3.8 and 1.3.7, is the logarithm of some infinitely divisible characteristic function,  $\ln \phi_c(t)$  say. By Theorem 5.4.8 (iv)  $\phi \in S_{\alpha}(\mathbb{R})$ .

6.2 Sums of uan triangular arrays

Jurek (1988) uses condition (ii) of Theorem 6.2.3 to define  $S_{\alpha}(\mathbb{R})$ .

REMARK 6.2.4. The implications (i)  $\Rightarrow$  (ii) for  $\alpha > 0$  and (i)  $\Rightarrow$  (iii) for  $\alpha > -2$  of Theorem 6.2.2 can also be proved directly by noting that (cf. (5.8))

$$\sum_{j=1}^{n} (j/n) \ln \phi_{c_j}((j/n)^{1/\alpha} t) = \sum_{j=1}^{n} (j/n) [\ln \phi((j/n)^{1/\alpha} t) - ((j-1)/j) \ln \phi(((j-1)/n)^{1/\alpha} t)]$$
  
= 
$$\sum_{j=1}^{n} (j/n) \ln \phi((j/n)^{1/\alpha} t) - \sum_{j=1}^{n} ((j-1)/n) \ln \phi(((j-1)/n)^{1/\alpha} t)$$
  
= 
$$\ln \phi(t),$$

with  $\alpha > 0$  and  $c_j = ((j-1)/j)^{1/\alpha}$  and observing that

$$\sum_{j=1}^{n} (j/n)^{\alpha} \ln \phi_{c_j}((j/n)t) = \ln \phi(t) , \alpha > 0 ,$$
  

$$(1/n)^{\alpha} \ln \phi(t/n) + \sum_{j=2}^{n} (j/n)^{\alpha} \ln \phi_{c_j}((j/n)t) = \ln \phi(t) , \alpha \in (-2,0] ,$$
  
here  $c_j = (j-1)/j$ .

where  $c_j = (j-1) / j$ .

REMARK 6.2.5. For  $\alpha > 0$  we can replace the existence of a sequence of infinitely divisible characteristic functions  $(\phi_j)$  in Theorem 6.2.2 (ii) by the existence of a uan triangular array  $(\phi_{k,n})$ , where

$$\ln \phi_{k,n}(t) = 1/n \sum_{j=k}^{n} \ln \phi_j((1/n)^{\alpha} t) , k = 1, 2, ..., n, n \in \mathbb{N}_+,$$

for some sequence of characteristic functions  $(\phi_j)$ . The implication (ii)  $\Rightarrow$  (i) is then proved, as in Theorem 6.2.2, by observing that

$$\sum_{j=1}^{n} (1/n) \ln \phi_j((1/n)^{\alpha} t) = \sum_{j=1}^{k_n} (1/k_n) (k_n/n) \ln \phi_j((1/k_n)^{1/\alpha} (k_n/n)^{1/\alpha} t) + \sum_{j=k_n+1}^{n} \ln \phi_{j,n}(t) + k_n \ln \phi_{k_n+1,n}(t), \qquad (6.2)$$

and letting  $n \to \infty$  such that  $(k_n/n) \to c \in (0,1)$ . Observe that the sum of the last two terms in (6.2) tends to the logarithm of an infinitely divisible characteristic function by the uan property of  $(\phi_{k,n})$ , Theorem 1.3.12, and Lemma 1.3.14. Conversely, by Remark 6.2.4 we need only show that  $(\phi_{k,n})$  is uan, where

$$\ln \phi_{k,n}(t) = 1/n \sum_{j=k}^{n} \ln \phi_{c_j}((j/n)^{1/\alpha} t) , k = 1, 2, ..., n, n \in \mathbb{N}_+,$$

and  $c_j = ((j-1)/j)^{1/\alpha}$ . An infinitely divisible characteristic function  $\gamma$  has no real zeros, hence for any T > 0 there exists a C > 0 such that  $|\ln \gamma(t)| \le C$ ,  $t \in [-T, T]$ . From (5.25), letting  $a(j) = (j/n)^{1/\alpha}$ , we obtain

#### a-SELF-DECOMPOSABILITY AND LIMIT LAWS

$$|\ln \phi_{k,n}(t)| \leq \frac{C}{n} \sum_{j=k}^{n} \frac{n}{j} \int_{a(j-1)}^{a(j)} u^{\alpha-1} du \leq \frac{C}{\alpha} \frac{\ln n}{n} \to 0 \text{ as } n \to \infty$$

uniformly on [-T, T]. By Lemma 1.3.13,  $(\phi_{k,n})$  is uan. Similarly we can replace the condition that  $(\phi_j)$  be infinitely divisible in part (iii) by the condition that the triangular array  $(\phi_{k,n})$  be uan (this is done in O'Connor (1979b) for  $\alpha \in (0,1)$ ) where

$$\ln \phi_{k,n}(t) = (1/n)^{\alpha} \sum_{j=k}^{n} \ln \phi_j((j/n)t) , k = 1, 2, ..., n, n \in \mathbb{N}_+ .$$

REMARK 6.2.6. We would like to find a suitable definition of the generalized stable laws in  $S_{\alpha}(\mathbb{R})$  analogous to the definition of the stable laws corresponding to the classical self-decomposable laws, i.e., to find the characteristic functions in  $S_{\alpha}(\mathbb{R})$ which can be written as a limit of weighted sums of identical characteristic functions. All the theorems in this section are of the same form as Theorem 6.2.1, where  $\phi$  is equal to the limit of weighted sums of identical characteristic functions. In this setting, the stable laws in  $S_{\alpha}(\mathbb{R})$  are just the limit laws in  $S_{\alpha}(\mathbb{R})$ . Therefore the approach of this section does not suggest a reasonable definition generalizing classical stability.

#### 6.3 Sums of shrunken random variables

٢

Throughout this section we will study  $S_{\alpha}(\mathbb{R})$  for  $\alpha \ge 0$  only. Define the oneparameter family of non-linear shrinking operators  $(U_{\alpha,t}), \alpha > 0$ , by

$$U_{\alpha,t}x = \begin{cases} 0 & \text{if } |x| \le t \\ (1 - (\frac{t}{|x|})^{\alpha})^{1/\alpha} x & \text{if } |x| > t \end{cases},$$
(6.3)

for  $t \ge 0$ . When  $\alpha = 1$ , we regain the shrinking operator  $U_t$  defined in Jurek (1981). In figure 6.1 on the next page we sketch  $U_{\alpha,t}$  as a function in x and  $\alpha$ . Let  $Y_t$  be a random variable defined by  $\mathbb{P}(Y_t=1)=t^{\alpha}=1-\mathbb{P}(Y_t=0)$ . We then define the one-parameter family of linear stochastic operators  $(T_{\alpha,t}), \alpha \ge 0$ , by

$$T_{\alpha,t} x = Y_t T_t x , t \in [0, 1].$$
(6.4)

If  $\alpha = 0$  then  $T_{\alpha,t}$  reduces to the linear operator  $T_t$  (cf. Section 1.4). In this section we show, using the results of the Appendix, that the set of random variables X given by

$$\sum_{k=1}^{n} U_{\alpha,t_{n}} X_{k} + b_{n} \xrightarrow{w} X \quad \text{as } n \to \infty,$$
(6.5)

$$\sum_{k=1}^{n} T_{\alpha, t_{k}}^{(k)} X_{k} + b_{n} \xrightarrow{w} X \quad \text{as } n \to \infty , \qquad (6.6)$$

for suitable  $(t_n)$ ,  $(b_n)$ ,  $(X_k)$  and  $(T_{\alpha,t_n}^{(k)})$ , is equal to the set of random variables with

characteristic functions in  $S_{\alpha}(\mathbb{R})$ . We also characterize the random variables X obtained as (6.5) and (6.6) where  $X_k$  are identically distributed.



figure 6.1

6.3.1  $U_{\alpha,t}$ -shrunken random variables.

Let the random variable X have distribution function F and denote by  $F_U$  the distribution function of  $U_{\alpha,t}(X)$ . Let  $U_{\alpha,t}$  and  $U_{\alpha,t}^{-1}$  act on Borel sets B by

 $U_{\alpha,t}B = \{ y \mid U_{\alpha,t}x = y, \text{ for } x \in B \} \text{ and } U_{\alpha,t}^{-1}B = \{ x \mid U_{\alpha,t}x \in B \}.$ 

Let I = (a, b), 0 < a < b, then

$$F_U(I) := \mathbb{P}(U_{\alpha,t} X \in I) = \mathbb{P}(X \in U_{\alpha,t}^{-1}I) = F(U_{\alpha,t}^{-1}I).$$

We therefore let the operator  $U_{\alpha,t}$  operate on set functions by  $U_{\alpha,t}F(B) = F(U_{\alpha,t}^{-1}B)$ , for any Borel set B. The following lemma gives a connection between  $U_{\alpha,t}$  and the notion of  $\alpha$ -unimodality.

LEMMA 6.3.1. Let  $\alpha > 0$ ,  $\varepsilon > 0$  and let  $\beta_{\varepsilon}$  be the set of Borel sets on  $\mathbb{R} \setminus (-\varepsilon, \varepsilon)$ . Further let M be a Lévy spectral function. The following statements are equivalent.

- (i) M has left and right derivatives on  $\mathbb{R} \setminus \{0\}$  and M'(x) is  $\alpha$ -unimodal;
- (ii)  $M(B) \ge U_{\alpha,t}M(B)$ ,  $t \in (0, \infty)$ ,  $B \in \beta_{\varepsilon}$ .

**PROOF.** First we prove (ii)  $\Rightarrow$  (i). Let  $\alpha \in \mathbb{R}$ , fix  $\varepsilon > 0$  and let  $B = (a, b), 0 < \varepsilon \le a < b$ . (ii) is equivalent to

$$M((b^{\alpha} + t^{\alpha})^{1/\alpha}) - M((a^{\alpha} + t^{\alpha})^{1/\alpha}) \le M(b) - M(a).$$
(6.7)

Let  $w(x) := M(x^{1/\alpha})$ . From (6.7) with  $x = b^{\alpha} + t^{\alpha}$ ,  $y = a^{\alpha} + t^{\alpha}$ ,  $x' = b^{\alpha}$  and  $y' = a^{\alpha}$  it follows that

$$\frac{w(x) - w(y)}{x - y} \le \frac{w(x') - w(y')}{x' - y'} .$$
(6.8)

Hence w is convex. By Proposition 16, p. 109, Royden (1968), w and hence also M has left and right derivatives on  $(\varepsilon, \infty)$ . Letting  $a \to b$  and hence  $y \to x$  and  $y' \to x'$  in (6.8) we obtain

$$(b^{\alpha}+t^{\alpha})^{-1+1/\alpha}M'((b^{\alpha}+t^{\alpha})^{1/\alpha}) \leq b^{1-\alpha}M(b)$$
.

Let  $c^{\alpha} = b^{\alpha} + t^{\alpha}$ , then  $c \ge b$  and

 $c^{1-\alpha}M'(c)\leq b^{1-\alpha}M'(b).$ 

For B = (a,b),  $0 > \varepsilon \ge b > a$  the proof is similar. So M' is  $\alpha$ -unimodal.

The implication (i) $\Rightarrow$ (ii) is proved for B = (a, b),  $0 < \varepsilon \le a < b$ , by observing that (cf. (5.9))

$$M(B) = \int_{B} M'(x) dx = \int_{B} (\int_{x}^{\infty} v^{-\alpha} dN(v)) d\alpha^{-1} x^{\alpha}$$
$$= \int_{U_{\alpha,t}^{-1}B} (\int_{(x^{\alpha} - t^{\alpha})^{1/\alpha}}^{\infty} v^{-\alpha} dN(v)) d\alpha^{-1} (x^{\alpha} - t^{\alpha})$$
$$\geq \int_{U_{\alpha,t}^{-1}B} M'(x) dx = U_{\alpha,t} M(B) .$$

We are now ready to prove

THEOREM 6.3.2. Let X have characteristic function  $\phi$ . Then  $\phi \in S_{\alpha}(\mathbb{R})$ ,  $\alpha > 0$  if and only if there exists a sequence  $(b_n)$  such that

$$\sum_{k=1}^{n} U_{\alpha,t_{n}} X_{k} + b_{n} \xrightarrow{w} X \quad as \ n \rightarrow \infty,$$
(6.5)

where  $(t_n)$  is a non-negative, non-decreasing sequence and  $(X_k)$  is an independent unbounded sequence of random variables such that the triangular array  $(U_{\alpha,t_n}X_k)$  is uan.

**PROOF.** In this proof we will frequently use the results of the Appendix. First we will show that the operator  $U_{\alpha,t}$  satisfies Assumptions A.2.1, A.2.2, A.2.3 and A.2.5. Fix  $\alpha > 0$  and let  $f_t \coloneqq U_{\alpha,t}$ ,  $t \in [0,\infty)$ . By (6.2) we have

$$f_{s} \circ f_{t}(x) = f_{s}(\operatorname{sgn}(x) (|x|^{\alpha} - t^{\alpha})^{1/\alpha}_{+}) = \operatorname{sgn}(x) (|x|^{\alpha} - t^{\alpha} - s^{\alpha})^{1/\alpha}_{+} = f_{s \oplus t}(x),$$

where  $(x)_+$  is equal to x if x > 0 and zero otherwise and  $s \oplus t$  is defined by

 $s \oplus t = (s^{\alpha} + t^{\alpha})^{1/\alpha}$ .

Obviously  $f_0(x) = x$  and so  $f_0 \circ f_t = f_t$  Hence  $S = (\{f_t\}_{t \in \mathbb{R}_+}, \circ)$  is a composition semigroup with respect to the semigroup  $(\mathbb{R}_+, \bigoplus)$ . If  $t \ge s$  then

$$|f_t(x)| = (|x|^{\alpha} - t^{\alpha})_+^{1/\alpha} \le (|x|^{\alpha} - s^{\alpha})_+^{1/\alpha} = |f_s(x)| \le |x|,$$

and so  $f_t$  in fact does shrink its arguments. From (6.2) it follows that  $f_t(x)$  is continuous in both t and x ( $f_t(x)$  is differentiable in both t and x on  $(t, \infty)$ ), that  $f_t$  is unbounded ( $f_t(x) \to \infty$  as  $x \to \infty$ ) and that  $f_t$  is one-to-one on  $(t, \infty)$ .

Suppose X is obtained by (6.5). Since  $(U_{\alpha,t_n}X_k)$  is uan, X is infinitely divisible and hence  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$ . By Corollary 1 to Theorem A.3.2 in the Appendix, *M* satisfies the inequality of Lemma 6.3.1 (ii). By Lemma 6.3.1 and Theorem 5.4.7,  $\phi \in S_{\alpha}(\mathbb{R})$ .

The converse is proved in three steps. First it is shown that the symmetric normal distribution is a limit as described in the theorem; secondly it is shown that a random variable in  $S_{\alpha}(\mathbb{R})$  without normal component is of the desired type and finally these two results are combined to prove the theorem.

Let X be a symmetric normal random variable, i.e.,  $\phi = [0, \sigma_{\phi}^2, 0]$ . Let  $F^{(1)}$  and  $F^{(2)}$  be Weibull distributions, defined by

$$F^{(1)}(x) = 1 - \exp(-\frac{1}{2} \theta x^{2\alpha}), x \ge 0,$$

$$F^{(2)}(x) = \exp(-\frac{1}{2}\theta |x|^{2\alpha}), x < 0.$$

and let G be given by

$$G(B) = \frac{1}{2} F^{(1)}(B \cap \mathbb{R}_{+}) + \frac{1}{2} F^{(2)}(B \cap (-\infty, 0)).$$

Choose  $\theta$  such that

$$\sigma_{\phi}^2 = \int_0^\infty x^2 d \left(1 - e^{-\theta x^{\alpha}}\right) \,.$$

Define  $(t_n)$  by

$$n = t_n^2 \exp(\frac{1}{2}\theta t_n^{2\alpha}).$$

Hence  $t_n \to \infty$  as  $n \to \infty$ . We will prove that the triangular array  $(F_{k,n})$  with  $F_{k,n} = U_{\alpha,t_n}G$ , k = 1, 2, ..., n is uan and satisfies conditions (i) and (ii) of Theorem 1.3.12 with  $M \equiv 0$  and hence that X is of the form (6.5).

The uan property. Let  $\varepsilon > 0$  be fixed. Observe that

$$F_{kn}(\{|x|\geq\varepsilon\}) = U_{\alpha,t}G(\{|x|\geq\varepsilon\}) = e^{-\frac{1}{2}\theta(\varepsilon^{\alpha}+t_{n}^{\alpha})^{2}}.$$

Hence  $(F_{k,n})$  tends to zero outside every neighbourhood of the origin, uniformly in k, as  $n \to \infty$  and so  $(F_{k,n})$  is uan.

Condition (i) of Theorem 1.3.12. Let  $\varepsilon > 0$  be fixed. Observe that

$$\sum_{k=1}^{n} F_{k,n}(\{ |x| \ge \varepsilon \}) = nU_{\alpha,t_n}G(\{ |x| \ge \varepsilon \}) = t_n^2 e^{-\frac{1}{2}\Theta(\varepsilon^{2\alpha} + 2t_n^{\alpha}\varepsilon^{\alpha})}$$

which tends to zero as  $n \to \infty$ . Hence X is infinitely divisible with  $M \equiv 0$ .

Condition (ii) of Theorem 1.3.12. Let  $\varepsilon > 0$  be fixed. Observe that

$$\lim_{n \to \infty} \sum_{k=1}^{n} \int_{|x| \le \varepsilon} x^2 dF_{k,n}(x) = \int_{0}^{\varepsilon t_n} x^2 d(1 - e^{-\theta x^{\alpha}} e^{-t/2\theta(x/t_n)^{2\alpha}})$$
$$\to \int_{0}^{\infty} x^2 d(1 - e^{-\theta x^{\alpha}}) = \sigma_{\phi}^2 \text{ as } n \to \infty.$$

Hence condition (ii) is also satisfied.

Next, let X be a random variable in  $S_{\alpha}(\mathbb{R})$  without normal component, i.e.,  $\phi = [a_{\phi}, 0, M]$ . By Theorem A.3.2 of the Appendix, X can be obtained as a limit of the form (6.5).

We now have proved that for any  $\phi \in S_{\alpha}(\mathbb{R})$  with  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$  we can find distribution functions  $F_{k,n}^{(1)} = U_{\alpha,t_n} G$  and  $F_{k,n}^{(2)} = U_{\alpha,t_n} F_k^{(2)}$  such that

$$\prod_{k=1}^{n} \phi_{k,n}^{(1)} \rightarrow \phi^{(1)} \coloneqq [0, 2\sigma_{\phi}^{2}, 0],$$
$$\prod_{k=1}^{n} \phi_{k,n}^{(2)} \rightarrow \phi^{(2)} \coloneqq [2a_{\phi}, 0, 2M]$$

Let

$$F_k := \frac{1}{2} \left( G + F_k^{(2)} \right).$$

It can be verified that the sequence  $(X_k)$  is unbounded and that  $(U_{\alpha,t_n}F_k)$  is uan. Let  $\phi_{k,n}$  be the characteristic functions of  $U_{\alpha,t_n}F_k$ . In view of Theorem 1.3.12

$$\prod_{k=1}^{n} \phi_{k,n} \to \phi := [a_{\phi}, \sigma_{\phi}^{2}, M].$$

THEOREM 6.3.3. Let X have characteristic function  $\phi$ . Then  $\phi$  is infinitely divisible, i.e.,  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$  with either  $M \equiv 0$  and  $\sigma_{\phi}^2 \geq 0$ , or  $\sigma_{\phi}^2 = 0$  and

$$M(x) = \begin{cases} -C_1 e^{-p_1 x^{\alpha}} & x > 0 \\ C_2 e^{-p_2 |x|^{\alpha}} & x < 0 \end{cases},$$

with  $\alpha > 0$ ,  $p_1 > 0$ ,  $p_2 > 0$ ,  $C_1 \ge 0$ ,  $C_2 \ge 0$  and  $C_1 + C_2 > 0$  if and only if there exists a sequence  $(b_n)$  such that

$$\sum_{k=1}^{n} U_{\alpha, t_{n}} \mathbf{X}_{k} + b_{n} \xrightarrow{w} \mathbf{X} \quad \text{as } n \to \infty$$

where  $(t_n)$  is a non-negative, non-decreasing sequence and  $(X_k)$  is an independent not bounded sequence of identically distributed random variables such that the triangular array  $(U_{\alpha,t_n}X_k)$  is uan.

**PROOF.** In this proof we will frequently use the results of the Appendix. From the proof of Theorem 6.3.2, we know that  $U_{\alpha,t}$  satisfies Assumptions A.2.1, A.2.2, A.2.3 and A.2.5 of the Appendix.

Let  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$  be as in the theorem. If  $\phi = [a_{\phi}, \sigma_{\phi}^2, 0]$ , then by the proof of Theorem 6.3.2, X is of the form (6.5), with (X<sub>k</sub>) identically distributed. Theorem A.4.1 of the Appendix proves the "only if" part for  $\phi = [0, 0, M]$ .

Suppose X can be obtained as a limit as described in the theorem. By Theorem 1.3.12, X is infinitely divisible with  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$ . By Theorem A.4.1, there exists a semigroup homomorphism q from  $(\mathbb{R}_+, \oplus)$  to  $([1, \infty), \cdot)$ , with  $\oplus$  defined in the proof of Theorem 6.3.2, such that

$$M(B) = q(t) U_{\alpha,t} M(B) .$$
(6.9)

Let  $g(x) := q(x^{1/\alpha})$ . Then since q is a homomorphism

 $g(x)g(y) = q(x^{1/\alpha})q(y^{1/\alpha}) = q((x+y)^{1/\alpha}) = g(x+y).$ 

Since  $g(x) \ge 1$  and g(0) = 1 then  $g(x) = \exp(px)$  and so

 $q(x) = \exp(px^{\alpha})$ ,

for some  $p \ge 0$ . If p = 0, then by repeated use of (6.9) we have

 $M(B) = U_{\alpha_{\perp} t n^{1/\alpha}} M(B) .$ 

Letting  $n \to \infty$  we see that  $M \equiv 0$  and hence  $\phi$  is normal. Now let p > 0 and  $B = (a^{\alpha}, b^{\alpha})$  with  $0 < a \le b$ ,  $t^{\alpha} = s$  and let  $w(x) := M(x^{1/\alpha})$ . By (6.9)

$$\frac{w(a) - w(b)}{a - b} = e^{ps} \frac{w(a + s) - w(b + s)}{(a + s) - (b + s)}$$

We know from Theorems 6.3.2 and 5.4.7 that M' exists. Letting  $b \rightarrow a$  we obtain that

 $e^{pa}w'(a) = e^{p(a+s)}w'(a+s) = q$ ,

with q a non-negative constant. Hence  $w(x) = ce^{-px}$  and therefore

$$M(x) = w(x^{\alpha}) = ce^{-px^{\alpha}}$$

with  $c \in \mathbb{R}_{-}$ . Likewise for  $0 > b \ge a$ . Hence *M* is of the desired form.

It remains to be shown that if M is not identically zero then  $\sigma_{\phi}^2 = 0$ . Let F be the distribution function of  $(X_k)$  and let  $M_n := nU_{\alpha,t_n}F$ . Also define

$$R_n(s) \coloneqq M_n(\{ |x| > s \}) \coloneqq nU_{\alpha,t_n}F(\{ |x| > s \}).$$

Let t be such that  $R_n(s)$  is non-zero in a neighbourhood of t and let (k(n)) be a sequence such that  $(t_{k(n)}^{\alpha} - t_n^{\alpha})^{1/\alpha} \to t$  as  $n \to \infty$  (this is possible by Lemma A.3.1 of the Appendix). Observe that

$$\begin{aligned} R_{k(n)}(s) &= n \left( k \left( n \right) / n \right) U_{\alpha, t_n} F\left( \left\{ |x| > s \right\} \right) \\ &= \left( k \left( n \right) / n \right) M_n\left( \left\{ |x| > \left( s^{\alpha} + t^{\alpha}_{k(n)} - t^{\alpha}_n \right)^{1/\alpha} \right\} \right) = \left( k \left( n \right) / n \right) R_n \left( s^{\alpha} + t^{\alpha}_{k(n)} - t^{\alpha}_n \right)^{1/\alpha} \right). \end{aligned}$$

Since  $M_n \to M$ ,  $R_n(s)$  converges for all s > 0 to a not identically zero limit, R(s) say, and we must have  $\lim_{n\to\infty} k(n)/n = c < \infty$ . Integrating by parts we have

$$\int_{|\leq \varepsilon} x^2 dM_{k(n)}(x) = -\varepsilon^2 R_{k(n)}(\varepsilon) + 2 \int_0^{\varepsilon} s R_n((s_1^{\alpha} + t_{k(n)}^{\alpha} - t_n^{\alpha})^{1/\alpha}) ds .$$

Letting  $n \to \infty$  we have

1x

$$\lim_{n\to\infty}\sum_{k=1}^n\int\limits_{|x|\leq\varepsilon} x^2\,dF_{k,n}(x)\leq 2\,c\int\limits_0^\varepsilon s\,R\,((s^\alpha+t^\alpha)^{1/\alpha})\,ds\leq 2\,c\,R\,(t)\,\varepsilon\,,$$

which tends to zero as  $\varepsilon \rightarrow 0$ . Hence, by Theorem 1.3.12,  $\sigma_{\phi}^2 = 0$ .

### 6.3.2 $T_{\alpha,t}$ -shrunken random variables.

Let the random variable X have distribution function F and denote by  $F_T$  the distribution function of  $T_{\alpha,t}X$ . Let I = (a,b), 0 < a < b. We then define  $F_T$  by

$$F_{\mathrm{T}}(I) \coloneqq \mathbb{P}(\mathrm{T}_{\alpha,t} \mathrm{X} \in I) = t^{\alpha} \mathbb{P}(\mathrm{X} \in T_{t}^{-1}I) = t^{\alpha} F(T_{t}^{-1}I).$$

We therefore let the operator  $T_{\alpha,t}$  operate on set functions by  $T_{\alpha,t}F(B) = t^{\alpha}F(T_t^{-1}B)$ , for any Borel set B and with  $T_t$  defined in Section 1.4. Hence for any Borel set B of  $\mathbb{R}\setminus\{-\varepsilon,\varepsilon\}, \varepsilon > 0$ ,

$$M(B) \ge c^{\alpha} T_c M(B) \text{ if and only if } M(B) \ge T_{\alpha,c} M(B).$$
(6.10)

We are now ready to prove

THEOREM 6.3.4. Let X have characteristic function  $\phi$ . Then  $\phi \in S_{\alpha}(\mathbb{R})$  ( $\alpha \ge 0$ ) if and only if there exists a sequence  $(b_n)$  such that

$$\sum_{k=1}^{n} T_{\alpha,t_n}^{(k)} X_k + b_n \xrightarrow{w} X \quad as \ n \to \infty,$$
(6.6)

where  $(t_n)$  is a non-negative, non-increasing sequence with  $t_1 \le 1$ ,  $(X_k)$  is an independent unbounded sequence of random variables such that  $(T_{\alpha,t_n}^{(k)}X_k)$  is uan and the random operators  $T_{\alpha,t_n}^{(k)}$  are independent and identically distributed.

PROOF. In this proof we will frequently use the results of the Appendix. First we will show that  $T_{\alpha,t}$  satisfies Assumptions A.2.1, A.2.2, A.2.3 and A.2.5 of the Appendix. Fix  $\alpha > 0$ . Let  $U_t := T_{\alpha,1/t}, f_t := T_{1/t}, t \in [1,\infty)$  and  $p(t) := t^{-\alpha}$ . We have

$$f_s \circ f_t(x) = f_s(t^{-1}x) = s^{-1} t^{-1} x = f_{s \oplus t}(x),$$

where  $s \oplus t$  is defined by

$$s \oplus t = t \cdot s$$
.

Obviously  $f_1(x) = x$  and so  $f_1 \circ f_t = f_t$ . Hence  $S = (\{f_t\}_{t \in [1,\infty)}, \circ)$  is a composition semigroup with respect to the semigroup  $([1,\infty), \oplus)$ . If  $t \ge s$  then

 $|f_t(x)| = |t^{-1}x| \le |s^{-1}x| = |f_s(x)| \le |x|,$ 

and so  $f_t$  in fact does shrink its arguments. From the definition of  $T_t$  (cf. Section 1.4) it follows that  $f_t(x)$  is continuous in both t and x ( $f_t(x)$  is differentiable in both t and x on  $(0,\infty)$ ), that  $f_t$  is unbounded ( $f_t(x) \to \infty$  as  $x \to \infty$ ) and that  $f_t$  is one-to-one on  $(0,\infty)$ . The operator  $T_{\alpha,t}$  thus satisfies Assumptions A.2.1, A.2.2, A.2.3 and A.2.5 of the Appendix.

Suppose X is obtained by (6.6). Since the sequence  $(T_{\alpha,t_n}X_k)$  is uan, X is infinitely divisible and so  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$ . By (6.10) and Corollary 1 to Theorem A.3.2 in the Appendix M satisfies the inequality of Lemma 5.4.5 (ii). By Lemma 5.4.5 and Theorem 5.4.7,  $\phi \in S_{\alpha}(\mathbb{R})$ .

The converse is proved in three steps. First it is shown that the symmetric normal distribution is a limit as described in the theorem; secondly it is shown that a random variable without normal component is of the desired type and finally these two results are combined to prove the theorem. The last step is identical with the last step of the proof of Theorem 6.3.2 and therefore omitted.

Let X be a symmetric normal random variable, i.e.,  $\phi = [0, \sigma_{\phi}^2, 0]$ . Let Y be a unbounded random variable on  $\mathbb{R}$  with finite second moment and let it have distribution function G. Let

$$\sigma_{\phi}^2 = \int_{-\infty}^{\infty} x^2 \, dG(x) \, .$$

Let  $(t_n)$  be defined by

 $t_n = n^{-1/(\alpha+2)} \ .$ 

We will prove that the triangular array  $(F_{k,n})$  with  $F_{k,n} = T_{\alpha,t_n}G$ , k = 1, 2, ..., n is uan and satisfies conditions (i) and (ii) of Theorem 1.3.12 with  $M \equiv 0$  and hence that X is of the form (6.6).

The uan property. Let  $\varepsilon > 0$  be fixed. Observe that

$$F_{k,n}(\{ |x| \ge \varepsilon\}) = T_{\alpha,t_n} G(\{ |x| \ge \varepsilon\}) \le t_n^{\alpha} = n^{-\alpha/(\alpha+2)}$$

Hence  $(F_{k,n})$  tends to zero outside every neighbourhood of the origin, uniformly in k, as  $n \to \infty$  and so  $(F_{k,n})$  is uan.

Condition (i) of Theorem 1.3.12. Let  $\varepsilon > 0$  be fixed. Observe that

$$\sum_{k=1}^{n} F_{k,n}(\{|x| \ge \varepsilon\}) = n \operatorname{T}_{\alpha, i_n} G(\{|x| \ge \varepsilon\})$$
$$= n^{2/(\alpha+2)} G(\{|x| \ge n^{1/(\alpha+2)}\varepsilon\}).$$
(6.11)

Since Y has finite second moment,  $\lim_{x\to\infty} x^2 (1-G(\varepsilon x)) = 0$  for  $\varepsilon > 0$ . Hence (6.11) tends to zero as  $n \to \infty$ . So X is infinitely divisible with  $M \equiv 0$ .

Condition (ii) of Theorem 1.3.12. Let  $\varepsilon > 0$  be fixed. Observe that

$$\lim_{n \to \infty} \sum_{k=1}^{n} \int_{|x| \le \varepsilon} x^2 dF_{k,n}(x) = nt_n^{\alpha+2} \int_0^{\varepsilon/t_n} y^2 dG(y)$$
$$\to \int_0^{\infty} y^2 dG(y) = \sigma_{\phi}^2 \text{ as } n \to \infty.$$

Hence condition (ii) is also satisfied.

Next, let X be a random variable without normal component, i.e.,  $\phi = [a_{\phi}, 0, M]$ . By Theorem A.3.2 of the Appendix, X can be obtained as a limit of the form (6.6).

THEOREM 6.3.5. Let X have characteristic function  $\phi$ . Then  $\phi$  is stable if and only if there exists a sequence  $(b_n)$  such that

$$\sum_{k=1}^{n} \operatorname{T}_{\alpha, t_{n}}^{(k)} X_{k} + b_{n} \xrightarrow{w} X \quad \text{as } n \xrightarrow{w} ,$$

where  $(t_n)$  is a non-increasing sequence in (0,1],  $(X_k)$  are independent not bounded identically distributed random variables and the random operators  $T_{\alpha,t_n}^{(k)}$  are independent and distributed as  $T_{\alpha,t_n}$ .

**PROOF.** In this proof we will use the results of the Appendix. From the proof of Theorem 6.3.4, we know that  $T_{\alpha,t}$ ,  $t \in (0,1]$  satisfies Assumptions A.2.1, A.2.2, A.2.3 and A.2.5 of the Appendix.

Let  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$  be stable. If  $\phi = [a_{\phi}, \sigma_{\phi}^2, 0]$ , then by the proof of Theorem 6.3.4, X is of the form (6.6), with  $(X_k)$  identically distributed. Theorem A.4.1 of the Appendix proves the 'only if' part for  $\phi = [0, 0, M]$ .

Suppose X can be obtained as a limit as described in the theorem. Observe that

$$\mathbb{P}(|X_{k,n}| \ge \varepsilon) := \mathbb{P}(|T_{\alpha,t}X_k| \ge \varepsilon) = t_n^{\alpha} \mathbb{P}(|T_{t}X_k| \ge \varepsilon) \le \mathbb{P}(|T_{1/t}X_k| \ge \varepsilon).$$

Since the sequence  $(T_{1/t_n}X_k)$  is uan,  $(T_{\alpha,t_n}X_k)$  is uan and hence X is infinitely divisible with characteristic function  $\phi$  of the form  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$ . From Theorem 1.3.12 we have that

$$\lim_{n\to\infty}\sum_{k=1}^n \mathrm{T}_{\alpha,t_n}F_k = \lim_{n\to\infty}n t_n^{\alpha}T_{t_n}F = M ,$$

outside every neighbourhood of the origin. Also

$$\lim_{\varepsilon \to 0} \lim_{n \to \infty} \sum_{k=1}^n \int_{|x| \le \varepsilon} x^2 dT_{\alpha, t_n} F_k(x) = \lim_{\varepsilon \to 0} \lim_{n \to \infty} \sum_{k=1}^n \int_{|x| \le \varepsilon} x^2 dn t_n^{\alpha} T_{t_n} F(x) = \sigma_{\phi}^2.$$

Obviously, if  $n t_n^{\alpha}$  is bounded then  $M \equiv 0$  and so  $\phi$  is normal. Suppose M is not identically zero, then  $n t_n^{\alpha} \to \infty$  as  $n \to \infty$ . Let  $N(n) = n t_n^{\alpha}$ . Then

$$M = \lim_{n \to \infty} N(n) T_{t_n} F , \qquad (6.12)$$

outside every neighbourhood of the origin and

$$\sigma_{\phi}^{2} = \lim_{\varepsilon \to 0} \lim_{n \to \infty} \sum_{k=1}^{n} \int_{|x| \le \varepsilon} x^{2} dN(n) T_{t_{n}} F(x)$$

By Lemma A.3.1 of the Appendix,  $t_{n+1}/t_n \rightarrow 1$  as  $n \rightarrow \infty$ . Observe that (cf. (6.12))

$$(N(n+1)-N(n))T_{t_n}F = (((n+1)t_{n+1}^{\alpha}/nt_n^{\alpha}) - 1)N(n)T_{t_n}F \to 0,$$
(6.13)

as  $n \rightarrow \infty$  outside every neighbourhood of the origin. Similarly

$$\lim_{\varepsilon \to 0} \lim_{n \to \infty} \sum_{k=1}^{n} \int_{|x| \le \varepsilon} x^2 d(N(n+1) - N(n)) T_{t_n} F(x) = 0.$$

For every  $l \in \mathbb{N}_+$  there exists an  $n \in \mathbb{N}_+$  such that  $N(n) \le l < N(n+1)$ . Let  $(t_l')_{l=1}^{\infty}$  be defined by

$$t_l' = t_n$$
,  $N(n) \le l < N(n+1)$ ,  $l = 1, 2, 3, ...$ 

For any  $l \in \mathbb{N}_+$  let n be such that  $N(n) \le l < N(n+1)$ . Then (cf. (6.12) and (6.13))

$$\lim_{l \to \infty} \sum_{k=1}^{l} T_{t_{i}} F = \lim_{l \to \infty} l T_{t_{i}} F = \lim_{l \to \infty} (N(n) T_{t_{n}} F + (l - N(n)) T_{t_{n}} F) = M,$$

outside every neighbourhood of the origin. Similarly

#### a-SELF-DECOMPOSABILITY AND LIMIT LAWS

$$\sigma_{\phi}^2 = \lim_{\varepsilon \to 0} \lim_{l \to \infty} \sum_{k=1}^l \int_{|x| \le \varepsilon} x^2 \, d \, l \, T_{t_i} F(x) \, .$$

By Theorem 1.3.12 we see that

$$\sum_{k=1}^{l} T_{t_i} X_k \to X \text{ as } l \to \infty,$$

where  $(X_k)$  are independent and identically distributed with distribution function F. Hence X is stable (cf. Section 1.4).

REMARK 6.3.6. We can also use Theorem A.4.1 to prove the "if" part of Theorem 6.3.5. This proof however must be split into two steps. The first (which we give below) shows that M is of the desired form. In the second step it is shown that if M is not identically zero then  $\sigma_{\phi}^2 = 0$ . This second step is quite involved (cf. proof of Theorem 6.3.3) and therefore omitted.

Suppose X can be obtained as a limit as described in the theorem. From the proof of Theorem 6.3.4 we know that X can be normal. By Theorem A.4.1 there exists a semigroup homomorphism q from  $([1,\infty),\oplus)$  to  $([1,\infty),\cdot)$  with  $\oplus$  defined in the proof of Theorem 6.3.4, such that

$$M(B) = q(t) t^{-\alpha} T_{1/t} M(B) .$$
(6.14)

Since  $q(x) \ge 1$  and q(1)=1 we have that  $q(x)=x^{\gamma}$  for some  $\gamma \ge 0$ . If  $\gamma=0$  then by repeated use of (6.11) we see that  $M \equiv 0$ . Let  $\gamma > 0$  and B = (a,b) with  $0 < a \le b$ . By (6.14)

$$\frac{M(a)-M(b)}{a-b} = t^{\gamma} t^{-\alpha} \frac{M(at)-M(bt)}{a-b}$$

We know from Theorems 6.3.4 and 5.4.7 that M' exists. Letting  $b \rightarrow a$  we obtain that

$$a^{\gamma-\alpha+1}M'(a)=(at)^{\gamma-\alpha+1}M'(at)=c_1$$
,

with  $c_1$  a non-negative constant. Hence

 $M(x) = c x^{\alpha - \gamma}$ ,

with  $c \in \mathbb{R}_{-}$ . Since *M* is a Lévy spectral function if and only if  $(\alpha - \gamma) \in (0, 2)$  we must have  $\gamma \in (\alpha, \alpha+2)$ . Likewise for  $0 > b \ge a$ . Hence *M* is the Lévy spectral function of a stable distribution.

REMARK 6.3.7. The unbounded condition on  $X_k$  in the theorems of this section is assumed, since if  $(X_k)$  were bounded and  $t_n \to 0$   $(t_n \to \infty)$  then the limit in (6.6) ((6.5)) would be degenerate.

## 6.4 Subsets of the set of self-decomposable limit laws

In this section we consider the subsets  $S_{\alpha}(\mathbb{R})$ ,  $\alpha \in (-2,0]$  of the selfdecomposable limit laws. We show that random variables X whose characteristic functions are in  $S_{\alpha}(\mathbb{R})$  can be written as (1.4), where  $(X_k)$  are independent and block wise identically distributed random variables. This result is in some sense, the counterpart of Theorem 6.3.2 for  $\alpha \in (-2,0]$ .

Let Y(t) be a stochastic process with independent stationary increments with Y(0)=0 with probability one (cf. Feller (1971)) and let  $(X_k)$  be a sequence of independent random variables, all distributed as Y(1). Then

$$Y(s) \stackrel{d}{=} \sum_{k=1}^{s} X_k , \ s \in \mathbb{R}_+ ,$$

where a non-integer sum of the  $X_k$  is defined by the non-integer power of its characteristic function. Let the stochastic operator  $T^*_{\alpha,t}$  act on infinitely divisible random variables Y(1) by

$$T^*_{\alpha,t}Y(1) = tY(t^{\alpha})$$
.

Hence, for any infinitely divisible random variable X we have that

$$\mathbf{T}_{\alpha,t}^* \mathbf{X} \stackrel{d}{=} \sum_{k=1}^{t^{\alpha}} t \mathbf{X}_k ,$$

where  $(X_k)$  are independent random variables distributed as X. If  $\alpha \in \mathbb{R}_{-}$  and  $t \in (0, 1]$ , then  $t^{\alpha} \ge 1$ . As  $T_{\alpha,t}^*X$  produces  $t^{\alpha}$  copies of X we can interpret  $T_{\alpha,t}^*$  with  $\alpha \in \mathbb{R}_{-}$  and  $t \in (0, 1]$ , as a stochastic 'breeding' operator. Let  $(t_n)$  be a sequence of non-increasing real numbers with  $t_1 = 1$  and  $t_n \to 0$  as  $n \to \infty$ . Theorem 6.2.3 (iii) states that a random variable X has a characteristic function in  $S_{\alpha}(\mathbb{R}), \alpha > -2$ , if and only if there exists a sequence  $(X_i)$  of independent infinitely divisible random variables such that

$$\sum_{j=1}^{n} T^*_{\alpha, t_n} X_j \xrightarrow{w} X .$$
(6.15)

Thus (6.15) is a limit of a sum of a triangular array as in Theorem 1.3.12.

To eliminate the condition that  $(X_k)$  be infinitely divisible we introduce a new stochastic 'breeding' operator  $T_{\alpha,t}$  as follows.

$$T_{\alpha,t}X = \sum_{k=1}^{d} tX_k, \ t \in (0,1], \ \alpha \in (-2,0],$$

where [x] denotes the integer part of x and  $(X_k)$  are independent random variables distributed as X. The following theorem gives us another classification of the  $S_{\alpha}(\mathbb{R})$  limit laws. For a proof we refer to Theorem A.5.1 of the Appendix.

THEOREM 6.4.1. Let  $\alpha \in (-2,0]$  and let X have characteristic function  $\phi$ . Then  $\phi \in S_{\alpha}(\mathbb{R})$  if and only if there exists a sequence  $(b_n)$  such that

$$\sum_{k=1}^{n} \mathrm{T}_{\alpha,t_{n}} \mathrm{X}_{k} + b_{n} \xrightarrow{w} \mathrm{X} \quad \text{as } n \to \infty ,$$

where  $(t_n)$  is a non-negative, non-increasing sequence and  $(X_k)$  are independent not bounded random variables such that the sequence  $(T_{\alpha,t_n}X_k)$  is independent and uan.

The set  $S_{\alpha}(\mathbb{R})$  thus contains characteristic functions  $\phi$  whose random variables X are weak limits of normed sums of block-wise identically distributed random variables. The size of these blocks is  $[t_n^{\alpha}]$ . For  $\alpha = 0$ , the block size is one and the limit in Theorem 6.4.1 reduces to the limit defining classical self-decomposable random variables (cf. Section 1.4).

## 6.5 Remarks and comments

Jurek (1988) and (1989) gives the following interpretation of Theorem 6.2.3: Let  $(Y_k(t))$  be a sequence of stochastic processes with independent stationary increments with  $Y_k(0)=0$  with probability one (cf. Feller (1971)) and let  $X_k$  be distributed as  $Y_k(1)$ . Rewritting Theorem 6.2.3 (ii) in terms of stochastic processes with independent stationary increments and letting  $t_n = 1/n$  we get

$$\frac{Y_1(n^{-\alpha}) + Y_2(n^{-\alpha}) + \dots + Y_n(n^{-\alpha})}{n} \xrightarrow{w} X \text{ as } n \to \infty.$$

Note that  $n^{-1} Y_j(n^{-\alpha}) = T^*_{\alpha,t} Y(1)$ , with  $T^*_{\alpha,t}$  defined in Section 6.4. The other theorems of Section 6.2 can also be reformulated in terms of limit distributions of sums of  $T_c$ -shrunken (or  $T^*_{\alpha,t}$ -shrunken) stochastic processes with independent stationary increments.

The shrinking operators introduced in Section 6.3 do have some practical justification. One could for example imagine a situation where a signal X must be measured.  $U_{\alpha,t}$  can then be interpreted as follows; if the signal is too small then our instruments can not register the signal and if the signal is registered, then we only measure a fraction of the strength of the true signal. Likewise, we can interpret  $T_{\alpha,t}$  as: with a certain probability we do not receive the signal, and if the signal is received then we only measure a fraction of the strength of the true signal.

## APPENDIX

# LIMIT DISTRIBUTIONS OF SUMS OF SHRUNKEN RANDOM VARIABLES

## A.1 Introduction

In this appendix we consider random variables X obtained as limits of sums of random variables as follows;

$$\sum_{k=1}^{n} U_{l_n}^{(k)} X_k \xrightarrow{w} X + b_n \text{ as } n \to \infty, \qquad (A.1)$$

where  $(b_n)$  is a sequence of real numbers,  $(X_k)$  are independent, unbounded random variables, the triangular array  $(U_{l_n}^{(k)}X_k)$  is uan, and  $(U_{l_n}^{(k)})_{k=1}^n$  are mutually independent and independent of  $(X_k)$  and all distributed as  $U_t$ , with  $U_t$  a 'generalized stochastic shrinking' operator defined by

$$U_t \mathbf{X} = \mathbf{Y}_t f_t(\mathbf{X}) , \qquad (A.2)$$

where X and  $Y_t$  are independent and  $\mathbb{P}(Y_t=1)=p(t)=1-\mathbb{P}(Y_t=0)$ . Let  $\delta_0$  be a distribution function with total mass at zero and let B be a Borel set. The distribution function of the random variable  $U_t X$  is given by

$$F_{U_{t}X}(B) \coloneqq p(t) \mathbb{P}(f_{t}(X) \in B) + (1-p(t)) \delta_{0}(B)$$
  
=  $p(t) \mathbb{P}(X \in \{x \mid f_{t}(x) \in B\}) + (1-p(t)) \delta_{0}(B)$ . (A.3)

The notion of a shrinking operator originated from Jurek (1981). Many of the proofs in this appendix are based on the proofs of the corresponding theorem in Jurek (1981), where (A.3) is considered with  $U_t = U_{1,t}$  where  $U_{\alpha,t}$  is defined in (6.3). As special cases of  $U_t$  we have the shrinking operators  $U_{\alpha,t}$  and  $T_{\alpha,t}$  introduced in Section 6.4. The assumptions on the 'shrinking functions'  $f_t$ , the norming sequence  $(t_n)$  and the probabilities p(t) are stated in Section A.2. In Section A.3 we characterize the random variables X in (A.1) in terms of an inequality on the Lévy spectral function of X. The special case where  $(X_k)$  are identically distributed is studied in Section A.4. In Section A.5 we give a classification of the set of random variables of the form (A.1) and in Section A.6 we list a few examples.

## **A.2 Preliminaries**

Let  $f_t$  and  $f_t^{-1}$  act on Borel sets B by

$$f_t B = \{ y \mid f_t(x) = y \text{ for } x \in B \} \text{ and } f_t^{-1} B = \{ x \mid f_t(x) \in B \}.$$
 (A.4)

We begin by listing the assumptions which we make on  $f_t$ ,  $(t_n)$  and p(t) in (A.1) and (A.2).

ASSUMPTION A.2.1. We make the following assumptions on  $U_{t_x}X_k$ ;

- (i)  $(X_k)$  are independent, unbounded random variables;
- (ii)  $(Y_{l_{a}}^{(k)})_{k=1}^{n}$  are mutually independent and independent of  $(X_{k})$ ;
- (iii) The triangular array  $(U_{l_n}^{(k)}X_k)$  is uan;
- (iv) The norming sequence  $(t_n)$  is non-decreasing.

ASSUMPTION A.2.2. We make the following assumptions on  $f_t$ ;

(i)  $S = (\{f_t\}_{t \in [0,\infty)}, \circ)$ , is a commutative composition semigroup with respect to the semigroup  $([0,\infty), \oplus)$ , i.e.,

for all  $t, s \in [0, \infty)$ ;  $f_t \circ f_s = f_{t \oplus s}$ ; for all  $t \in [0, \infty)$ ;  $f_0 \circ f_t = f_t = f_t \circ f_0$ ;

(ii)  ${f_t}_{t \in [0,\infty)}$  are shrinking operators, i.e.,

for all  $t, s \in [0, \infty)$  with t > s,  $|f_t(x)| < |f_s(x)|, x \in \mathbb{R}$ ;

for all 
$$x \in \mathbb{R}$$
,  $\lim f_t(x) = 0$ ;

(iii)  $f_t(x)$  is continuous in both t and x, unbounded in x and for any interval I, not containing zero,  $f_t(f_t^{-1}(I)) = I$ .

## A.2 Preliminaries

ASSUMPTION A.2.3. We make the following assumptions on p(t);

(i) p is a semigroup homomorphism from  $([0,\infty), \oplus)$  to  $((0,1], \cdot)$ , i.e.,

$$p(s \oplus t) = p(s)p(t) \text{ for all } t, s \in [0, \infty).$$

Assumption A.2.1 (iii) ensures that the limit in (A.1) exists, namely an infinitely divisible random variable. Part (i) is assumed, since if  $(X_k)$  was bounded and  $t_n \to \infty$  with  $f_{\infty} \equiv 0$ , then (A.1) would have a degenerate limit. Requirement (iv) is equivalent with the assumption that  $(t_n)$  be monotone  $((t_n)$  non-increasing implies that  $(s_n)$  with  $s_n = t_n^{-1}$  is non-decreasing) which is a normal assumption in central limit problems.

Assumption A.2.2 (ii) states that  $f_t$  in fact does shrink its arguments and provides an ordering of  $\{f_t\}$ . Conditions (i) and (iii) of Assumption A.2.2 as well as Assumption A.2.3 are essential for solving the limit problem on hand.

As a consequence of our assumptions we have the following lemma.

LEMMA A.2.4. Let  $\{f_t\}_{t \in [0,\infty)}$  and p(t) satisfy Assumptions A.2.2 and A.2.3, respectively. Then

(i) For all  $s, t \in [0, \infty)$  with  $t \ge s$  and any interval I not containing zero:

 $f_t \circ f_s^{-1}(I) = f_{t \ominus s}(I)$  with  $t \ominus s \in [0, \infty)$ ;

(ii) p(t) is continuous, non-increasing and non-zero on  $[0,\infty)$  with p(0)=1 and for all  $t, s \in [0,\infty)$  with  $t \ge s$ 

 $p(t \ominus s) = p(t) / p(s).$ 

**PROOF.** We first prove part (i). Let  $s, t \in [0, \infty)$  be arbitrary with  $t \ge s$ . We can always select a  $y \in [0, \infty)$  such that

 $|f_{s\oplus y}(x)| \ge |f_t(x)|,$ 

for all  $x \in \mathbb{R}$  and  $s \oplus y \le t$ . By Assumption A.2.2 (ii) it is impossible to have  $w \le t$  and  $|f_w(x_1)| > |f_t(x_1)|$  and  $|f_w(x_2)| < |f_t(x_2)|$  for some  $x_1, x_2 \in \mathbb{R}$ . Hence, by the continuity of  $f_t$  in t, there exists a unique  $y \in [0, \infty)$  such that  $f_{y \oplus s} = f_t$ . We therefore have

 $f_t \circ f_s^{-1}(x) = f_v(x) := f_{t \ominus s}(x),$ 

for any  $x \neq 0$ .

For part (ii) let  $y = t \ominus s$  and hence  $t = y \oplus s$ . Observe that (cf. Assumption A.2.3)

#### 88 LIMIT DISTRIBUTIONS OF SUMS OF SHRUNKEN RANDOM VARIABLES

$$p(t \ominus s)p(s) = p(y \oplus s) = p(t).$$

If  $p(t_n) \to c > 0$  as  $n \to \infty$ , then upon replacing  $X_k$  in (A.1) by  $Y_k = ZX_k$  with  $\mathbb{P}(Z=0) = 1 - \mathbb{P}(Z=1) = 1 - c$  and p(t) by  $p^*(t) \equiv 1$ , we obtain the same limit problem. We therefore make the following and last assumption.

Assumption A.2.5. p(t) is either constantly one or p(t) is non-increasing with  $\lim_{t\to\infty} p(t)=0$ .

We conclude this preliminary section with a lemma and a notation.

LEMMA A.2.6. If 
$$F_n \xrightarrow{w} F$$
 and  $t_n \to t$  as  $n \to \infty$ , then  $U_{t_n} F_n \xrightarrow{w} U_t F$  as  $n \to \infty$ .

**PROOF.** Let  $x_n \to x$ , then by the continuity of  $f_t(x)$  in both t and x,  $U_{t_n}x_n \to U_tx$ . The lemma now follows from Theorem 5.5 in Billingsley (1968).

NOTATION A.2.7. Let U(S, p) denote the set of characteristic functions whose random variables can be described as limits of the form (A.1), under Assumptions A.2.1, A.2.2, A.2.3 and A.2.5.

## A.3 A characterization of U(S,p)

Let the operators  $U_t$  and  $U_t^{-1}$  act on set functions by

$$U_t F(B) = p(t) F(f_t^{-1}B)$$
 and  $U_t^{-1}F(B) = p(t)^{-1} F(f_tB)$ .

Let  $t, s \in [0, \infty)$  with  $t \ge s$ . From Assumptions A.2.2 (i) and Lemma A.2.4 we have

$$U_{t}U_{s}^{-1}F(B) = p(t)/p(s)F(f_{s} \circ f_{t}^{-1}B) = p(t \ominus s)F((f_{t} \circ f_{s}^{-1})^{-1}B)$$
$$= p(t \ominus s)F(f_{t}^{-1} \ominus sB) = U_{t \ominus s}F(B),$$
(A.5)

and that (cf. (A.5))

$$U_{t}F(B) = p(t)F(f_{s} \circ f_{s}^{-1} \circ f_{t}^{-1}B) = p(t)p(s)/p(s)F(f_{s} \circ f_{t}^{-1} \circ f_{s}^{-1}B)$$
  
=  $p(s)U_{t}U_{s}^{-1}F(f_{s}^{-1}B) = p(s)U_{t \ominus s}F(f_{s}^{-1}B)$   
=  $U_{s}U_{t \ominus s}F(B) = U_{t \ominus s}U_{s}F(B).$  (A.6)

Before proving the main theorem of this section, we prove a preparatory lemma.

A.3 A characterization of U(S,p)

LEMMA A.3.1. Let  $\phi$  be infinitely divisible with  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$ . If  $\phi \in U(S, p)$  and if *M* is not identically zero, then

- (i)  $t_n \to \infty as \ n \to \infty$ ;
- (ii)  $\hat{t}_n \coloneqq t_{n+1} \ominus t_n \to 0 \text{ as } n \to \infty$ .

**PROOF.** We first prove part (i). Suppose (i) is not true, then there exists  $t_0 \in [0,\infty)$  such that  $t_n \le t_0 < \infty$  for all  $n \in \mathbb{N}_+$ . By Assumptions A.2.2 (ii) and A.2.5 we have for each  $\varepsilon > 0$ ,

$$\mathbb{P}(|U_{t_n}^{(k)}\mathbf{X}_k| \ge \varepsilon) = p(t_n)\mathbb{P}(|f_{t_n}(\mathbf{X}_k)| \ge \varepsilon) \ge p(t_0)\mathbb{P}(|f_{t_0}(\mathbf{X}_k)| \ge \varepsilon).$$

Since  $(U_{t_n}^{(k)} X_k)$  is uan, then

 $\sup_{k} \mathbb{P}(|f_{t_0}(\mathbf{X}_k)| \ge \varepsilon) = 0.$ 

From Assumption A.2.2 (iii),  $f_t$  is unbounded and hence  $(X_k)$  is bounded. This contradicts Assumption A.2.1 (i). Hence (i) holds.

To prove part (ii) suppose  $(\hat{t}_n)$  has a limit point  $t_0 \in (0,\infty)$ . Then there exists a subsequence (k(n)) such that  $\hat{t}_{k(n)} \rightarrow t_0$ . By Theorem 1.3.12 it follows that outside every neighbourhood of the origin,

$$M_n := \sum_{k=1}^n U_{l_n} F_k \xrightarrow{w} M \text{ as } n \to \infty .$$
(A.7)

Let I be a continuity set of M, with I and interval bounded away from the origin. From (A.7) and (A.6) it follows that

$$M_{k(n)+1}(I) = U_{t_{k(n)+1}}F_{k(n)+1}(I) + \sum_{k=1}^{\kappa(n)} U_{t_{k(n)+1}\ominus t_{k(n)}}U_{t_{k(n)}}F_{k}(I)$$
  
=  $U_{t_{k(n)+1}}F_{k(n)+1}(I) + U_{\hat{t}_{k(n)}}M_{k(n)}(I)$ . (A.8)

Letting  $n \to \infty$  in (A.8) we obtain, by Lemma A.2.6, (A.7) and the uan property of  $(U_L^{(k)} X_k)$  that

$$M(I) = U_{I_0}M(I) . \tag{A.9}$$

By repeated use of (A.9) we see that

$$M(I) = U_{f_{e}} M(I) \leq M(f_{f_{e}} I)$$

for  $k \in \mathbb{N}_+$ . Applying Assumption A.2.2 (ii) yields M(I)=0, which contradicts the assumption of the lemma.

THEOREM A.3.2. Let  $\phi$  be infinitely divisible with  $\phi = [a_{\phi}, 0, M]$ . Then  $\phi \in U(S, p)$  if and only if for every  $t \in (0,\infty)$  and for every Borel set B of  $\mathbb{R} \setminus (-\varepsilon, \varepsilon)$ , for any  $\varepsilon > 0$ , the

#### 90 LIMIT DISTRIBUTIONS OF SUMS OF SHRUNKEN RANDOM VARIABLES

#### Lévy spectral function M satisfies

 $M(B) \ge U_t M(B) \, .$ 

**PROOF.** We first prove the 'only if' part. If M vanishes identically then the 'only if' part is trivial. Suppose M is not identically zero. From Lemma A.3.1 it follows that for any  $t \in (0,\infty)$  there exists a subsequence (k(n)) such that  $k(n) \ge n$  and

$$\tilde{t}_n \coloneqq t_{k(n)} \ominus t_n \to t \text{ as } n \to \infty . \tag{A.10}$$

For any continuity set I of M, with I an interval bounded away from the origin, we have (cf. (A.7) and (A.6))

$$M_{k(n)}(I) = \sum_{k=1}^{k(n)} U_{t_{k(n)}} F_k(I) \ge \sum_{k=1}^n U_{t_{k(n)} \ominus t_n} U_{t_n} F_k(I)$$
$$= \sum_{k=1}^n U_{\tilde{t}_n} U_{t_n} F_k(I) = U_{\tilde{t}_n} M_n(I) .$$

Letting  $n \to \infty$  we have by (A.7), Lemma A.2.6 and (A.10) that *M* satisfies the inequality of the theorem for B = I. Since the Borel sets on  $\mathbb{R} \setminus (-\varepsilon, \varepsilon)$  are generated by the intervals bounded away from zero, the 'only if' part is proved.

Conversely, suppose M satisfies the inequality of the theorem. If M vanishes identically then the 'if' part is trivial. Suppose M does not vanish identically. Suppose  $(r_k)$  is a non-decreasing sequence satisfying conditions (i) and (ii) of Lemma A.3.1 and such that  $p(r_k)^{-1} \in \mathbb{N}_+$  (this is possible by Assumption A.2.5). Define  $L_s$  by

$$L_{s} = M - U_{s}M$$
Let  $s_{k} = r_{k} \ominus r_{k-1}$ ,  $k = 2, 3, ...$  and  $s_{1} = \infty$ . Then  $L_{s_{1}} = M$  and (cf. (A.7) and (A.5))

$$\sum_{k=1}^{\infty} U_{r_k} U_{r_k}^{-1} L_{s_k} = M .$$
 (A.11)

We now approximate  $(f_{r_k}^{-1}L_{s_k})$  by distribution functions  $(F_k)$  and proceed to show that  $(U_{r_k}F_k)$  satisfies the conditions of Theorem 1.3.12.

#### Define $(\varepsilon_k)$ by

 $\varepsilon_k = \inf \{ \varepsilon > 0 \mid L_{s_k} (\{ |x| > \varepsilon \}) \le 1 \} , k \in \mathbb{N}_+,$ 

and  $(\eta_k)$  by  $\eta_k = L_{s_k}(|x| > \varepsilon_k), k \in \mathbb{N}_+$ . Also let

$$F_{k}^{*}(B) = f_{r_{k}}^{-1} L_{s_{k}}(B \cap \{ |x| > \varepsilon_{k} \}) + (1 - \eta_{k}) \delta_{0}(B) , k \in \mathbb{N}_{+},$$

for any Borel set B of  $\mathbb{R}$ , with  $\delta_0$  a distribution function with total mass at zero. Finally, let k(0) = 0,  $k(n) = \sum_{k=1}^{n} p(r_k)^{-1}$  and

$$t_l = r_n \text{ for } k(n-1) < l \le k(n), n \in \mathbb{N}_+;$$
  
$$F_l = F_n^* \text{ for } k(n-1) < l \le k(n), n \in \mathbb{N}_+.$$

Observe that  $\varepsilon_k \to 0$  as  $n \to \infty$ , since, if  $\varepsilon_k \ge \varepsilon_0 > 0$ , then  $L_{s_k}(\{|x| > \varepsilon_0\}) \ge 1$  and so (cf. Lemma A.3.1 (ii))

$$1 \leq \lim_{k \to \infty} L_{s_k}(\{ |x| > \varepsilon_0 \}) = M(\{ |x| > \varepsilon_0 \}) - \lim_{k \to \infty} U_{s_k}M(\{ |x| > \varepsilon_0 \}) = 0.$$

For every  $n \in \mathbb{N}_+$  we can choose an  $N \in \mathbb{N}_+$  such that  $k(N-1) < n \le k(N)$ . Thus

$$M_{n} := \sum_{l=1}^{n} U_{t_{n}} F_{l} = \sum_{l=1}^{k(N)} U_{t_{n}} F_{l} - \sum_{l=n+1}^{k(N)} U_{t_{n}} F_{l}$$
  
$$= \sum_{k=1}^{N} p(r_{k})^{-1} U_{r_{N}} F_{k}^{*} - (k(N) - n) U_{r_{N}} F_{N}^{*}.$$
(A.12)

The uan property. For every  $\varepsilon > 0$ , define the sequence (j(n)) by

$$\sup_{1\leq k\leq n} U_{t_n}F_k(\{|x|>\epsilon\}) = U_{t_n}F_{j(n)}(\{|x|>\epsilon\}).$$

If (j(n)) is bounded, then by Lemmas A.2.6 and A.3.1

$$\lim_{n \to \infty} \sup_{1 \le k \le n} U_{t_n} F_k(\{|x| > \varepsilon\}) = 0.$$
(A.13)

Suppose (j(n)) is unbounded. Observe that  $F_{j(n)} = F_{i(n)}^*$  for some  $i(n) \le N$ . Hence by the definition of  $F_k$  and Lemma A.2.4 (i),

$$U_{t_n}F_{j(n)}(\{ |x| > \varepsilon \}) \le U_{r_N}f_{r_{i(n)}}^{-1}L_{i(n)}(\{ |x| > \varepsilon \}) = p(r_N)f_{r_N \ominus r_{k(n)}}L_{i(n)}(\{ |x| > \varepsilon \})$$

From Assumption A.2.2 (ii) and the unboundedness of  $f_t$  we have that

 $f_{r_N \ominus r_{k(\alpha)}}^{-1}(\{ |x| > \varepsilon\}) \subseteq (\{ |x| > \varepsilon\}).$ 

Hence

$$U_{t_n}F_{j(n)}(\{|x|>\varepsilon\}) \leq p(r_N)L_{i(n)}(\{|x|>\varepsilon\}).$$

Since  $i(n) \to \infty$  as  $n \to \infty$  and so  $s_{i(n)} \to \infty$  as  $n \to \infty$ ,  $L_{s_{i(n)}} \to 0$ . Formula (A.13) is thereby satisfied. The triangular array  $(U_t, F_k)$  is therefore uan.

Condition (i) of Theorem 1.3.12. Let I be a continuity set of M with I an interval bounded away from the origin. Since  $\varepsilon_k \to 0$  as  $k \to \infty$  there exists a  $k_0$  such that  $\varepsilon_k \notin I$  for all  $k > k_0$ . Observe that (cf. (A.12), the definition of  $F_k^*$ , (A.5) and (A.6))

$$M_{k(N)}(I) = \sum_{k=1}^{k_0} p(r_k)^{-1} U_{r_N} F_k^*(I) + \sum_{k=k_0}^{N} p(r_k)^{-1} U_{r_N} f_{r_k}^{-1} (M - L_{s_k} M)(I)$$
  
$$= \sum_{k=1}^{k_0} p(r_k)^{-1} U_{r_N} F_k^*(I) + \sum_{k=k_0}^{N} U_{r_N} U_{r_k}^{-1} M(I) - \sum_{k=k_0}^{N} U_{r_N \ominus r_k} U_{r_k} U_{r_{k-1}}^{-1} M(I)$$
  
$$= \sum_{k=1}^{k_0} p(r_k)^{-1} U_{r_N} F_k^*(I) + M(I) - U_{t_N \ominus t_{k-1}} M(I) .$$

Letting  $N \rightarrow \infty$  we obtain

## 92 LIMIT DISTRIBUTIONS OF SUMS OF SHRUNKEN RANDOM VARIABLES

$$M_{k(N)} \xrightarrow{w} M \text{ as } N \to \infty$$
 (A.14)

outside every neighbourhood of the origin. Observe that

$$(k(N) - n)U_{r_N}F_N^*(I) \le p(r_n)^{-1}U_{r_N}F_N^*(I) \le L_{s_N}(I) .$$
(A.15)

Since  $L_{s_N}(I) \to 0$  as  $N \to \infty$  we have by (A.12), (A.14) and (A.15), that  $M_n \to M$  outside every neighbourhood of the origin. Hence condition (ii) of Theorem 1.3.12 is satisfied.

Condition (ii) of Theorem 1.3.12. By the definition of  $F_k$  we have (cf. (A.12) and (A.11))

$$\sum_{l=1}^{n} \int_{|x| \leq \varepsilon} x^2 \, dU_{t_n} F_l(x) \leq \sum_{k=1}^{N} \int_{|x| \leq \varepsilon} x^2 \, dp \, (r_k)^{-1} U_{r_N} F_k^*(x)$$
$$\leq \sum_{k=1}^{N} \int_{|x| \leq \varepsilon} x^2 \, dU_{r_N} U_{r_k}^{-1} L_{s_k}(x)$$
$$= \int_{|x| \leq \varepsilon} x^2 \, dM(x) \, .$$

Letting  $\varepsilon \rightarrow \infty$ , we see that condition (ii) of Theorem 1.3.12 is met.

In the 'only if' part of the proof of Theorem A.3.2 it was not necessary to assume that  $\sigma_{\phi}^2 = 0$ . We therefore have

COROLLARY 1. If  $\phi = [a_{\phi}, \sigma_{\phi}^2, M]$  and  $\phi \in U(S, p)$ , then M satisfies the inequality of Theorem A.3.2.

## A.4 Stability in U(S,p)

In this section we consider (A.1) with the added assumption that the sequence  $(X_k)$  is identically distributed. We call such limit distributions U(S, p)-stable.

THEOREM A.4.1. Let  $\phi$  be infinitely divisible with  $\phi = [a_{\phi}, 0, M]$ . Then  $\phi$  is U(S, p)-stable if and only if there exists a semigroup homomorphism g from  $([0,\infty), \oplus)$  to  $([1,\infty), \cdot)$ , such that for every  $t \in (0,\infty)$  and for every Borel set B of  $\mathbb{R} \setminus (-\varepsilon, \varepsilon)$ , for any  $\varepsilon > 0$ , the Lévy spectral function M satisfies

$$M(B) = g(t) U_t M(B). \tag{A.16}$$

**PROOF.** We first prove the 'only if' part. If M vanishes identically then the 'only if' part is trivial. Suppose M is not identically zero. From Lemma A.3.1 it follows that for any  $t \in (0,\infty)$  there exists a subsequence (k(n)) such that  $k(n) \ge n$  and (A.10) is satisfied. For any continuity set I of M, with I an interval bounded away from the origin, we have (cf. (A.6))

$$M_{k(n)}(I) := k(n)U_{I_{k(n)}}F(I) = k(n)/n \cdot U_{\tilde{I}_n}M_n(I)$$

Letting  $n \to \infty$  we have by (A.7), Lemma A.2.6 and (A.10) that both  $M_{k(n)}$  and  $U_{\tilde{l}_n}M_n$  converge and hence k(n)/n must converge. Let  $g(t) = \lim_{n\to\infty} k(n)/n$ . Then M satisfies (A.16) with B = I. Since the Borel sets on  $\mathbb{R} \setminus (-\varepsilon, \varepsilon)$  are generated by the intervals bounded away from zero, (A.16) holds for all Borel sets B.

Suppose there is a  $t_0 > 0$  such that  $g(t_0) \le 1$ . By repeated use of (A.16) we see that

 $M(B) \leq U_t \oplus M(B) = p(t \oplus k) f_t \oplus M(B),$ 

for  $k \in \mathbb{N}_+$ . Applying Assumption A.2.2 (ii) we have  $M \equiv 0$ . Hence g(t) > 1 on  $(0, \infty)$ . Observe that  $M(B) = g(t \oplus s) U_{t \oplus s}(B)$  and

 $M(B) = g(t)M(B) = g(t)g(s)U_sU_tM(B) = g(s)g(t)U_{t \oplus s}M(B)$ .

Hence g is a semigroup homomorphism from  $([0,\infty),\oplus)$  to  $([1,\infty),\cdot)$ .

Conversely, suppose M satisfies (A.16). If M vanishes identically then the 'if' part is trivial. Suppose M does not vanish identically. Let  $(t_n)$  be a non-decreasing sequence satisfying conditions (i) and (ii) of Lemma A.3.1 and such that  $g(t_n) = n$  (this is possible since g is unbounded and continuous). Define  $\overline{\varepsilon}$  by

 $\overline{\varepsilon} = \inf \{ \varepsilon > 0 \mid M(\{ |x| > \varepsilon\}) \le 1 \},\$ 

and set  $\eta = M(\{|x| > \overline{\varepsilon}\})$ . Define the probability distribution F by

 $F(B) = M(B \cap \{ |x| > \overline{\varepsilon} \}) + (1-\eta)\delta_0(B),$ 

for any Borel set B of  $\mathbb{R}$ . We now show that  $(U_{t_n}F)$  satisfies the conditions of Theorem 1.3.12.

Condition (i) of Theorem 1.3.12. Let I be an interval bounded away from the origin. For any a there exists an N such that for all  $n \ge N$ 

 $f_{t_n}^{-1}(I) \cap \{ |x| > a \} = f_{t_n}^{-1}(I).$ 

If this were not true then  $f_{t_n}^{-1}(I) \supset B$  for all *n* large and for some non-empty Borel set *B* contained in  $\{ |x| \le a \}$ . Hence  $I = f_{t_n} f_{t_n}^{-1}(I) \supset f_{t_n}(B)$ . By Assumption A.2.2 (ii) for any  $x \in \mathbb{R}, f_{t_n}(x) \to 0$  as  $n \to \infty$ , contradicting the assumption that *I* is bounded away from the origin. Hence for *n* large

$$nU_{t_n}F(I) = g(t_n)p(t_n)(M(\{|x| > \overline{\varepsilon}\} \cap f_{t_n}^{-1}(I)) + (1-\eta)\delta_0(f_{t_n}^{-1}(I)))$$
  
=  $g(t_n)U_{t_n}M(I) = M(I)$ . (A.17)

Condition (i) of Theorem 1.3.12 is thus satisfied.

The uan property. For any  $n \ge N$  and any  $\varepsilon > 0$  we have by (A.17)

 $\sup_{1 \le k \le n} U_{t_n} F\left(\{ |x| \ge \varepsilon\}\right) = n^{-1} M\left(\{ |x| \ge \varepsilon\}\right).$ 

Letting  $n \to \infty$  we see that  $(U_{t_n}F)$  is uan.

Condition (ii) of Theorem 1.3.12. Finally, observe that for  $n \ge N$ , (cf. (A.17))

$$\sum_{k=1}^{n} \int_{|x| \leq \varepsilon} x^2 dU_{t_n} F(x) = \int_{|x| \leq \varepsilon} x^2 dM(x) .$$

Condition (ii) of Theorem 1.3.12 is now met upon letting  $n \to \infty$  and  $\varepsilon \to 0$ .

### A.5 Some subsets of U(S,p)

Let the stochastic 'breeding operator'  $B_t$  be defined by

$$B_{t}X = \sum_{k=1}^{d} U_{t}^{(k)}X_{k},$$

where  $X_1, ..., X_{[q(t)]}$  are independent random variables, all distributed as X and q(t) is a semigroup homomorphism from  $([0,\infty), \oplus)$  to  $([1,\infty), \cdot)$ . We are interested in the random variables X which have the form

$$\sum_{k=1}^{n} B_{l_n}^{(k)} X_k + b_n \xrightarrow{w} X \text{ as } n \to \infty,$$
(A.18)

where  $(b_n)$  is a sequence of real numbers,  $(B_{l_n}^{(k)}X_k)$  are uan,  $(X_k)$  are independent and unbounded,  $B_{l_n}^{(k)}$  are independent and identically distributed, and all independent of  $(X_k)$  and  $(t_n)$  is non-decreasing. We therefore introduce the notation

NOTATION A.5.1. Let B(S, p, q) denote the set of characteristic functions whose random variables can be described as limits of the form (A.18).

Let  $F_k$  be the distribution function of  $X_k$ . Then

$$M_n^* := \sum_{k=1}^n \sum_{l=1}^{[q(t)]} U_{t_n} F_k = [q(t)] \sum_{k=1}^n U_{t_n} F_k.$$
(A.19)

As in the proof of Lemma A.3.1 (i) we see that the uan property of  $(B_{t_n}X_k)$  implies that  $t_n \to \infty$  as  $n \to \infty$ . If  $q(t) \to c < \infty$  as  $t \to \infty$ , then  $c = \lim_{n \to \infty} q(t^{\oplus n}) = cq(t)$  and hence q(t) = 1 for all t. Hence assume without loss of generality that

 $q(t) \rightarrow \infty$  as  $t \rightarrow \infty$ .

It follows from (A.19) and Theorem 1.3.12, that  $M_n^*$  tends to a Lévy spectral function

### A.5 Some subsets of U(S,p)

M and so

$$\sum_{k=1}^{n} U_{t_n} F_k \to 0 \text{ as } n \to \infty,$$

outside every neighbourhood of the origin. Therefore  $M_n^*$  and  $M_n$  with

$$M_n := q(t_n) \sum_{k=1}^n U_{t_n} F_k,$$

have the same limits. The proof of Theorem A.3.2 can now easily be adapted to prove

THEOREM A.5.2. Let  $\phi$  be infinitely divisible with  $\phi = [a_{\phi}, 0, M]$ . Then  $\phi \in B(S, p, q)$  if and only if there exists a semigroup homomorphism q from  $([0,\infty), \oplus)$  to  $([1,\infty), \cdot)$ , such that for every  $t \in (0,\infty)$  and for every Borel set B of  $\mathbb{R} \setminus (-\epsilon, \epsilon)$ , for any  $\epsilon > 0$ , the Lévy spectral function M satisfies

 $M(B) \ge q(t) U_t M(B).$ 

## A.6 Examples

We conclude this appendix with a few examples. A special case of Example A.6.1 yields the shrinking operator  $T_{\alpha,t}$  of Section 6.4.1 and of Example A.6.2, the operator  $U_{\alpha,t}$  considered in Section 6.4.2.

EXAMPLE A.6.1.

 $f_t(x) = t^{\gamma} x, \quad t \in (0,1], \quad \gamma > 0,$   $p(t) = t^{\alpha}, \quad t \in (0,1], \quad \alpha \ge 0,$  $s \oplus t = s t.$ 

Hence  $U_t = T_{\alpha/\gamma, t^{\gamma}}$ .

EXAMPLE A.6.2.

 $\begin{aligned} f_t(x) &= \operatorname{sgn}(x) \left( |x|^{\alpha} - t^{\alpha} \right)_+^{1/\alpha}, \quad t \in [0, \infty), \quad \alpha > 0, \\ p(t) &= e^{-pt^{\alpha}}, \quad t \in [0, \infty), \\ s \oplus t &= (s^{\alpha} + t^{\alpha})^{1/\alpha}. \end{aligned}$ Hence  $U_t &= U_{\alpha, t}$  if p = 0.

.

EXAMPLE A.6.3.

 $f_t(x) = \operatorname{sgn}(x) \left[ (|x|^{\alpha} + 1)^t - 1 \right]^{1/\alpha}, \quad t \in (0, 1], \ \alpha > 0,$   $p(t) = t^{\gamma}, \quad t \in (0, 1], \ \gamma \ge 0,$  $s \oplus t = s t.$ 

.

#### References

## References

- Abouammoh, A.M. (1987). On discrete  $\alpha$ -unimodality. *Statistica Neerlandica* 41, 239-244.
- Ahmad, R. and Abouammoh, A.M. (1977). On the Structure and Applications of Infinite Divisibility, Stability and Symmetry in Stochastic Inference. Recent Developments in Statistics. North-Holland Publishing Company, Amsterdam, The Netherlands.
- Alf, C. and O'Connor, T.A. (1977). Unimodality of the Lévy spectral function. Pacific J. Math. 69, 285-290.
- Bendat, J. and Sherman, S. (1955). Monotone and convex operator functions. *Trans. Amer. Math. Soc.* **79**, 58-71.
- Berg, C. and Forst, G. (1983). Multiple self-decomposable probability measures on  $\mathbb{R}_+$  and  $\mathbb{Z}_+$ . Z. Wahrsch. verw. Gebiete 62, 147-163.
- Billingsley, P. (1968). Convergence of Probability Measures. John Wiley and Sons, New York, London, Sidney.
- Billingsley, P. (1979). Probability and Measure. John Wiley and Sons, New York, London, Sidney.
- Bouwkamp, C.J. (1986). Solution to problem 85-16\*, SIAM Review 28, 568-569.
- Brown, M. (1980). Bounds, inequalities and monotonicity properties for some specialized renewal processes, Ann. Probab. 8, 227-240.
- Bruijn, N.G. de and Erdös, P. (1953). On a recursion formula and some Tauberian theorems. J. Research Nat. Bureau Stand. 50, 161-164.
- Carasso, A.S. (1987). Infinite divisible pulses, continuous deconvolution, and the characterization of linear time invariant systems. *SIAM J. Appl. Math.* 47, 892-927.
- Feller, W. (1968). An Introduction to Probability Theory and Its Applications. Vol. I, 3rd. ed., John Wiley and Sons, New York, London, Sidney.
- Feller, W. (1971). An Introduction to Probability Theory and Its Applications. Vol. II, 2nd. ed., John Wiley and Sons, New York, London, Sidney.
- Forst, G. (1979). A characterization of self-decomposable probabilities on the halfline. Z. Wahrsch. verw. Gebiete 49, 349-352.
- Frenk, J.B.G. (1987). Renewal theory and completely monotone functions. Private communication.

- Goldie, C.M. (1967). A class of infinitely divisible distributions. *Proc. Cambridge Phil. Soc.* **63**, 1141-1143.
- Hall, P. (1981). A comedy of errors: the canonical form for the stable characteristic function. *Bull. London Math. Soc.* 13, 23-27.
- Hansen, B.G. (1988). On logconcave and logconvex infinitely divisible sequences and densities. Ann. Probab. 16, 1832-1839.
- Hansen, B.G. (1989a). Self-decomposable distributions and branching processes. Memorandum COSOR 89-06, Technical University of Eindhoven, Eindhoven, The Netherlands.
- Hansen, B.G. (1989b). Note on Urbanik's class  $L_n$ . Memorandum COSOR 89-07, Technical University of Eindhoven, Eindhoven, The Netherlands.
- Hansen, B.G. (1989c). Reverse self-decomposability. *Memorandum COSOR* 89-08, Technical University of Eindhoven, *Eindhoven, The Netherlands*.
- Hansen, B.G. and Frenk, J.B.G. (1989). A simple proof of some properties of the renewal function. In preparation.
- Hansen, B.G. and Steutel, F.W. (1988). On moment sequences and infinitely divisible sequences. J. Math. Anal. Appl. 136, 304-313.
- Hansen, B.G. and Willekens, E. (1989). The generalized logarithmic series distribution. *Stat. Probab. Letters*, to appear.
- Harn, K. van (1978). Classifying Infinitely Divisible Distributions by Functional Equations. Mathematical Center Tracts 103, Mathematisch Centrum, Amsterdam, The Netherlands.
- Harn, K. van, Steutel, F.W. and Vervaat, W. (1982). Self-decomposable discrete distributions and branching processes. Z. Wahrsch. verw. Gebiete 61, 97-118.
- Horn, R.A. (1970). On moment sequences and renewal sequences. J. Math. Anal. Appl. 31, 130-135.
- Ibragimov, I.A. (1956). On the composition of unimodal distributions. *Theor. Probab. Appl.* 1, 255-260.
- Jurek, Z.J. (1981). Limit Distributions for Sums of Shrunken Random Variables. Dissertationes Math. (Rozprawy Math.) 185. PWN 1981.
- Jurek, Z.J. (1985). Relations between the s-selfdecomposable and selfdecomposable measures. Ann. Probab. 13, 592-608.
- Jurek, Z.J. (1988). Random integral representations for classes of limit distributions similar to Lévy class  $L_0$ . Probab. Theory Rel. Fields 78, 473-490.

#### References

- Jurek, Z.J. (1989). Random integral representations for classes of limit distributions similar to Lévy class  $L_0$ , II. Nagoya Math. J. 114, 53-64.
- Kaluza, T. (1928). Über die Koeffizienten Reziproker Potenzreihen. Math. Z. 28, 161-170.
- Karlin, S. (1968). Total Positivity. Vol. I, Stanford University Press, Stanford, California.
- Keilson, J. and Gerber, H. (1971). Some results on discrete unimodality. J. Amer. Statist. Assoc. 64, 472-483.
- Keilson, J. and Servi, L.D. (1987). A distributed Poisson approximation for preemptresume clocked schedules. *Technical report*, OR-170-87, MIT, Boston, Mass.
- Khintchine, A.Y. (1938). On unimodal distributions (In Russian). Izv. Nauchno-Issed. Inst. Mat. Mekh. Tomsk. Gos. Univ. 2, 1-7.
- Kingman, J.F.C. (1972). Regenerative Phenomena. John Wiley and Sons, London, New York, Sidney, Toronto.
- Loève, M. (1977). Probability Theory. Vol. I, 4th. ed.. Graduate Texts in Mathematics, Vol. 45. Springer-Verlag, New York, Heidelberg, Berlin.
- Lukacs, E. (1970). Characteristic Functions. 2nd ed., Griffin, London.
- O'Connor, T.A. (1979a). Infinitely divisible distributions with unimodal Lévy spectral functions. Ann. Probab. 7, 494-499.
- O'Connor, T.A. (1979b). Infinitely divisible distributions similar to class L distributions. Z. Wahrsch. verw. Gebiete 50, 265-271.
- O'Connor, T.A. (1981). Some classes of limit laws contairing the stable distributions. Z. Wahrsch. verw. Gebiete 55, 25-33.
- Olshen, R.A. and Savage, L.J. (1970). A generalized unimodality. J. Appl. Probab. 7, 21-34.
- Roberts, A.W. and Varberg, V.E. (1973). Convex Functions. Academic Press, New York.
- Ross, S.M. (1983). Stochastic Processes. John Wiley and Sons, New York, London, Sidney.
- Royden, H.L. (1968). Real Analysis. 2nd. Ed., Macmallian, New York.
- Sato, K. and Yamazato, M. (1978). On distribution functions of class L. Z. Wahrsch. verw. Gebiete 43, 273-308.

- Shohat, J.A. and Tamarkin, J.D. (1943). *The Problem of Moments*, Mathematical Surveys I, Amer. Math. Soc., *Providence*, *R.I.*
- Steutel, F.W. (1969). Note on completely monotone densities. Ann. Math. Stat. 40, 1130-1131.
- Steutel, F.W. (1970). Preservation of Infinite Divisibility under Mixing and Related Topics. Mathematical Center Tracts 33, Mathematisch Centrum, Amsterdam, The Netherlands.
- Steutel, F.W. (1974). On the tails of infinitely divisible distributions. Z. Wahrsch. verw. Gebiete 28, 273-276.
- Steutel, F.W. (1988). Note on discrete  $\alpha$ -unimodality. *Statistica Neerlandica* 42, 137-140.
- Steutel, F.W. and Hansen, B.G. (1988). Haight's distribution and busy periods. Statist. Probab. Letters 7, 301-302.
- Steutel, F.W. and Harn, K. van (1979). Discrete analogues of self-decomposability and stability. Ann. Probab. 7, 893-899.
- Sumita, U. and Masuda, Y. (1987). Classes of probability density functions having Laplace transforms with negative zeros and poles. Adv. Appl. Probab. 19, 632-651.
- Thorin, O. (1977). On infinite divisibility of the lognormal distribution. Scand. Actuarial J. 121-148.
- Widder, D.V. (1946). The Laplace Transform. Princeton University Press, Princeton.
- Wolfe, S.J. (1971). On the unimodality of L functions. Ann. Math. Statist. 42, 912-918.
- Yamazato, M. (1978). Unimodality of infinitely divisible distribution functions of class L. Ann. Probab. 6, 523-531.
- Yamazato, M. (1982). On strongly unimodal infinitely divisible distributions. Ann. Probab. 10, 589-601.
## List of symbols and index

## List of symbols

| $A_T$                        | Definition 3.2.6             | 21 |
|------------------------------|------------------------------|----|
| $A_T^*$                      | Definition 3.2.6             | 21 |
| D                            | Section 4.4, equation (4.17) | 44 |
| <i>ID</i> (·)                | Notation 1.3.5               | 5  |
| $MF(\cdot)$                  | Definition 3.5.1             | 32 |
| <i>MF</i> <sup>*</sup> (·)   | Definition 3.5.1             | 32 |
| $MS(\cdot)$                  | Notation 3.2.1               | 19 |
| $MS^*(\cdot)$                | Notation 3.2.1               | 19 |
| $MS_T(\cdot)$                | Notation 3.2.1               | 19 |
| $MS_T^*(\cdot)$              | Notation 3.2.1               | 19 |
| $S_{\alpha}(\mathbb{N}_0)$   | Definition 5.3.1             | 49 |
| $S_{\alpha}(\mathbb{R})$     | Notation 5.4.9               | 62 |
| $S_{\alpha}(\mathbb{R}_{+})$ | Definition 5.5.1             | 64 |
| T <sub>c</sub>               | Section 1.4                  | 8  |
| $T_{\alpha,t}$               | Section 6.3, equation (6.4)  | 72 |
| $U_{\alpha,t}$               | Section 6.3, equation (6.3)  | 72 |
| $U_t$                        | Section A.1, equation (A.2)  | 85 |

## Index

| a-monotone, 48                        |
|---------------------------------------|
| $\alpha$ -selfdecomposable            |
| - on IN <sub>0</sub> , 49             |
| - on <b>R</b> +, 64                   |
| - on <b>R</b> , 53                    |
| α-unimodal, 47                        |
| bounded sequence of random variables, |
| б                                     |
| classification, 3                     |
| domain of normal attraction, 10       |
| infinitely divisible, 3               |
| log-concave                           |
| - function, 41                        |
| - sequence, 38                        |
| log-convex                            |
| - function, 41                        |
| - sequence, 38                        |
| mixture of                            |

- exponential distributions, 65 - geometric distributions, 34 moment function, 32 moment sequence, 19 representing measure, 19, 32 self-decomposable, 8 s-self-decomposable, 46 stable, 8 triangular array, 6 uniformly asymptotically negligible (uan), 6

CWI TRACTS

1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984

2 J.J. Dijkstra. Fake topological Hilbert spaces and characteri-zations of dimension in terms of negligibility. 1984. 3 A.J. van der Schaft. System theoretic descriptions of physical ns. 1984

4 J. Koene. Minimal cost flow in processing networks, a primal proach. 1984.

5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.

6 A.P.W. Böhm. Dataflow computation. 1984.

7 A. Blokhuis. Few-distance sets. 1984.

8 M.H. van Hoorn. Algorithms and approximations for queue-ing systems. 1984.

9 C.P.J. Koymans. Models of the lambda calculus. 1984. 10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984. 11 N.M. van Dijk. Controlled Markov processes; time-discretization. 1984.

12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.

13 D. Grune. On the design of ALEPH. 1985.

14 J.G.F. Thiemann. Analytic spaces and dynamic program-ming: a measure theoretic approach. 1985.

15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.

16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial differential operators: a case-study in Fourier integral opera-tors. 1985.

17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.

18 A.D.M. Kester. Some large deviation results in statistics. 1985.

19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. 1986.

20 B.F. Schriever. Order dependence. 1986. 21 D.P. van der Vecht. Inequalities for stopped Brownian 1. 1986

22 J.C.S.P. van der Woude. Topological dynamix. 1986.

23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution, 1986.

24 J.C.M. Bacton. Filters and ultrafilters over definable subsets of admissible ordinals. 1986.

25 A.W.J. Kolen. Tree network and planar rectilinear location theory. 1986.

26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.

27 A.J.M. van Engelen. Homogeneous zero-dimensional abso-lute Borel sets. 1986.

28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.

29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.

30 A.G. de Kok. Production-inventory control models: approxi-mations and algorithms. 1987.

31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987. 32 J.H.J. Einmahl. Multivariate empirical processes. 1987.

J. Vrieze. Stochastic games with finite state and action spaces. 1987.
 P.H.M. Kersten. Infinitesimal symmetries: a computational approach. 1987.

35 M.L. Eaton. Lectures on topics in probability inequalities. 1987.

36 A.H.P. van der Burgh, R.M.M. Mattheij (eds.). Proceed-ings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.

applies maintenautics (ICLERT 07), 1501.
37 L. Stougie. Design and analysis of algorithms for stochastic integer programming, 1987.
38 J.B.G. Frenk. On Banach algebras, renewal measures and regenerative processes. 1987.

39 H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.

40 J.L. Geluk, L. de Haan. Regular variation, extensions and Tauberian theorems. 1987.

41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987. 42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, for-malisms, and tools. 1987.

43 H.L. Bodlaender. Distributed computing: structure and complexity. 1987.

44 A.W. van der Vaart. Statistical estimation in large parameter spaces, 1988.

45 S.A. van de Geer. Regression analysis and empirical processes. 1988.

46 S.P. Spekreijse. Multigrid solution of the steady Euler equa-tions. 1988.

47 J.B. Dijkstra. Analysis of means in some non-standard situations. 1988.

48 F.C. Drost. Asymptotics for generalized chi-square goodness-of-fit tests. 1988.

49 F.W. Wubs. Numerical solution of the shallow-water equa-tions. 1988.

50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.

51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing. 1988.

52 P.M. van Loon. Continuous decoupling transformations for linear boundary value problems. 1988.

53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequential quadratic pro-gramming in function space. 1988.

54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.

55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.

56 J.C. Ebergen. Translating programs into delay-insensitive circuits. 1989.

57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.

58 M.C.M. de Gunst. A random model for plant cell popula-tion growth. 1989.

59 D. van Dulst. Characterizations of Banach spaces not con-taining l<sup>1</sup>. 1989.

60 H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.

61 P. de Jong, Central limit theorems for generalized multi-linear forms. 1989.

62 V.J. de Jong. A specification system for statistical software. 1989

63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1990.

64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1990.

65 B.M.M. de Weger. Algorithms for diophantine equations. 1989

66 A. Jung. Cartesian closed categories of domains. 1989. 67 J.W. Polderman. Adaptive control & identification: Conflict of conflux?. 1989.

68 H.J. Woerdeman. Matrix and operator extensions. 1989.

69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1990.

70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.). Twenty-five years of operations research in the Netherlands: Papers dedi-cated to Gijs de Leve. 1990.

91 J.M. Geysel. Transcendence in fields of positive characteris-tic, 1979.

92 P.J. Weeda. Finite generalized Markov programming. 1979. 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977.

94 A. Bijlsma. Simultaneous approximations in transcendental number theory. 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978. 96 P.M.B. Vitånyi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; function equations and algorithms. 1984.

98 R. Geel. Singular perturbations of hyperbolic type. 1978. 99 J.K. Lenstra, A.H.G. Rinnooy Kan, P. van Emde Boas (eds.). Interfaces between computer science and operations research. 1978.

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 1. 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 2. 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces. 1978

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978.

104 J.M. van Wouwe. GO-spaces and generalizations of metri-zability. 1979.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982.

106 A. Schrijver (ed.). Packing and covering in combinatorics. 1979

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979.

108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979.

109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 2. 1979.

110 J.C. van Vliet. ALGOL 68 transput, part I: historica review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part II: an implemen-tation model. 1979.

112 H.C.P. Berbee. Random walks with stationary increments and renewai theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979.

114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979. 115 P.C. Baayen, J. van Mill (eds.). Topological structures II, part I. 1979.

part 1. 1979. 116 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 2. 1979.

117 P.J.M. Kallenberg, Branching processes with continuous state space. 1979. 118 P. Groeneboom. Large deviations and asymptotic

efficiencies, 1980.

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980.

120 W.P.M. de Ruyter. On the asymptotic analysis of large scale ocean circulation. 1980.

121 W.H. Hacmers. Eigenvalue techniques in design and graph theory. 1980.

122 J.C.P. Bus. Numerical solution of systems of nonlinear equations. 1980.

123 I. Yuhasz. Cardinal functions in topology - ten years later.

124 R.D. Gill. Censoring and stochastic integrals. 1980.

R. Eising. 2-D systems, an algebraic approach. 1980.
 I.Z. G. van der Hoek. Reduction methods in nonlinear pro-gramming. 1980.
 J.W. Klop. Combinatory reduction systems. 1980.

128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. *ILP: intermediate language for pictures.* 1980.

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.

132 H.M. Mulder. The interval function of a graph. 1980.

133 C.A.J. Klaassen. Statistical performance of location esti-mators. 1981.
134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68. 1981.

135 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part I. 1981.

(cds.), Formal methods in the study of language, part 1. For 136 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.), Formal methods in the study of language, part II. 1981. 137 J. Telgen. Redundancy and linear programs. 1981.

138 H.A. Lauwerier. Mathematical models of epidemics. 1981. 139 J. van der Wal. Stochastic dynamic programming, succes-sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis. 1981.

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981.

142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. 1981.

145 A.J. Brentjes. Multi-dimensional continued fraction algo-rithms. 1981.

146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.

147 H.H. Tigelaar. Identification and informative sample size. 1982

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems. 1983.

149 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg, W.K. Vietsch (eds.). From A to Z, proceedings of a sympo-sium in honour of A.C. Zaanen. 1982.

150 M. Veldhorst. An analysis of sparse matrix storage schemes, 1982.

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequencies. 1982.

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired ser-vices. 1982.

Vices. 1962.
154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.
155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two-dimensional smooth commutative formal groups over an alge-braically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983.

160 A. Rezus. Abstract AUTOMATH. 1983.

161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.

an agenual: approach. 1905.
 162 JJ. Dik. Tests for preference. 1983.
 163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983.
 164 F.A. van der Duyn Schouten. Markov decision processes with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983.

166 H.B.M. Jonkers. Abstraction, specification and implem tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program-ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.

MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points. 1963.

2 A.R. Bloemena. Sampling from a graph. 1964.

3 G. de Leve. Generalized Markovian decision processes, part 1: model and method. 1964.

4 G. de Leve. Generalized Markovian decision processes, part II: probabilistic background. 1964.
5 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.

6 M.A. Maurice. Compact ordered spaces. 1964.

7 W.R. van Zwet. Convex transformations of random variables. 1964.

8 J.A. Zonneveld. Automatic numerical integration. 1964.

9 P.C. Baayen. Universal morphisms. 1964.

10 E.M. de Jager. Applications of distributions in mathematical physics. 1964.

11 A.B. Paalman-de Miranda. Topological semigroups. 1964. 12 J.A.Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. Formal properties of newspaper Dutch. 1965.

13 H.A. Lauwerier. Asymptotic expansions. 1966, out of print: replaced by MCT 54. 14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

15 R. Doornbos. Slippage tests. 1966.

16 J.W. de Bakker. Formal definition of programming languages with an application to the definition of ALGOL 60. 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2. 1968.

19 J. van der Slot. Some properties related to compactness. 1968.

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and topology. 1968.

22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part J. 1968.

23 T.J. Dekker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968. 24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paërl. Representations of the Lorentz group and projec-tive geometry. 1969.

26 European Meeting 1968. Selected statistical papers, part I. 1968.

27 European Meeting 1968. Selected statistical papers, part II. 1968.

28 J. Oosterhoff. Combination of one-sided statistical tests. 1969.

29 J. Verhoeff. Error detecting decimal codes. 1969. 30 H. Brandt Corstius. Exercises in computational linguistics.

1970 31 W. Molenaar. Approximations to the Poisson, binomial and hypergeometric distribution functions. 1970.

hypergeometric distribution junctions, 15-6.
 32 L. de Haan. On regular variation and its application to the weak convergence of sample extremes. 1970.
 33 F.W. Steutel. Preservations of infinite divisibility under mixing and related topics. 1970.
 34 Lubberg. A Varback NS Kroopenpherg Cardinal func-

34 I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal func-tions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971.

36 J. Grasman. On the birth of boundary layers. 1971.

37 J.W. de Bakker, G.A. Blaauv, A.J.W. Duijvest 1911. J. Kuseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G. Zoutendijk. MC-25 Informatica Symposium. 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972.

39 H. Bavinck. Jacobi series and approximation. 1972. 40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972.

42 W. Vervaat, Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence, 1973,

44 H. Bart. Meromorphic operator valued functions. 1973.

45 A.A. Balkema. Monotone transformations and limit laws. 1973.

46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.

Jormula manipulation systems, part 1: the language. 1975. 47 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 2: the compiler. 1973. 48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973.

49 H. Kok. Connected orderable spaces. 1974.

50 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A. Koster, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G. Fisker (eds.). Revised report on the algorithmic language ALGOL 68, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory. 1974.

52 P.C. Baayen (ed.). Topological structures. 1974.

53 M.J. Faber. Metrizability in generalized ordered spaces. 1974. 54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.

55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974. 56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory, foundations, partitions and combinatorial geometry. 1974.

57 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.

58 W. Albers. Asymptotic expansions and the deficiency con-cept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1975.

60 F. Göbel. Queueing models involving buffers. 1975.

63 J.W. de Bakker (ed.). Foundations of computer science. 1975.

64 W.J. de Schipper. Symmetric closed categories. 1975. 65 J. de Vries. Topological transformation groups, 1: a categorical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

68 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977.

70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

71 J.A.E.E. van Nunen. Contracting Markov decision processes. 1976.

72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveguides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979.

74 H.J.J. te Riele. A theoretical and computational study of generalized aliquot sequences. 1976.

75 A.E. Brouwer. Treelike spaces and related connected topo-logical spaces. 1977.

76 M. Rem. Associons and the closure statements. 1976. 77 W.C.M. Kallenberg. Asymptotic optimality of likelihood ratio tests in exponential families. 1978.

78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.

79 M.C.A. van Zuijlen. Empirical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary problems. 1977.

K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 1. 1976.
 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 2. 1976.

83 L.S. van Benthem Jutting. Checking Landau's "Grundlagen" in the AUTOMATH system. 1979.

84 H.L.I. Busard. The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii-xii. 1977.

85 J. van Mill. Supercompactness and Wallmann spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program-ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977.

89 J.W. de Roever. Complex Fourier transformation and ana-lytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen. Characterization of optimal strategies in dynamic games. 1981.

·